

Advance Planning & Engineering Division (619) 446-5460

# **ENVIRONMENTAL IMPACT REPORT**

Project No. 180357 SCH No. 2010091079

SUBJECT: THE WATERMARK. GENERAL PLAN AMENDMENT, COMMUNITY PLAN AMENDMENT, REZONE, VESTING TENTATIVE MAP, PLANNED DEVELOPMENT PERMIT, CONDITIONAL USE PERMIT and a STREET VACATION to change the current land use designation from Industrial/Business Park to Commercial/Residential Prohibited and remove the Prime Industrial Lands identification on a portion of the project site, a rezone of a portion of the project site from IP-2-1 to CR-2-1 to construct a mixed-use development comprised of commercial office space, commercial retail space, and a movie theater. The Street Vacation is required to vacate Scripps Gateway Court. In addition to new development, the proposed project includes existing development that is occurring on the project site in accordance with existing/previous project approvals, including an approved CUP, Planned Industrial Development (PID) Permit, and Planned Development Permit (PDP) (CUP/PID No. 991027; CUP No. 174323/PDP No. 174234 Extension of Time). The 34.39-acre project site is located at 10137 Scripps Gateway Court and is designated Industrial/Business Park and the IP-2-1 zone within the Miramar Ranch North Community Plan area. (LEGAL DESCRIPTION: Lots 1-6, 8, 10, and 11; Scripps Gateway Unit No. 2; Map No. 14004). Applicant: Sudberry Properties.

UPDATE: September 18, 2013. Revisions and/or minor corrections have been made to this document, in response to comments submitted, when compared to the draft Environmental Impact Report. In accordance with the California Environmental Quality Act, Section 15088.5, the addition of new information that clarifies, amplifies, or makes insignificant modifications does not require recirculation as there are no new impacts and no new mitigation identified. An environmental document need only be recirculated when there is the identification of new significant environmental impacts or the addition of a new mitigation measure required to avoid a significant environmental impact. The modifications within the final environmental document do not affect the analysis or conclusions of the Environmental Impact Report. All revisions are shown in a strikethrough and/or underline format

# **CONCLUSIONS:**

Based on the analysis conducted for the project described above, the City has prepared the following Environmental Impact Report (EIR) in accordance with the California Environmental Quality Act (CEQA) to inform public agency decision-makers and the public of the significant environmental effects that could result if the project is approved and implemented, identify possible ways to minimize the significant effects, and describe reasonable alternatives to the project (State CEQA Guidelines Section 15121). As further described in the attached EIR, the City has determined that the project would have a significant environmental effect in the following areas: Transportation/Traffic Circulation/Parking (direct and cumulative) and Air Quality (direct relative to construction).

It is further demonstrated in the attached EIR that the project would not result in a significant environmental effect in the following areas: Land Use, Visual Quality/Neighborhood Character, Global Climate Change, Energy, Noise, Biological Resources, Historical Resources, Geologic Conditions, Paleontological Resources, Hydrology/Water Quality, Health and Safety, Public Services and Facilities, and Public Utilities.

Mitigation measures are proposed (Section 5.0) to reduce all Project impacts to below a level of significance, with the exception of **Transportation/Traffic Circulation/Parking**. Significant and unmitigated impacts relative to **Transportation/Traffic Circulation/Parking** would result on several street segments of Scripps Poway Parkway. The attached EIR and Technical Appendices document the basis for the above Determination.

## MITIGATION, MONITORING AND REPORTING PROGRAM:

A series of mitigation measures relative to **Transportation/Traffic Circulation/Parking** and **Air Quality** are identified within each issue area discussion in Section 5.0, Environmental Analysis, of the EIR to reduce environmental impacts. The mitigation measures are also fully contained in Section 11.0, Mitigation Monitoring and Reporting Program, of the EIR.

# **RECOMMENDED ALTERNATIVES FOR REDUCING SIGNIFICANT UNMITIGATED IMPACTS**

Based on the requirement that alternatives reduce significant impacts associated with the proposed project, the EIR considers the following Project Alternatives which are further detailed in the Executive Summary and Section 10.0 of the EIR:

- 1. No Project/Development Under Existing Approvals
- 2. Light Industrial Park
- 3. Prime Industrial Lands
- 4. Reduced Intensity 17 Percent Reduction in Trips

# Under CEQA Guideline Section 15126.6(e)(2), if the No Project Alternative is the

environmentally superior alternative, the EIR must also identify which of the other alternatives is environmentally superior. The EIR identifies both the Light Industrial Park or Prime Industrial Lands alternative as environmentally superior to the proposed project. These alternatives result in similar impacts when compared to each other and would result in reductions in the level of impacts associated with Transportation/Traffic Circulation/Parking. However, neither of these alternatives would accomplish the project's main goals and objectives. The Reduced Intensity – 17 Percent Reduction in Trips alternative would result in a slight reduction in Transportation/Traffic Circulation/Parking impacts and would meet the project objectives, but would not provide for the same amount of office space as the proposed project and does not include a hotel, which is included in the proposed project.

# **PUBLIC REVIEW DISTRIBUTION:**

Individuals, organizations, and agencies that received a copy or notice of the draft EIR and were invited to comment on its accuracy and sufficiency is provided below. Copies of the Draft EIR, the Mitigation Monitoring and Reporting Program and any technical appendices may be reviewed in the office of the Advanced Planning & Engineering Division, or purchased for the cost of reproduction.

# **RESULTS OF PUBLIC REVIEW:**

- () No comments were received during the public input period.
- () Comments were received but did not address the accuracy or completeness of the Draft Environmental Impact Report (EIR). No response is necessary and the letters are attached at the end of the EIR.
- (X) Comments addressing the accuracy or completeness of the Draft Environmental Impact Report (EIR) were received during the public input period. The letters and responses are located immediately after the Table of Contents.

arte,

Cathy Winterrowd Assistant Deputy Director Development Services Department

February 28, 2013 Date of Draft Report

September 18, 2013 Date of Final Report

Analyst: Shearer-Nguyen

# DISTRIBUTION OF DRAFT ENVIRONMENTAL IMPACT REPORT:

The following individuals, organizations and agencies received a copy or notice of the draft EIR and were invited to comment on its accuracy and sufficiency.

# <u>Federal</u>

MCAS Miramar (13)

### STATE OF CALIFORNIA

Caltrans District 11 (31) California Department of Fish and Game, South Coast Region (32) Department of Toxics and Substances (39) Regional Water Quality Control Board Region 9 (44) State Clearinghouse (46A) California Transportation Commission (51) California Transportation Commission (51A) California Transportation Commission (51B) Native American Heritage Commission (56)

## CITY OF SAN DIEGO

Mayor's Office (91) Councilmember Lightner, District 1 (MS 10A) Councilmember Faulconer District 2 (MS 10A) Councilmember Gloria, District 3 (MS 10A) Councilmember, District 4 (MS 10A) Councilmember Kersey, District 5 (MS 10A) Councilmember Zapf, District 6 (MS 10A) Councilmember Sherman, District 7 (MS 10A) Councilmember Alvarez, District 8 (MS 10A) Councilmember Emerald, District 9 (MS 10A) Development Services

EAS

Transportation Development Engineering Fire

Planning Review

Landscaping

Geology

Long-Range Planning

Plan Airports

Park and Recreation

Facilities Financing

DPM

Public Utilities Department Water/Wastewater

Long Range Planning & Water Resources Warren Lovell, San Diego Police Department (MS 776) Larry Trame, San Diego-Fire and Rescue Department (MS603) Transportation Development – DSD (78) Development Coordination – DSD (78A) Fire and Life Safety Services (79) Kenneth Barnes, San Diego Fire-Rescue (80) Library, Government Documents (81) Central Library (81A) Scripps Miramar Branch Library (81FF) Lisa Wood, Environmental Services Department (93A) Tom Tomlinson, Facilities Financing (93B) City Attorney (93C)

OTHER AGENCIES, ORGANIZATIONS AND INDIVIDUALS San Diego Audubon Society (167) Mr. Jim Peugh (167A) San Diego County Archaeological Society, Inc. (218) Beeler Canyon Conservancy (436) Scripps Miramar Ranch Planning Group (437) Alliant International University (438) Miramar Ranch North Planning Committee (439) Scripps Ranch Civic Association (440) Acquisitions, Walter Library USIS (441) Valerie Mosqueda, Briggs Law Corporation Melodie Hallett **Richard Hallett** Sal Magallanez Febbie P. Moore Jack Little Marian Marum Lorayne Burley Jennifer Cesaro Michael Cesaro Kevibn Clements Brad Field Jeff He, PhD Catherine Kastner Valorie McClelland Chistina Moseman Jim Moseman Tina Nelson Mark Sampson Mary Scudder Connie Simonson Annette Sorenson

James Sullivan

OTHER AGENCIES, ORGANIZATIONS AND INDIVIDUALS (CONTINUED) Chris Todd Deanna Wong Cathy Wood





# FINAL ENVIRONMENTAL IMPACT REPORT

SEPTEMBER 2013

SCH NO. 2010091079 PTS NO. 180357

# WATERMARK

# FINAL ENVIRONMENTAL IMPACT REPORT

SEPTEMBER 2013

SCH NO. 2010091079 PTS NO. 180357

# **TABLE OF CONTENTS**

# LETTERS OF COMMENT AND RESPONSES

# LIST OF ACRONYMS AND ABBREVIATIONS

EXECUTIVE	SUMMARY	ES-1
1.0 INT	RODUCTION	
1.1 Pur	Pose and Legal Authority	
1.1.1	Authority and Intended Uses of the EIR	1-1
1.1.2	Availability and Review of the Draft EIR	1-2
1.2 Scc	DPE AND CONTENT OF EIR	
1.2.1	Scope of EIR	1-3
1.2.2	Format of EIR	1-3
1.2.3	Incorporation by Reference	1-7
1.3 EVA	LUATION OF ENVIRONMENTAL EFFECTS	
1.3.1	Existing Conditions	1-8
1.3.2	Impact Analysis	1-8
1.4 Resi	PONSIBLE AND TRUSTEE AGENCIES	
1.4.1	California Department of Transportation	1-10
1.4.2	Regional Water Quality Control Board	1-10
1.4.3	Federal Aviation Administration	1-10
2.0 EIN	GIONAL SETTING	
2.2 Pro	DJECT LOCATION	
2.3 Pro	DJECT HISTORY	
2.4 Exis	STING SITE CONDITIONS	
2.4.1	Topography	
2.4.2	Biological Resources	
2.4.3	Cultural Resources	2-7
2.4.4	Geologic Conditions	
2.4.5	Paleontological Resources	
2.4.6	Visual Resources	
2.5 SUR	ROUNDING LAND USES	
2.6 PUB		
2.6.1		2-10
2.6.2	Fire Safety	2-11
2.1 PLA	NNING CONIEXI	
2.7.1	City of San Diego General Plan	2-13
2.1.2	wiramar Kanch North Community Plan	2-14
2.8 ZON	NING	Z-14

2.9 MC	AS MIRAMAR ALUCP	2-14
2.9 Pro	DJECT BASELINE	2-19
3.0 Pr	DJECT DESCRIPTION	
3.1 Pro	DJECT BACKGROUND	3-1
3.2 Pur	POSE AND OBJECTIVES OF THE PROPOSED PROJECT	3-2
3.3 Pro	DJECT CHARACTERISTICS	3-3
3.3.1	Miramar Ranch North Community Plan/General Plan Amendment	3-4
3.3.2	Proposed Zoning	3-7
3.3.3	Vesting Tentative Map	3-7
3.3.4	Planned Development Permit	
3.3.5	Street Vacation	3-31
3.5.6	Conditional Use Permit	3-31
3.4 Dis		3-31
4.0 His	TORY OF PROJECT CHANGES	4-1
		F /
5.0 EN	VIRONMENTAL ANALYSIS	
5.1 Lar		5.1-1
5.1.1	Existing Conditions	5.1-1
5.1.2	Impact Analysis	3.1-1 /
5.2 IIa 5.2 I	Existing Conditions	<b></b>
5.2.1	Impact Analysis	5 2 1 2
5.2.2	ual Quality/Neighborhood Character	J.2-12
<b>J.J VI</b>	Existing Conditions	5 3 <sub>-</sub> 1
532	Impact Analysis	5 3-8
54 Air	Ouality	<u>5 4-</u>
5.4.1	Existing Conditions	5.4-1
5.4.2	Impact Analysis	5.4-10
5.5 Glo	bal Climate Change	5.5-
5.5.1	Existing Conditions	5.5-1
5.5.2	Impact Analysis	5.5-14
5.6 Ene	ergy	5.6-
5.6.1	Existing Conditions	5.6-1
5.6.2	Impact Analysis	5.6-2
5.7 No	ise	<u>5.7-</u>
5.7.1	Existing Conditions	5.7-1
5.7.2	Impact Analysis	5.7-5
5.8 Bio	logical Resources	
5.8.1	Existing Conditions	5.8-1
5.8.2	Impact Analysis	5.8-8
5.9 His	torical Resources (Archaeological Resources and Historic Resources)	5.9-
5.9.1	Existing Conditions	5.9-1
5.9.2	Impact Analysis	5.9-2

5.10 Geologic Conditions	5.10-1
5.10.1 Existing Conditions	5.10-1
5.10.2 Impact Analysis	5.10-4
5.11 Paleontological Resources	5.11-1
5.11.1 Existing Conditions	5.11-1
5.11.2 Impact Analysis	5.11-2
5.12 Hydrology/Water Quality	<u>5.12-1</u>
5.12.1 Existing Conditions	
5.12.2 Impact Analysis	
5.13 Health and Safety	<u>5.13-1</u>
5.13.1 Existing Conditions	
5.13.2 Impact Analysis	
5.14 Public Services and Facilities	5.14-1
5.14.1 Existing Conditions	5.14-1
5.14.2 Impact Analysis	
5.15 Public Utilities	<u>5.15-1</u>
5.15.1 Existing Conditions	
5.15.2 Impact Analysis	5.15-8
	6 1
0.0 CUMULATIVE EFFECTS	0-1
6.1 Conoral Dian	<b>0-3</b>
6.1.2 Miramar Banch North Community Plan	6.3
6.2 Projects Considered for Cumulative Effects Analysis	0-J
6.2 MedImpact	0-3
6.2.2 Sharp Health	6-4
6.2.3 Casa Mira View	6-4
6.3 Cumulative Effects Analysis	
6.3.1 Land Use	<u></u> б-5
6.3.2 Transportation/Traffic Circulation/Parking	
6.3.3 Visual Quality/Neighborhood Character	
634 Air Quality	6-7
6.3.5 Global Climate Change	6-7
636 Energy	6-8
637 Noise	6-8
638 Biological Resources	6-8
6.3.9 Historic Resources (Archaeological Resources and Historic Resources)	6-9
6.3.10 Geologic Conditions	6-9
6 3 11 Paleontological Resources	6-9
6.3.12. Hydrology/Water Quality	6-9
6 3 13 Health and Safety	6-9
6.3.14 Public Services and Facilities	6-10
6 3 15 Public Utilities	
	0 10

7.0 EFFECTS NOT FOUND TO BE SIGNIFICANT	7-1
7.1 Agricultural Resources and Forestry	7-1
7.2 Mineral Resources	
7.3 Recreation	
7.4 Population and Housing	
8.0 GROWTH INDUCEMENT	8-1
8.1 Existing Conditions	
8.2 Impact Analysis	
9.0 SIGNIFICANT IRREVERSIBLE ENVIRONMENTAL CHANGES	
10.0 Alternatives	10-1
10.1 Alternatives Considered But Rejected	10-2
10.2 Alternatives Considered	10-3
10.2.1 Alternatives Analysis	10-3
10.2.2 Alternative 1 – No Project/Development under Existing Appro	vals10-4
10.2.3 Alternative 2 – Light Industrial Park	10-9
10.2.4 Alternative 3 – Prime Industrial Lands	10-16
10.2.5 Alternative 4 – Reduced Intensity – 17 Percent Reduction in Tri	ps10-23
10.3 Environmentally Superior Alternative	10-29
11.0 MITIGATION MONITORING AND REPORTING PROGRAM	11-1
11.1 Monitoring Activities	11-1
11.2 Program Procedures	11-2
11.3 Summary of Project Impacts and Mitigation Measures	11-2
12 References	12-1
13 INDIVIDUALS AND AGENCIES CONSULTED	13-1
14 Certification	14-1

# **A**PPENDICES

- Appendix A Notice of Preparation and Comment Letters
- Appendix B Public Scoping Meeting Comments and Transcript
- Appendix C Traffic Impact Analysis for Watermark
- Appendix D Air Quality Technical Report
- Appendix E Greenhouse Gas Evaluation
- Appendix F Noise Analysis
- Appendix G Watermark Biological Resources Report
- Appendix H Drainage Study
- Appendix I Update Geotechnical Investigation Report
- Appendix J Sewer Study Amendment
- Appendix K Preliminary Water Quality Technical Report
- Appendix L Water Supply Assessment Report
- Appendix M Letters/Responses to Service Providers
- Appendix N Federal Aviation Regulations Part 77 Letters on Non-Obstruction and ALUCP Consistency Letter
- Appendix O Prime Industrial Lands Criteria Analysis
- Appendix P Collocation/Conversion Suitability Analysis
- Appendix Q Preliminary Waste Management Plan
- Appendix R Preliminary Hydromodification Management Study
- Appendix S Final Paleontological Mitigation Report Scripps Gateway

# **FIGURES**

Figure 2-1.	Regional Map	2-2
Figure 2-2.	Vicinity Map	2-3
Figure 2-3.	Project Location Map	2-4
Figure 2-4.	Existing Site Conditions	2-6
Figure 2-5.	Surrounding Land Uses	2-9
Figure 2-6.	City of San Diego General Plan Land Use Map	2-15
Figure 2-7.	Miramar Ranch North Community Plan Land Use Map	2-16
Figure 2-8.	Existing Zoning	2-17
Figure 2-9.	MCAS Miramar – Airport Influence Area Map	2-18
Figure 3-1.	Miramar Ranch North Community Plan – Proposed Land Use Designations	
Figure 3-2.	Proposed Zoning	
Figure 3-3.	Watermark Vesting Tentative Map	
Figure 3-4.	Proposed Lotting	
Figure 3-5.	Watermark Site Plan	
Figure 3-6.	Watermark Pedestrian Plan	
Figure 3-7a.	Project Elevations	
Figure 3-7b.	Project Elevations	
Figure 3-7c.	Project Elevations	
Figure 3-8.	Watermark Landscape Plan	
Figure 3-9	Entry from Scripps Highlands Road Landscape Design	3-23
Figure 3-10	First Project Entry from Private Street Landscape Design	3-24
Figure 3-11	Primary Public Plaza Landscape Design	3-25
Figure 3-12.	Secondary Driveway from Private Street Landscape Design	
Figure 3-13	Off-ramp Frontage Landscape Design	3-27
Figure 3-14	Entry off Scripps Poway Parkway Landscape Design	3-28
1 iguite 5-14.	Entry on oenpps roway rankway Landscape Design	
Figure 4-1.	Proposed Multi-Use Path	
Figure 4-2.	Multi-Use Path Cross-Section	
Eigung 5 1 1	City of San Diago Conoral Dian Willogo Brononoity Man	512
Figure $5.1$ -1.	Drimo Lodustrial Landa Man	
Figure 5.1-2.	Multi Habitat Dlanging Area	
Figure 5.1-5. Eigenve 5.1.4	Multi-Habitat Planning Area.	5.1-10 5 1 10
Figure 5.1-4.	MCAS Miraman Compatibility Dalian Mari	5.1-19 5 1 26
Figure 5.1-5. $\Gamma'$	MCAS Miramar Compatibility Policy Map: Noise	5.1-20
Figure 5.1-6. $\mathbf{F}$	MCAS Miramar Compatibility Policy Map: Safety	5.1-2/
Figure 5.1-/.	MCAS Miramar Compatibility Policy Map: Airspace Protection	5.1-28
Figure 5.1-8.	MCAS Miramar Compatibility Policy Map: Overflight	5.1-29
Figure 5.2-1a.	Study Area Boundary and Intersection Key	5.2-2
Figure 5.2-1b.	Study Area Boundary and Intersection Key	5.2-3
Figure 5.2-2a.	Existing Average Daily Traffic Volumes	5.2-6

Figure 5.2-2b.	Existing Average Daily Traffic Volumes	5.2-7
Figure 5.2-3a.	Project Trip Distribution	5.2-16
Figure 5.2-3b.	Project Trip Distribution	5.2-17
Figure 5.2-4a.	Project Only (ADT) Traffic Assignment	5.2-18
Figure 5.2-4b.	Project Only (ADT) Traffic Assignment	5.2-19
Figure 5.2-5a.	Existing with Project Average Daily Traffic Volumes	5.2-21
Figure 5.2-5b.	Existing with Project Average Daily Traffic Volumes	5.2-22
Figure 5.2-6a.	Near Term without Project Average Daily Traffic Volumes	5.2-27
Figure 5.2-6b.	Near Term without Project Average Daily Traffic Volumes	5.2-28
Figure 5.2-7a.	Near Term with Project Average Daily Traffic Volumes	5.2-33
Figure 5.2-7b.	Near Term with Project Average Daily Traffic Volumes	5.2-34
Figure 5.2-8a.	Horizon year 2030 without Project Average Daily Traffic Volumes	5.2-39
Figure 5.2-8b.	Horizon year 2030 without Project Average Daily Traffic Volumes	5.2-40
Figure 5.2-9a.	Horizon year 2030 with Project Average Daily Traffic Volumes	5.2-45
Figure 5.2-9b.	Horizon year 2030 with Project Average Daily Traffic Volumes	5.2-46
0		
Figure 5.3-1.	Current Conditions Aerial	5.3-2
Figure 5.3-2a.	Existing Development - Scripps Poway Parkway and I-15	5.3-15
Figure 5.3-2b.	Photo Simulation – Scripps Poway Parkway and I-15	5.3-16
Figure 5.3-3a.	Existing Development - Scripps Poway Parkway and Scripps Highlands Drive.	5.3-17
Figure 5.3-3b.	Photo Simulation - Scripps Poway Parkway and Scripps Highlands Drive	5.3-18
0		
Figure 5.4-1.	Wind Rose – MCAS Miramar	5.4-2
0		
Figure 5.8-1.	Existing Vegetation and Project Impacts	5.8-2
0		
Figure 5.10-1.	Geologic Map	5.10-3
_		
Figure 5.11-5.	Grading into Native Soils	5.11-5
Figure 5.14-1.	Location of Public Services	5.14-2
Figure 6-1.	General Location of Cumulative Projects	6-2
Figure 10-1.	Reduced Intensity – 17 Percent Reduction in Trips Alternative Site Plan	10-24

# TABLES

Table ES-1.	Baseline Conditions	ES-4
Table ES-2.	Proposed Project Development Intensity	ES-5
Table ES-3.	Summary of Environmental Impacts and Mitigation Measures	ES-7
Table ES-4.	Summary of NOP Comments and Scoping Meeting Comments	ES-9
Table 2-1.	Baseline Conditions	2-20

Table 3-1.	Proposed Project Development Intensity	
Table 3-2.	Watermark – Maximum Structure Height	
Table 3-3.	Landscaping Materials	3-29
Table 5.2-1.	Study Area Street Segments and Intersections	5.2-4
Table 5.2-2.	Existing Street Segment Levels of Service	5.2-9
Table 5.2-3.	Existing Intersection Levels of Service	5.2-10
Table 5.2-4.	Existing Ramp Meter Analysis	5.2-10
Table 5.2-5.	Existing Freeway Segment LOS Summary	5.2-11
Table 5.2-6.	Watermark Project Traffic Generation	5.2-15
Table 5.2-7.	Existing with Project Street Segment Levels of Service	5.2-23
Table 5.2-8.	Existing with Project Intersection Levels of Service	5.2-24
Table 5.2-9.	Existing with and without Project Ramp Meter Analysis	5.2-25
Table 5.2-10.	Existing with and without Project Freeway LOS Summary	5.2-25
Table 5.2-11.	Near Term without Project Street Segment Levels of Service	5.2-29
Table 5.2-12.	Near Term without Project Intersection Levels of Service	5.2-30
Table 5.2-13.	Near Term with and without Project Ramp Meter Analysis	5.2-31
Table 5.2-14.	Near Term with and without Project Freeway Segment LOS Summary	5.2-31
Table 5.2-15.	Near Term with Project Street Segment Levels of Service	5.2-35
Table 5.2-16.	Near Term with Project Intersection Levels of Service	5.2-36
Table 5.2-17.	Horizon Year 2030 without Project Street Segment Levels of Service	5.2-41
Table 5.2-18.	Horizon Year 2030 without Project Intersection Levels of Service	5.2-42
Table 5.2-19.	Horizon Year 2030 without Project Ramp Meter Analysis	
Table 5.2-20.	Horizon Year 2030 with and without Project Freeway Segment LOS Summary	
Table 5 2-21	Horizon Year 2030 with Project Street Segment Levels of Service	5 2-47
Table 5 2-22	Horizon with Project Intersection Levels of Service	5 2-48
Table 5 2-23	Near Term and Horizon Year 2030 Intersection Mitigation LOS Comparison	5 2-50
Table 5 2-24	Roadway Classifications and Levels of Service	5 2-52
Table 5.2-25	Level of Service Criteria	5 2-53
Table 5.2-26.	Congestion Management Program Requirements	5.2-53
Table 5.4-1.	Ambient Background Concentrations	5.4-3
Table 5.4-2.	Ambient Air Quality Standards	5.4-7
Table 5.4-3.	Significance Criteria for Air Quality Impacts	5.4-11
Table 5.4-4.	Operational Emissions	5.4-14
Table 5.4-5.	CO Hot Spots Evaluation	5.4-16
Table 5.4-6.	Construction Equipment Requirements	5.4-18
Table 5.4-7.	Estimated Maximum Daily Construction Emissions	5.4-20
Table 5.5.1	State of California GHG Emissions by Sector	550
Table 5.5-1. Table 5.5.2	Clobal Warming Dotentials and Atmospheric Lifetimes of CUCa	J.J-Z
Table 5.5-2. Table 5.5.2	San Diogo County 2006 CHC Emissions by Cators	
Table 5.3-3. Table $E = 4$	Construction CHC Emissions	
1 able 5.5-4.	Construction GHG Emissions	
1 able 3.3-3.	Summary of Operational Greenhouse Gas Emissions –	E E 4 7
	business as Usual Scenario	
1 adle 5.5-6.	Summary of Estimated Operational Greenhouse Gas Emissions	

Table 5.5-7.	Summary of Emissions Reduction	5.5-22
Table 5.6-1.	Watermark Project Sustainable Design Features	5.6-3
Table 5.7-1.	City of San Diego Noise Land Use Compatibility Chart	5.7-3
Table 5.7-2.	Measured Ambient Noise Levels	5.7-4
Table 5.7-3.	Sound Level Limits in Decibels	5.7-5
Table 5.7-4.	Delivery Truck Noise Levels	5.7-9
Table 5.7-5.	Project HVAC Noise Levels (Northern Property Line)	5.7-10
Table 5.7-6.	Project HVAC Noise Levels (Southern Property Line)	5.7-11
Table 5.7-7.	Project HVAC Noise Levels (Eastern Property Line)	5.7-11
Table 5.7-8.	Cumulative Noise Levels	5.7-12
Table 5.7-9.	On-site HVAC Noise Levels (worst case)	5.7-13
Table 5.7-10.	Traffic Parameters	5.7-14
Table 5.7-11.	Future Exterior Noise Levels	5.7-14
Table 5.7-12.	Near Term Noise Levels without Project	5.7-17
Table 5.7-13.	Near Term + Project Noise Levels	5.7-18
Table 5.7-14.	Near Term versus Near Term + Project Noise Levels	5.7-18
Table 5.7-15.	Construction Noise Levels	5.7-21
Table 5.8-1.	Summary of Existing Habitats	
Table 5.8-2.	Plant Species Observed On-Site	5.8-4
Table 5.8-3.	Wildlife Species Observed On-Site	5.8-6
Table 5.8-4.	Sensitive Species with the Potential to Occur On-site	5.8-7
Table 5.8-5.	Summary of Impacts	5.8-9
Table 5.10-1.	Probabilistic Site Parameters for Selected Faults	5.10-2
Table 5.11-1.	Paleontological Determination Matrix	5.11-3
Table 5.12-1.	Summary of Peak Discharges for Basin 1	5.12-5
Table 5.15-1.	Watermark Water Demand Analysis	5.15-11
Table 5.15-2.	Estimated Solid Waste Generation from the Watermark – Occupancy P	hase5.15-13
Table 6-1.	Cumulative Noise Levels	6-8
Table 10-1.	Alternative 4 – Reduced Intensity – 17 Percent Reduction in Trips	
	Development Intensity	
Table 10-2.	Comparison of Alternatives to Proposed Project	
Table 11-1.	Mitigation Monitoring and Reporting Program	11-6

ESPONSES		the draft EIR. A	t in identifying and ording to Section project at issue, the test or perform all spond to significant in the EIR. Many ufficiency of the Where letters of ke-out/underline to the Final EIR	Page Number of Letter		4		7	11	16
COMMENTS AND R	SS	ls during the public review of	on the sufficiency of the documen might be avoided or mitigated. Acco actors such as the magnitude of the quire a lead agency to conduct every mments, lead agencies need only re aith effort at full disclosure is made. ddress the adequacy and/or s a courtesy to the commenters: dicated in the Final EIR in stri dicated in the thave been made ocument.	Representing		United States Marine		Caltrans Development Review Branch	Native American Heritage Commission	Department of Toxic Substances Control
LETTERS OF	R COMMENT LETTER	ttions, and individua has been included.	n EIR should focus ant effects of the project by feasible, in light of fo ct. CEQA does not re- When responding to co vers, as long as a good fo Draft EIR did not a Sees as appropriate as those changes are in- those changes are in- won as <u>underlined</u> ). F the environmental do	Date	GENCIES	March 20, 2013	ENCIES	April 17, 2013	March 4, 2013	April 8, 2013
	WATERMARK DRAFT EI	were received from agencies, organizing with corresponding staff responses	idelines Section 15204(a), review of a environment and ways in which the signific transment and ways in what is reasona apacts, and the geographic scope of the proje recommended or demanded by commentors. to provide all information requested by revien ing public review of the Watermark 1 wer, staff endeavored to provide respon ons to the February 2013 Draft EIR, nown as stricken and added text is sho ntained in the EIR or the adequacy of	Address	FEDERAL A	Marine Corps Air Station P.O. Box 452001 San Diego, CA 92145-2001	STATE AG	Caltrans, District 11 4050 Taylor Street San Diego, 92110	Native American Heritage Commission 915 Capitol Mall, Room 364 Sacramento, CA 95814	Department of Toxic Substances Control 5796 Corporate Avenue Cypress, CA 90630
		ving comment letters ch comment letter alo	unce with CEQA Gu- be possible impacts on the [I]be adequacy of an EII is likely environmental in dy, and experimentation al issues and do not need numents received duri intal document; howe have resulted in revisi- nere omitted text is sl act the conclusions co	Author		C.L. Thorton Community Plans and Liaison Officer		Jacob Armstrong, Chief	Dave Singleton Program Analyst	Dr. Nirumpa Suryavanshi Project Manager
		The follov copy of ea	In accorde analyzing ti 15204(a), <sub>1</sub> severity of in research, sta environment, of the cor environment format (wh do not affe	Letter		Υ		B	C	Q

Response to Letters of Comment – Page 1 September 2013

Page Number of Letter	20		24		25	33	34	35	37	38	39	57	58	60	61	79	67	68	70	71	72	£L
Representing	California State Clearinghouse		San Diego County Archaeological Society, Inc. Environmental Review Committee		Self	Self	Self	Self	Self	Self	Scripps Ranch Civic Association	Self	Self	Self	Self	Self	Self	Self	Self	Self	Self	Self
Date		SENCIES	March 14, 2013	UALS	April 7, 2013	March 28, 2013	April 14, 2013	April 12, 2013	March 24, 2013	April 14, 2013	April 14, 2013	April 5, 2013	April 13, 2013	April 7, 2013	April 14, 2013	April 14, 2013	April 2, 2013	April 14, 2013	March 28, 2013	March 30, 2013	March 29, 2013	April 10, 2013
Address	1400 Tenth Street P.O. Box 3044 Sacramento, CA	LOCAL AG	San Diego County Archaeological Society, Inc. P.O. Box 81106 San Diego, CA 92138-1106	DIVIDNI	11223 Walking Fern Cove San Diego, CA 92131 E-mail: lwitte@san.rr.com	E-mail: jenniwinston@yahoo.com	E-Mail: Michael.Cesaro@audatex.com	E-mail: kclements@corelogic.com	E-mail: Bmfields@aol.com	6044 Cornerstone Ct. W. Ste C San Diego, CA 92121 E-mail: ihe@accelagen.com	E-Mail: srilko4@aol.com	E-mail: cdk@kasner.com	E-mail: rsakurauchi@san.rr.com	E-mail: valmac0214@sbcglobal.net	E-mail: tjmoseman@aol.com	E-mail: jmoseman@san.rr.com	11414 Ann Arbor Lane San Diego, CA 92131 E-mail: mintchip@san.tr.com	10636 Frank Daniels Way San Diego, CA 92131 E-mail: marksampson@sbcglobal.net	E-mail: mary.scudder@yahoo.com	E-mail: ksimonsen@san.rr.com	E-mail: SorensonMarketing@att.net	9950-130 Scripps Vista Way San Diego, CA 92131 E-mail: jimsully@gmail.com
Author			James W. Royle, Jr. Chairperson		Lorayne Burley	Jennifer Cesaro	Michael Cesaro	Kevin Clements	Brad Fields	Jeff He, PhD	Bob Ilko	Catherine Kastner	Ruriko Kusumoto	Valorie McClelland	Christina Moseman	Jim Moseman	Tina Nelsen	Mark Sampson	Mary Scudder	Connie Simonsen	Annette Sorensen	James Sullivan
Letter	Ъ		ц		J	Н	Ι	ſ	К	L	Μ	Z	0	Ъ	Q	R	S	T	n	Λ	W	Х

Response to Letters of Comment – Page 2 September 2013

The Watermark Final Environmental Impact Report

Page Number of Letter	82	80	82	84
Representing	Self	Self	Self	Self
Date	April 10, 2013	April 3, 2013	March 28, 2013	March 28, 2013
Address	9950-130 Scripps Vista Way San Diego, CA 92131 E-mail: jimsully@gmail.com	E-mail: ctodd@wingertlaw.com	E-mail: deemwong@aol.com	E-mail: CWood@taxcomp.com
Author	James Sullivan	Chris Todd	Deanna Wong	Cathy Wood
Letter	Y	Z	$\mathbf{V}\mathbf{V}$	BB

The Watermark Final Environmental Impact Report

Response to Letters of Comment – Page 3 September 2013

RESPONSE							A-1 Comments noted.	
COMMENT	UNITED STATES MARINE CORPS MARINE CORPSAIR STATION MARINE CORPSAIR STATION P.O. BOX 452001 SAN DIEGO, CA S2145-2001 III 103 CP&L/180357 May 29, 2009	CITY OF SAN DIEGO DEVELOPMENT SERVICES DEPARTMENT PROJECT MANAGERENT DIVISION ATTN RENEE MEZO 1222 FIRST AVENUE MS 302 SAN DIEGO CA 92101-4154	RE: MIRAWAR RANCH NORTH COMMUNITY PLAN; THE WATERWARK, 10137 SCRIPPS GATEWAY COURT, JOB ORDER NUMBER 43-2396, PN 180357, APN 319-810-01 to 319-810-11	Dear Ms. Mezo,	This is in response to the project review submittal of May 20, 2009, which proposes commercial development within the Miramar Ranch North Community Planning area.	On October 2, 2008, the Airport Land Use Commission (ALUC) approved the adoption of a revised Marine Corps Air Station (MCAS) Miramar Airport Land Use Compatibility Plan (ALUCP) that is consistent with the 2005 MCAS Miramar Air Installations Compatible Use Zones (AICUZ) Update. As a result, any development on this site is subject to safety, height, and noise guidelines consistent with the 2005 AICUZ Update.	The proposed site is contained within the "WCAS Miramar AICUZ Study Area" identified in the 2005 AICUZ Update for MCAS Miramar. It has been determined that this project is: 1) within the adopted 2008 MCAS Miramar AUCP Airport Influence Area (AIA) Review Area II, 2) outside the 60+ dB Community Noise Equivalent Level (CNBL) noise contours, 3) outside all Accident Potential Zones (APZ), 4) beneath the Outer Horizontal Surface of MCAS Miramar (Federal Aviation Regulation Petuyiny Elight corridors for aircraft transiting to and from MCAS Miramar.	It has been determined that the proposed project is consistent with AICUZ noise and safety compatibility guidelines, and the structural heights of the proposed buildings do not appear to penetrate the Federal Aviation Administration (FAA) Part 77 Outer

11103 CP&L/180357 May 29, 2009	Horizontal Surface and/or any Terminal Instrument Procedures (TERPS) surfaces. However, please note that the FAA is the only agency that can officially determine if a structure exceeds an airspace surface and/or what impact it would have on air navigation.	This location will experience noise impacts from the Ground Controlled Approach (GCA) Box Pattern and Julian Flight Corridors for fixed-wing operations. The site will also experience noise impacts from the GCA Box Pattern and I-15 Flight Corridors for helicopter operations.	Occupants will routinely see and hear military aircraft and experience varying degrees of noise and vibration. Consequently, we are recommending full disclosure of noise and visual impacts to all initial and subsequent purchasers, lessees, or other potential occupants.	Since the project is within the AIA for the MCAS Miramar ALUCP, and to ensure that the project is consistent with ALUCP guidelines, we recommend that ALUC staff be contacted to determine if an official consistency determination is required.	Normal hours of operation at MCAS Miramar are as follows:	Monday through Thursday 7:00 a.m. to 12:00 midnight Friday 7:00 a.m. to 6:00 p.m. Saturday, Sunday, Holidays 8:00 a.m. to 6:00 p.m.	MCAS Miramar is a master air station, and as such, can operate 24 hours per day, 7 days per week. Fiscal and manpower constraints, as well as efforts to reduce the noise impacts of our operations on the surrounding community, impose the above hours of operation. Circumstances frequently arise which require an extension of these operating hours.	Thank you for the opportunity to review this land use proposal. If we may be of any further assistance, please contact Mr. Juan Lias at (858) 577-6603.

11103 CP6L/180357 May 29, 2009 Sincerely, CMLONTON Community Plans and Liaison Officer By direction of the Commanding Offic	Copy to: san Diego Courty Regional Airport Authority, Sandi Sawa

	STATE OF CALEORAIA—BUSINESS, TRANSPORTATION AND PROUSING AGENCY		
	DEPARTMENT OF TRANSPORTATION DISTRICT 11, DIVISION OF PLANNING ASO TAXLORS T. MS. 240 SAN DECO, CA. 2210 SAN DECO, CA. 2210 SAN DECO, CA. 2210 RAY (619) 683–696 FLAX (619) 683–699 RAY (619) 683–699		
		B-1	Comment noted. At the time of the traffic counts establishing existing
	April 17, 2013 11-SD-15 PM 17.31 The Watermark Elizabeth Sthearer-Nguyen City of San Diego		TIA is reflective of conditions as they existed when the traffic counts were complete and environmental analysis commenced.
	Planning and Development Services 1222 First Avenue, MS-501 San Diego, CA 92101 Dear Mrs. Shearer-Nguyen:	B-2	Comment noted. As shown on Figure 1-1 of the TIA, a widening into the median of Scripps Poway Parkway is proposed as suggested by this
	The California Department of Transportation (Caltrans) has reviewed the Draft Environmental Innuact Report (DEIR-SCH 2010091079) for the monosed Watermark noise: near the 1-15.		
	Calirans has the following comments: Traffic Analysis	Ч	Comment Noted. As shown on Figure 1-1 of the TIA, the planned turn lane has a "bulb-out" at the intersection and is currently planned to begin
B-1	<ul> <li>The proposed reconfiguration of the 1-15/Scripps Poway Parkway Interchange to shift the westbound (WB) through lanes on Scripps Poway Parkway to the north and provide additional queuing length for WB traffic on Scripps Poway Parkway has already been</li> </ul>	B-4	approximately 100 cast of the out-tamp. The Synchro Worksheets are contained in the annendices to the TIA as
	completed by Calitans.	-	discussed in the TIA. These worksheets will be made available to Caltrans
B-2	<ul> <li>The project should widen one lane (on Scripps Poway Parkway) into the existing landscaped median island east of the northbound (NB) 1-15 ramp intersection to create a second trap/storage lane for the left-turning traffic from westbound Scripps Poway Parkway to the southbound (SB) 1-15 entrance ramp.</li> </ul>		for their information. All State-owned signalized intersections were analyzed using the industry standard HCM methodology using Synchro software as required by the Caltrans, Guide for the Preparation of Traffic
B-3	<ul> <li>To minimize potential backup on the off-ramp and potential rear-end collisions at this intersection, begin the proposed right turn lane 100 feet beyond the end of curb return.</li> </ul>		Impact Studies, December 2002.
B-4	<ul> <li>The Synchro Worksheets were not provided for review as part of the EIR traffic analysis technical appendices. In order for Calitrans to properly review the EIR traffic analysis, a copy of the Synchro Worksheets should be provided for our Department's review. All State-owned signalized intersections affected by this project should also be analyzed using the intersecting lane vehicle (ILV) procedure from the Calitrans Highway Design Manual, Topic 406, page 400-21.</li> </ul>	B-5	A queue analysis for freeway on-ramps was provided in the TIA (Tables 1- 7 thru 1-9). Queue analysis for the off-ramp legs of the intersections was not provided because the projected LOS at these locations is anticipated to be acceptable with the project and mitigation. Excessive queues on the
B-5	<ul> <li>Given the proposed project's driveway proximity to the 1-15 NB off-ramp, as well as high volumes on the freeway off-ramps and Scripps Poway Parkway, a queue analysis for the freeway off-ramps should be provided as part of the EIR's traffic analysis to determine potential project impacts to the freeway.</li> </ul>		off-ramps has not been observed in the field and is unlikely in the future. Excessive queues resulting from the metered on-ramps and backing onto Scripps Poway Parkway were observed and are shown in tables 1-7 thru 1- 9 of the TTA Mitioarion in the form of additional storage is shown in
			Figure 1-1 (re-printed below as <i>Scripps Ponuy Parkway / 1-15</i> Reconfigurations
	"Caltrana improves mobility across California"		with response no. M-/) of the 11A and included as MM 5.2-1 in the EJK.

Response to Letters of Comment – Page 7 September 2013

Response to Letters of Comment – Page 8 The additional westbound turn pocket discussed in Comment B-2 and shown in Figure 1-1 of the TIA (re-printed below as Comment Noted. Signage will be coordinated with Caltrans when final Scripps Poway Parkway / I-15 Reconfigurations with response no. M-7) will leave a minimum of six feet in width in the median to maintain a safe LETTERS OF COMMENTS AND RESPONSES design for the improvements shown in Figure 1-1 of the TIA (re-printed below as Scripps Poway Parkway / I-15 Reconfigurations with response no. M-Comment Noted. If the project would install detectable warning surfaces, storage area for bicyclists and pedestrians. as required by Caltrans. Comment Noted. Comment Noted. Comment Noted. Comment Noted. Comment Noted. Comment Noted. 7) is completed. B-12 B-10 B-11 B-13 B-9 B-6 8-8 B-7 Please note, all changes to existing conditions with any potential impact to the States facilities, such as traffic signals, exit and entrance ramps, Americans with Disabilities Act (ADA) curb ramps, etc, must comply with all applicable State and Federal Standards. may impact pedestrians and bicyclists. The existing raised median provides a refuge area for bicyclists and pedestrians and includes enough space for most bicycles to store safely if they are Reduction in the width of the raised median on Scripps Poway Parkway east of the interchange caught between signal phases. With the additional west bound turn pocket that is required by Caltrans Traffic Operations, there should be enough width to accommodate a refuge area. A Per the Manual of Uniform Traffic Control Devices (MUTCD) Section 2B.18 Support 13c, install an Activated Blank-Out "R3-1" sign at the intersection of northbound 1-15 off ramp and Scripps Poway Parkway. Further coordination with Caltrans Traffic Electrical Do not use the term "maximum slopes" on plan sheets even if the maximum slope is flatter than the maximum standard (see DIB 82 Operations Policy Directive Number 12-01. "Change the Pedestrian Clearance Time to a Maximum Walking Speed of 3.5 feet per second. Also, if more than 50 percent of the south side of Scripps Poway Parkway. Therefore, it is recommended to install detectable revised median. All improvements must comply with all ADA requirements and Section 4450 of the California Government Code. Caltrans standards now require curb ramps with slopes flatter than the legal maximum to account for construction tolerances. Therefore, design for less than maximum slope, e.g., 1.5% instead of 2%, 7.5% instead of 8.33%, etc. Try to obtain a 50-inch width instead of the minimum 48-inch clear width standard. corner of this intersection. Also, it is recommended to provide Island Passageway at the minimum of six feet in width is recommended to maintain storage area for bicyclists and existing pedestrian signals or pedestrian pushbutton is being replaced, install Accessible Pedestrian Signal (APS) System and Pedestrian Countdown Timers at this intersection." provided. Please refer to 2010 Caltrans Standard Plans A88A and A88B Curb This location is within close proximity to residential homes and businesses. It is anticipated that there would be an increase of pedestrian traffic on both the north and warning surface, per Caltrans 2010 Standard Plan A88A, at the north and southwest For ADA curb ramps and island passageway, construction details should be Any modifications to the traffic signals will require implementing Caltrans Traffic Caltrans improves mobility across California" http://www.dot.ca.gov/hq/oppd/dib/dib82-04.pdf) Ramp and Island Passageway Details. and Signal Branch would be required. Mrs. Shearer-Nguyen pedestrians. Non-motorized April 17, 2013 0 ò 0 The Watermark Page 2 B-10 B-12 B-13 <del>В</del>-11 B-6 8-8 B-9 B-7

The Watermark Final Environmental Impact Report

sponse to Letters of Comment – Page 8 September 2013

	Mrs. Shearer-Nguyen April 17, 2013 Page 3		
B-14	<ul> <li>Caltrans Complete Streets Deputy Directive 64-R1 (DD64-R1) requires Caltrans to ensure that the needs of all users of the transportation system are accommodated in all phases of work performed within Caltrans right-of-way (R/W). Any proposed work to be performed in Caltrans R/W should be inclusive of all users of the transportation system.</li> </ul>	B-14	Comment Noted.
B-15	<ul> <li>The proposed bicycle facility should not be called a bike lane, per California Highway Design Manual (HDM) Chapter 1000, California Vehicle Code 21207, and California Streets and Highway Code section 891.</li> </ul>	B-15	Comment Noted.
B-16	The removal of the bike lane on Scripps Poway Parkway is not supported by Caltrans. The addition of the bike path on Scripps Poway Parkway does not absolve all comflict between non-motorized users and motor vehicles. The intent to remove bicycle traffic from eastbound (EB) traffic lanes on Scripps Poway Parkway in lieu of having bicyclists use the proposed path will not remove all bicyclists from eastbound lanes on Scripps Poway Parkway, rather than use the proposed path. Forcing bicyclists to use the path would require them to negotiate the conflicts at the ingress and egress to the Project, as well as freeway off-ramp traffic tuming right from the NB 1-15 off-ramp. The project should appropriately identify the impacts to bicyclists and pedestrians. It is assumed as part of Caltrans review that the City of San Diego's TSW/Transportation Engineering Operations, Multi-Modal Program has reviewed any proposed changes to the non- motorized circulation as they relate to this project.	B-16	Comment Noted. Based on an examination of the proposal for a channelized right-in/out-access point on Scripps Poway Parkway, a 12-foot non-contiguous bike/pedestrian path was determined to provide protection to bicyclists while serving to help eliminate the conflict of traffic weaving between right turn lanes and thru lanes on Scripps Poway Darkway. As a result, the existing bike lane on Scripps Poway Parkway in the eastbound direction will be removed and replaced with the bike/pedestrian path as shown in Figure 4-1 and Figure 4-1 of the EIR.
	General	!	
B-17	Please coordinate with Caltrans District 11 Landscape Architect and Maintenance for landscaping adjacent to freeway R/W.	B-17	Comment Noted.
B-18	Grading for this proposed project which would modify existing drainage and increase nunoff to State facilities will not be allowed.	B-18	Comment Noted. The Project has been designed so that runoff rates and durations are controlled to maintain or reduce pre-project flows.
B-19	If any work is performed within Caltrans R/W an Encroachment Permit will be required. Please refer to Caltrans Encroachment Permits Manual (http://www.dot.ca.gov/hq/traffops/developserv/permits/encroachment permits_manual/inde x.html).		downstream erosion conditions and protect stream habitat by the use of Hydromodification Controls. The City approved, June 25, 2012 Drainage Report calculations and conclusions prove compliance to Hydromodification Plan Controls There will be no off-site runoff onto
B-20	Current policy allows Highway Improvement Projects costing \$1 million or less to follow the Caltrans Encroachment Permit process. Highway Improvement Projects costing greater than \$1 million but less than \$3 million would be allowed to follow a streamlined project development process similar to the Caltrans Encroachment Permit process. In order to determine the appropriate permit processing of projects funded by others, it is recommended the concept and project approval	B-19	Caltans' facilities. Comment Noted.
	"Caltrans improves mobility across California"	B-20	Comment Noted.

The Watermark Final Environmental Impact Report

<b>B-21</b> Comment Noted.	<b>B-22</b> Comment Noted.	<b>B-23</b> Comment Noted.	<b>B-24</b> Comment Noted.	<b>B-25</b> Comment Noted.		
Mrs. Shearer-Nguyen April 17, 2013 Page 4 for work to be done on the State Highway System be evaluated through the completion of a Permit Engineering Evaluation Report (PEER). A PEER should always be prepared, regardless of the cost of improvements, when new operating improvements are constructed by the permittee that become	Part of the State Highway System. These include but are not limited to, signalization, channelization, turn pockets, widening, realignment, public road connections, and bike paths and lanes. After approval of the PEIR and necessary application and supporting documentation an Encoachment Permit can be issued. 2 Highway Improvement Projects greater than \$3 million, or considered complex projects, would be required to after a project and cooperative Agreements). A caltrans District responsible unit will be notified and a supporting documentation an Encoachment Permit can be issued.	In order to expedite the process for projects sponsored by a local agency or private developer, it is recommended a PEER be prepared and included in the Lead Agency's CEQA document. This will help expedite the Caltrans Encroachment Permit Review process. The PEER document forms and procedures can be found in the Caltrans Project Development Procedures Manual (PDPM). http://www.dot.ca.gov/hq/traffiopgd/pghm.htm http://www.dot.ca.gov/hq/traffiops/developserv/permits/pdff/forms/PEER_(TR-0112).pdf	A As part of the Encroachment Permit process, the applicant must provide an approved final environmental document including the California Environmental Quality Act (CEQA) determination addressing any environmental impacts within the Cattarans RVA, and any corresponding technical studies. If these materials are not included with the encroachment permit application, the applicant will be required to acquire and provide these to Calitans before the permit application, the applicant will be endurised. If then the reconcernent of measures will be a condition of the encroachment permit approval as well as procurement of any necessary regulatory and resource agency permits. Encroachment permit approval.	5 Additional information regarding Encroachment Permits may be obtained by contacting the Caltrans Permits Office at (619) 688-6158. Early coordination with Caltrans is strongly advised for all encroachment permits. If you have any questions, or require further information, please contact Marisa Hampton at (619) 688-6954.	Sinceroof Mark Control Chief JACOB M. ARMSTRONG, Chief Development Review Branch Development Review Branch	

The Watermark Final Environmental Impact Report

Response to Letters of Comment – Page 10 September 2013

	Comments noted. The Native American Heritage Commission was contacted in writing on February 16, 2007, in accordance with State Government Code § 65352.3. Tribes were given 90 days in which to request consultation; no consultation was requested. No comments were received during the 90-day consultation period. Additionally, local Native American Tribes were provided with notification of the availability of the draft EIR. In addition, archaeological resources are addressed in Section 5.9, Historiaal Resources (Arrbaeological Resources), of the EIR. As concluded in that section, no cultural resources have been identified on the project site. Additionally, the project site has been graded in accordance with previous project site in a completely altered state. The Watermark project site in significant impacts to historical resources.	
STATE OF CALFORNIA         Amoid Schwarzsensoger, Gormon           NATIVE AMERICAN HERITAGE COMMISSION         Amoid Schwarzsensoger, Gormon           Native American American American Amoid Schwarzsensoger, Gormon         Amoid Schwarzsensoger, Gormon           Native American A	<ul> <li>RE: SCH# 2010091079 CECA Notice of Completion, draft Environmental Impact Report (DEIR) - The Watermark Project; "located within the Miramar Ranch North Community Plan, City of San Diego; San Diego County. Catiforma</li> <li>Dear Ms. Shearer-Nguyen:</li> <li>Dear Ms. Shearer-Nguyen:</li> <li>The Native American Heritage Commission (NAHC) has reviewed the CECA Notice regarding the avereferenced project. In the 1965 Appellate Count decision (17) Col al App 3" Goly, the count held that the NAHC has juristicition and special expertise, as a state agency. over afficted matrixe American resources impacted by proposed projects, including archaeological places of religious significance to Native American Native American burial sites.</li> <li>The California Environmental Quality Act (CECA) sudelines 15064(b). To adequately comply with this provision and migrate project-related impacts on archaeological resources, its comply with this provision and migrate project-related impacts on archaeological resources, its comply with this provision and migrate project-related impacts on archaeological resources, its comply with this provision and migrate project related impacts on archaeological resources, its comply with this provision and migrate project related impacts on archaeological resources, its comply with this provision and which we known cultural resources are the impact of a part or all of the area of project effect (APE) has been previously surveyed for cultural resources. Which we known cultural resources recorded on or adjacent to the APE to life and the trant Environmental impact resources with the NAHC recommendation at a resources with we known cultural resources recorded on or adjacent to the APE to life and the trant Environmental impact resources recorded on or adjacent to the APE to life and the trant Environmental trans of the records search and field areavery. We suggest that this be conditioned and with the NAHC, if possible.</li> <li>The NAHC recommends that known cultur</li></ul>	

Response to Letters of Comment – Page 11 September 2013

Amold Setteraceases. Accenter Amold Setteraceases. Accenter Amold Setteration ansitivity, a certified archaeologist and a cuturally plan provisions for the disposition of recovered Native American human remains in their covery of Native American human human remains in their covery of Native American human	
state of calleonina state of calleonina State of calleonina scorptor Langoon ast scorptor Langoon ast scorptor Langoon ast scorptor Langoon ast scorptor Langoon ast scorptor as	

Response to Letters of Comment – Page 12 September 2013

|--|

Native American Contacts San Diego County March 4, 2013	Kumeyaay Cultural Repatriation Committee Steve Banegas, Spokesperson av 1095 Barona Road Diegueno/Kumeyaay Lakeside . CA 92040 sbenegas50@gmail.com (619) 742-5587 (619) 443-0881 FAX	Ewitapaary Tribal Office Will Micklin, Executive Director 4054 Willows Road Alpine , CA streot wmicklin@leaningrock.net (619) 445-6315 - voice (619) 445-9126 - fax	Ipay Nation of Santa Ysabel Clint Linton, Director of Cultural Resources P.O. Box 507 Santa Ysabel. CA 92070 cjinton73@aol.com (760) 803-5694 cjinton73@aol.com	Kumeyaay Diegueno Land Conservancy Mr. Kim Bactad, Executive Director 2 Kwaaypaay Court Diegueno/Kumeyaay El Cajon · CA 91919 guassacl@onebox.com (619) 445-0238 - FAX (619) 655-1008 - Office kimbactad@gmail.com	responsibility as defined in Section 7880.5 of the Maalth and Safety Code, of the Public Resources Code.	In regard to cultural resources for the proposed Impact Report (DEIR) for The Watermark Project; located in the County, California.
	Jamul Indian Village Raymond Hunter, Chairperson P.O. Box 612 Diegueno/Kumeys Jamul - CA 91935 jamulrez@sctdv.net (619) 669-4785 (619) 669-48178 - Fax	Mesa Grande Band of Mission Indians Mark Romero, Chairperson P.O Box 270 Diegueno Santa Ysabel, CA 92070 mesagrandeband@msn.com (760) 782-3092 Fax	Kwaaymi Laguna Band of Mission Indians Carmen Lucas P.O. Box 775 Pine Valley , CA 91962 (619) 709-4207	Inaja Band of Mission Indians Rebecca Osuna, Chairman 2005 S. Escondido Blvd. Diegueno Escondido · CA 92025 (760) 737-7528 (760) 747-8568 Fax	This list is current only as of the date of this document. Distribution of this list does not relieve any person of the standroy ( Section 50973,94 of the Public Resources Code and Section 5997,343.	This list is only applicable for contacting local Native Americans will SCH2010991070, CEOA Notice of Completion, draft Environmental Mirmar Ranch Community Plan Ares, City of San Diego, San Diego

Response to Letters of Comment – Page 14 September 2013

Native American Contacts San Diego County March 4, 2013 Inter-Tribal Cultural Resource Protection Council Frank Brown Coordinator 240 Brown Road Alpine - CA 91901 frankbrown6928@gmail.com (519) 884-6437	Kumeyaay Cultural Repatriation Committee Bernice Paipa, Yice Spokesperson 1095 Barona Road Diegueno/Kumeyaay Lakeside CA 92040 (619) 478-2113 (KCRC is a Coalituon of 12 Kumeyaay Governments) bp@lapostatribe.com	This list is current only as of the date of this document. Distribution of this list does not relieve any person of the statutory responsibility as dafined in Section 7080.5 of the Health and Safety Code, Section 5097.34 of the public Resources Code and Section 5097.96 of the Public Resources Code. This is only applicable for controlling local Native Americane with regard to cultural resources for the proposed SCH8201095107. ECIA Moles Compilerion, fait Environmental Impact Resources for the proposed Mirmar Ranch Community Plan Area; City of San Diego County, California.

Response to Letters of Comment – Page 15 September 2013

<text><text><text><text><text><text><text><text><text><text><text><text><text><text><text><text><text><text><text></text></text></text></text></text></text></text></text></text></text></text></text></text></text></text></text></text></text></text>				TTERS OF COMMENTS AND RESPONSES
<ul> <li>B.1 M. Nayar.</li> <li>D.1 The Destruction of Control of</li></ul>		Mathew Redrigues Exercise To Control         Image: Control         Image: Control         Image: Control           Mathew Redrigues To Control         Department of Toxic Substances Control         Image: Control         Image: Control           Mathew Redrigues To Control         Department of Substances Control         Image: Control         Image: Control           Mathew Redrigues To Control         To Control         Control         Image: Control         Image: Control           Mathew Redrigues To Control         To Control         Control         Control         Image: Control         Image: Control           Mathew Redrigues To Control         To Control         Control         Control         Image: Control         Image: Control           April 8, 2013         April 8, 2013         Entert         Entert         Control         Control           To Control         Control         Control         Control         Control         C		
<ul> <li>D-2 D-2 D-2 D-2 D-2 D-2 D-2 D-2 D-2 D-2</li></ul>	- -	NOTICE OF AVAILABILITY OF A DRAFT ENVIRONMENTAL IMPACT REPORT FOR THE WATERMARK PROJECT, (SCH No. 2010091079), SANDIEGO COUNTY Dear Mr. Nguyen: The Department of Toxic Substances Control (DTSC) has received your submitted Draft Environmental impact Report (ER) for the above-mentioned project. The following project description is statistic in your document. The Project is the Watemark Project. General Plan Amendment, Community Plan Amendment, Razon, Vesting Tentative Map, Planned Development Permit, Conditional Use Permit and a Street Vacation to change the current land use designation from IndustrialBusiness Park to Commercial. Residential Prohibited and remove the Prime Industrial Lands identification on a portion of the project site, a rezone of a portion of the project site from IPXZX1 to construct a mixed-used evelopment, the proposed project site from Prozens Gateway Court. In addition to new development, the proposed project includes existing development that is an exponsed CUP, Planned Industrial Danned Industrial Detention and Planned Development, Propp. (CLI)PIDID NO 501007, CUP NO.	D-1	Comments noted. These comments provide a summary of the proposed project description. No responses are necessary.
	D-2	<ul> <li>TA323/PDP No. 174234 Extension of Time). The 34.39-acre project site is located at 10137 Scripps Gateway Court and is designated Industrial/Business Park and the IPX2X1 zone within the Miramar Ranch North Community Plan area."</li> <li>Based on the review of the submitted document DTSC has the following comments:</li> <li>The EIR should evaluate whether conditions within the Project area may pose a threat to human health or the environment. Following are the databases of some of the regulatory agencies:</li> </ul>	D-2	Comments noted. The EIR evaluates whether conditions within the project area may pose a threat to human health or the environment in Section 5.13, <i>Health and Safety</i> , of the EIR. The EIR concluded that the project is not listed on a hazardous materials sites list; and, therefore, no impacts would occur.

Response to Letters of Comment – Page 16 September 2013
							Comments noted. The analysis contained in Section 5.13, <i>Health and Safety</i> , of the EIR indicates that that are no hazardous materials sites identified on the project site or within one-quarter mile of the site. No required	investigations and/or remediation is necessary. Comments noted. The analysis contained in Section 5.13, <i>Health and Safey</i> , of the EIR indicates that that are no hazardous materials sites identified on the project site or within one-quarter mile of the site. No required	nivesuganons anu/ or remembri is necessary.
							D-3	D-4	
<ul> <li>Mr. E. Shearer-Nguyen April 8, 2013</li> <li>Page 2</li> <li>National Priorities List (NPL): A list maintained by the United States Environmental Protection Agency (U.S.EPA).</li> </ul>	<ul> <li>EnviroStor (formerly calSites): A Database primarily used by the California Department of Toxic Substances Control, accessible through DTSC's website (see below).</li> <li>EnviroStor (formerly CalSites): A Database primarily used by the California Department of Toxic Substances Control, accessible through DTSC's website (see below).</li> </ul>	<ul> <li>Resource Conservation and Recovery Information System (RCRIS): A database of RCRA facilities that is maintained by U.S. EPA.</li> <li>Comprehensive Environmental Response Compensation and Liability Information System (CERCLIS): A database of CERCLA sites that is maintained by U.S. EPA.</li> </ul>	<ul> <li>Solid Waste Information System (SVIS): A database provided by the California Integrated Waste Management Board which consists of both open as ell as closed and inactive solid waste disposal facilities and transfer stations.</li> </ul>	<ul> <li>GeoTracker: A List that is maintained by Regional Water Quality Control Boards.</li> </ul>	<ul> <li>Local Counties and Cities maintain lists for hazardous substances cleanup sites and leaking underground storage tanks.</li> </ul>	<ul> <li>The United States Army Corps of Engineers, 911 Wilshire Boulevard, Los Angeles, California, 90017, (213) 452-3908, maintains a list of Formerty Used Defense Sites (FUDS).</li> </ul>	2) The EIR should identify the mechanism to initiate any required investigation and/or remediation for any site within the proposed Project area that may be contaminated, and the government agency to provide appropriate regulatory oversight. If necessary, DTSC would require an oversight agreement in order to review such documents.	3) Any environmental investigations, sampling and/or remediation for a site should be conducted under a Workplan approved and overseen by a regulatory agency that has jurisdiction to oversee hazardous substance cleanup. The findings of any investigations, including any Phase I or II Environmental Site Assessment Investigations should be summarized in the document. All sampling results in which hazardous substances were found above regulatory standards should be	
	ont.						ņ	4	

The Watermark Final Environmental Impact Report

Response to Letters of Comment – Page 17 September 2013

LETTERS OF COMMENTS AND RESPONSES	Mr. E. Shearer-Nguyen April 8, 2013 Page 3	clearly summarized in a table. All closure, certification or remediation approval reports by regulatory agencies should be included in the EIR.	<ul> <li>4) If buildings, other structures, asphalt or concrete-paved surface areas are being planned to be demolished, an investigation should also be conducted for the presence of other hazardous chemicals, mercury, and asbestos containing materials (ACMS). If other hazardous chemicals, lead-based paints (LPB) or products, mercury or ACMs are identified, proper precautions should be taken during demolition and the contaminants and incompliance with California environment reculations.</li> <li>4) If buildings, other structures, asphalt or concrete-paved surface areas.</li> <li><b>D-5</b> Comments noted. The project does not propose the demolition of structures, asphalt or concrete-paved surface areas.</li> </ul>	<ul> <li>5) Future project construction may require soil scontaminated, it must be properly disposed and not simply placed in another location onsite. Land Disposal Restrictions (LDRs) may be applicable to such soils. Also, if the project proposes to import soil is non-advertised and small structured samping should be conducted to ensure that the immortant scills frace of contamination.</li> <li>D-6 Comments noted.</li> <li>D-7 Comments noted.</li> </ul>	6) Human health and the environment of sensitive receptors should be protected during any construction or demolition activities. If necessary, a health risk assessor is demolition activities. If necessary, a health risk assessor is determine if there are should be conducted by a qualified health risk assessor to determine if there are have been, or will be, any releases of hazardous materials that may pose a risk to human health or the environment.	<ul> <li>7) If the project site was used for agricultural, livestock or related activities, onsite soils and groundwater might contain pesticides, agricultural chemical, organic waste or other related residue. Proper investigation, and remedial actions, if necessary, should be conducted under the oversight of and approved by a government agency at the site prior to construction of the project.</li> <li>D-8 Comments noted. The project site was not used for agriculture, livestock or related activities.</li> </ul>	<ul> <li>a) If it is determined that hazardous wastes are, or will be, generated by the proposed operations, the wastes must be managed in accordance with the proposed operations, the wastes must be managed in accordance with the california Hazardous Waste Control Law (California Hazardous Waste Control Regulations, Title 22, Division 4.3). If a determined that hazardous wastes area, the fallity should be generated, the fallity should be generated, the fallity should be generated, the fallity should be determined that hazardous wastes and would not use by the project would not generate hazardous wastes and would not use hazardous wastes obtain a United States Environmental Protection Number (yo contactions and the learning is to a profession of CUPA).</li> <li>b) f) formation action of the project would not generate hazardous wastes and would not use hazardous wastes and would not use hazardous wastes and would not use that hazardous wastes and would not use that hazardous wastes and would not use that harding storage or second action a state autorization from the learning attend and state autorization materials. Inaufing, storage or second clurk, autorization provention approvent and the determined that hard state autorization materials. Inaufing storage or second clurk, autorization from the requirement for autorization from your locading action approvent and the project would not generate hazardous wasters and would not use that hazardous wasters and would not use that hazardous wasters and would not use and in the project would not generate hazardous wasters and would not use that hazardous wasters and would not use thazardous wasters and would not use that hazardous wasters an</li></ul>	DTSC can provide cleanup oversight through an Environmental Oversight Agreement (ECA) for government agreement (VCA) for private parties, or a Voluntary Cleanup Agreement (VCA), please see
			D-5	D-6	D-7	D-8	D-9	01-0

-11 Comments noted.			
<ul> <li>Mr. E. Shearer-Nguyen</li> <li>April 8, 2013</li> <li>Page 4</li> <li>www.dtsc.ca.gov/SiteCleanup/Brownfields, or contact Ms. Maryam Tasnif- www.dtsc.ca.gov/SiteCleanup/Brownfields, or contact Ms. Maryam Tasnif- bubasi, DTSC's Voluntary Cleanup Coordinator, at (714) 484-5489.</li> <li>10) Also, in future CEQA document, please provide your e-mail address, so DTSC can send you the comments both electronically and by mail.</li> <li>If you have any questions regarding this letter, please contact me via e-mail Suryavanshi. Nirupma@dtsc.ca.gov, or phone (714) 484-5375.</li> </ul>	Sincerely, Nou Dur Suryavanshi Dr. Nirupma Suryavanshi Project Manager Brownfields and Environmental Restoration Program cc. Governor's Office of Planning and Research State Clearinghouse P.0. Box 3044 Sacramento, California 95812-3044 State clearinghouse@opr.ca.gov.	CEQA Tracking Center Department of Toxic Substances Control Office of Environmental Planning and Analysis P.0. Box 806 Sacramento. California 95812 Attn: Nancy Ritter Intiter@dtsc.ca.gov CEQA # 3723	

The Watermark Final Environmental Impact Report

Response to Letters of Comment – Page 19 September 2013

				This letter acknowledges compliance with the State Clearinghouse review requirements for draft environmental documents.							
WINNEL .		- 1		hat E-1		ų	eed	the			 
STATE OF CALIFORNIA Governor's Office of Planning and Research State Clearinghouse and Planning Unit Edmmd G BoomJr.	April 15, 2013	Elizabeth Shearee-Nguyen City of San Diage 1222 First Avenue, MS-501 San Diego, CA 92101	Subject: The Watermark SCH#: 2010091079 Dear Elizabeth Shearer-Nguyen:	The State Clearinghouse submitted the above named Draft EIR to selected state agencies for review. O, the enclosed Document: The review period closed on April 12, 2015, and the comments from the responding agency (res) is (arc) enclosed. If this comment package is not in order, please notify the Stat Clearinghouse immediately. Please refer to the project's ten-digit State Clearinghouse number in future correspondence so that we may respond promptly.	Please note that Section 21104(c) of the California Public Resources Code states that:	"A responsible or other public agency shall only make substantive comments regarding those activities involved in a project which are within an area of expertise of the agency or which are required to be carried out or approved by the agency. Those comments shall be supported by specific documentation."	These comments are forwarded for use in preparing your final environmental document. Should you ne more information or clarification of the enclosed comments, we recommend that you contact the commenting agency directly.	This letter acknowledges that you have complied with the State Clearinghouse review requirements for draft environmental documents, pursuant to the California Environmental Quality Act. Please contact th State Clearinghouse at (916) 445-0613 if you have any questions regarding the environmental review process.	Sinceredy, Scott Morgan Director, State Clearinghouse	Enclosures cc: Resources Agency rel. TEL (916) 445-613 FAX (916) 325-3018 www.opr.cs.gov	
				E-1					1		

The Watermark Final Environmental Impact Report

Response to Letters of Comment – Page 20 September 2013

orval of an Amendment to nerotanent to change the a General Plan on of the project site, a or (Scripps General), refeormencesinfregional), ins are described in detail in are described in detail
EIR Draft EIR To implement the Watermark project, the project applicant is requesting appr the Miramar Ranch North Community Plan and associated General Plan Am the Miramar Ranch North Community Plan and associated General Plan Am Amendment to remove the Prime Industrial Lands identification from a portion Reazone for a project step from P-2s-1 funduatiani-partyto-CRea- vesting Tanathore May, a PDP with Design Guidelines, a Street Vacation for and a CUP for a movie Ineader. The elements of these various project action in Section 3.0, Project Description, of this EIR. <b>y Contact</b> Elizabeth Shearer-Nguyen G19 of San Diego Fax 1222 First Avenue, MS-501 San Diego San Diego Sante CA Zip 92101

Response to Letters of Comment - Page 21 September 2013

<b>E-2</b> This letter is the same letter as C-1, above. Please see res				
STATE OF CALFORNIA MATIVE AMERICAN HERITAGE COMMISSION Scremmed under account and account account and account acco	RE: SCH# 2010091079 CEQA Notice of Completion; draft Environmental Impact Report (DEIR) – "The Watermark Project;" boated within the Miramar Ranch North Community Plan; City of San Diego; San Diego County , California Dear Ms. Shearer-Nguyen:	The Native American Heritage Commission (NAHC) has reviewed the CEQA Notice regarding the above referenced project. In the 1985 Appellate Court decision (170 Cal App 3 <sup>rd</sup> 504), the court held that the NAHC has jurisdiction and special expertise, as a state agency, over affected Native American resources impacted by proposed projects, including archaeological places of religious significance to Native Americans, and to Native American burial sites.	The California Environmental Quality Act (CEQA) states that any project that causes a substantial adverse change in the significance of an historical resources, which includes archeological resources, is a splittrain effect requiring the preparation of an EIR (CEQA guidelines 15064(p)). To adequately compty with this provision and mitigger project-related impacts on archaeological resources, the Commission recommends the following actions be required:	<ul> <li>Contact the appropriate Information Center for a record search to determine:</li> <li>If a part or all of the area of project effect (APE) has been previously surveyed for cultural resources, which we know that it has.</li> <li>The NAHC recommends that known cultural resources recorded on or adjacent to the APE be listed in the draft Environmental impact Report.</li> <li>If an additional archaeological investory survey is required, the final stage is the preparation of a proflessional report detailing the findings and recommendations of the records search and field survey. We suggest that this be coordinated with the NAHC, if possible.</li> <li>In the final report containing site forms, and associated functary objects should be submitted fitmediately to the planning department. All information regarding site locations. Nature American human remains, and associated functary objects should be to a submitted fitme defanction.</li> <li>Contract has been made to the the Native American humal registion for a confidential addenoum. and not be made available for public discource pursuant to California Government Code Section 6254.10.</li> <li>Contact has been made to the the Native American function regarding site locations. Native American function and not be made available for public discource pursuant to California Government Code Section 6254.10.</li> <li>Contact has been made to the the Native American Jentitage Commission for a separate been provided and is attached to his latech.</li> <li>A list of appropriate Native American Contacts for consultation concerning the project site has been made to the the Native American submitsion for a second search and admittent and the admittent and the American structures for a courd-breaking activity begins. If that occurs, the NAHC suggests that information for the order discoveries be coordinated minit to Native (CEOA) accidential discovered archeological resources, per California Environmental Quality Act (CEOA) accidentally discovered archeological resources, pe</li></ul>

Jouan		rally bing	ed their s Code human					2		
Arnold Schwarzenegger, Gol	~	chaeologist and a cultu onitor all ground-disturi	e disposition of recover ican human remains in ), and Public Resource dental discovery of any							
		nsitivity, a certified arr resources, should m	alan provisions for the attive Americans. overy of Native Ameri 5, CEQA §15064.5(e) the event of an accid atery.	y. Bieton	3-625					
	COMMISSION	archaeological ser owledge in cultural	In their mitigation pulturally affiliated N provisions for disco afety Code §7050.4 is to be followed in a dedicated ceme	Sincerel	(916) 65 Contacts list					
	AN HERITAGE	reas of identified merican, with kno	s should include insultation with cu s should include i 1. Health and Sa idates the proces idates the proces		iringhouse lative American C					
	0 2 8	In ar	jencie; in col jencies on plan 8 man 18 man		ate Clea nt: N					

Response to Letters of Comment – Page 23 September 2013

	F-1 Comments noted.
San Diego County Archaeological Society, Inc. Environmental Review Committee Environmental Review Committee 14 March 2013 14 March 2013 15 March 2013 16 March 2013 17 March Services Department City of San Diego 1222 First Avenue, Mail Station 501 Subject: Draft Environmental Impact Report The Watermark Project No. 180357	F1 In the Nationary Archaeological Society. In the reviewed the cultural resources aspects of the subject DMND on behalf of this committee of the San Diego County Archaeological Society. In the information contained in DEIR, we agree that no impacts to cultural resources an inguiton measures are sources are likely and, therefore, that no outhard resources an inguiton measures are sources are likely and, therefore, that no outhard resources an inguiton measures are sources are likely and, therefore, that no outhard resources an inguiton measures are sources are likely and, therefore, that no outhard resources and therefore, that no outhard resources are likely and therefore. In the City's environmental review process for this process for the courter and the city's environmental review process for the courter and the city's environmental review process for the courter and the city's environmental review process for the courter and the city's environmental review process for the courter and the city's environmental review process for the courter and the city's environmental review process for the courter and the city's environmental review process for the courter and the city's environmental review process for the courter and the

Response to Letters of Comment – Page 24 September 2013

й с с с с с	The second of the declaration of the stretch of SD Development Services Center mer, City of SD Development Services Center mercany in the antice mercass of specific comments regarding the adequacy of The Watermark DER - project No.         G-1         This comments necessary.           MS SQ1         Comments regarding the adequacy of The Watermark DER - project No.         G-2         Comments no.           MS SQ1         Comments regarding the adequacy of The Watermark DER - project No.         G-3         As describid and more access with the adding the adding the antice access with the street system based on the table presented under Thresholds above?         G-3         As describid and with the antice access with a digracent to the astron at the street system based on the table presented under Thresholds above?         Comments nect with a digracent to the astron at the street system based on the street sys
-------------	--

50		
5	$\sim$	
Φ	ò	
g	2	
P	0 G	
1	ž	
Jt	Ð	
ē	bţ	
Ξ	Se	
Ξ		
2		
0		
ō		
LS		
ē		
et		
_		
5		
Φ		
ns		
0		
Sp		
Ø		
<u>_</u>		

-	1
S	
ПĹ	
0)	
Z	
Q	
<u>с</u>	
$\overline{\mathbf{\Omega}}$	
ŭí	
щ	
$\mathbf{M}$	
$\cap$	
$\equiv$	
2	
$\triangleleft$	
~	
0)	
$\vdash$	
-	
<u></u>	
ш	
$\geq$	
$\geq$	
$\cap$	
$\mathbf{X}$	
$\bigcirc$	
ш	
$\cap$	
~	
S	
Ń	
ш	
in	
<u> </u>	

~1	
22	
-	
-	
c	
ā	
ē	
c	
e	
c	
0	
<u>ה</u>	
~	

G-6 In section 5.2, Transportation/Traffic Circulation/Parking, 5.2.2 Impact Analysis, Issue 3 Would the project result in a change in traffic patterns, including either an increase in traffic levels or a change in location that results in substantial safety risks? G-7 (2a) The proximity of elementary school aged children walking to school, relative to the project site location, is of great concern. The increased traffic volumes on Scripps Poway Parkway will impact 3 elementary schools and 1 park, all within a few miles from the project site.

The 3 Scripps Ranch, San Diego Unified School District, schools reside east of the project site. The schools are:

- Innovations Academy (corner of Spring Canyon Rd and Scripps Poway Parkway)
  - Dingeman Elementary (on Scripps Creek Drive)
- Ellen Browning Scripps Elementary (on Cypress Canyon Rd)

At all three of these schools, children utilize the crosswalks on Scripps Poway Parkway.

Currently, at all Scripps Ranch schools, AM drop-off and PM pickup generate substantial frustration to drivers, as well as safety risks to the children, neighboring residents and commuters. These school facilities have very limited parking for events, AM/drop-off or PM/pickup. The traffic congestion permeates into surrounding neighborhoods roadways. Spring Canyon Park is located on Scripps Poway Parkway within the Spring Canyon Rd/Scripps Creek Dr roadway segment. Spring Canyon Park is jointly used by Dingemen Elementary for withites and is a prime location for family AM drop-off and PM pickup. Unfortunately, I have withessed parents stopping along Scripps Poway Parkway to allow children to enter or exit vehicles.

- (2b) There would be a change in traffic pattern for SR residents who reside on and nearby G-8 Scripps Highlands Drive, Trail Crest Drive, Winding Ridge Road and Farmingdale Street. As congestion and volumes increase near the project site, commuters and local residents will utilize this alternative route. This is currently a known problem with MedImpact employee traffic. Employees want to avoid Scripps Highland Drive and Scripps Poway Parkway and seek alternative access than to Awile Bivd to Mira Mesa Bivd.
- G-9 (2C) An increase in traffic levels, with the proposed project, might only be due to cars and smaller trucks/vans but the added volume, presents safety risks due to the "type of vehicles" that utilize Scripps Poway Parkway. Scripps Poway Parkway is a major truck route between I-15 and R-67 and can consist of many large medium and heavy trucks.

Medium trucks can include; box, platform, flatbed trucks and recreational vehicles. Heavy trucks can include; concrete transports, cranes, dump trucks, log carriers, semi-trailer/18-wheelers trucks.

- **G-6** This comment re-states the Issue 3 question, presented in Section 5.2, *Transportation/Traffic Circulation/Parking*, of the EIR. No response is necessary.
- **G-7** All three schools referenced in this comment are located east of the project site, as shown in the aerial photo on the following page. Innovations Academy is located about 0.75 mile to the east, Dingeman Elementary School is about 1.0 mile east of the project site, and Ellen Browning Elementary School is located even farther east, approximately 1.5 miles from the project site.

As presented in the EIR, the proposed project would result in an increase in traffic travelling east of the project site. Some trips could be generated from residential neighborhoods east of the project site due to residents working at the project site or using the shops, restaurants, and other services and amenities that the project would provide. Based on the Traffic Impact Analysis prepared for the project, 36 percent (or 7,743 trips) of the traffic generated by the project would travel to/from the east. Increased traffic volumes would not result in a significant impact to safety for children walking to school along Scripps Poway Parkway. Sidewalks separated by a landscaped parkway are provided along Scripps Poway Parkway. Additionally, there are many opportunities to cross at signalized intersections with protected crosswalks. **G-8** Anticipated project traffic is shown in Figure 3-2 of the TIA and in Figures 5.2-4a and 5.2-4b of the EIR. As shown in these figures, little or no traffic from the project is expected to travel south from the project site on Scripps Highlands Drive, Trail Crest Drive, Winding Ridge Road and Farmingdale Street. As discussed in this comment, a certain amount of traffic may avoid the adjacent Scripps Poway Parkway/I-15 Interchange attempting to access I-15 via the Mira Mesa Boulevard Interchange.

LETTERS OF COMMENTS AND RESPONSES	As shown in Figure 3-1 of the TIA and Figures 5.2-3a and 5.2-3b, this traffic is expected to travel on Spring Canyon Road/Scripps Ranch Boulevard to reach Mira Mesa Boulevard. This is consistent with experience from travel time surveys during peak hours completed by the applicant. Based on these surveys, the travel time from the project site to the Mira Mesa Boulevard/I-15 interchange was about half a minute to one minute shorter when traveling via Spring Canyon Road/Scripps Ranch Boulevard versus the routing mentioned in this comment.	The shorter travel time experienced when leaving the project site is consistent with the design of these roadways as Major and Prime Arterial roadways with major signalized intersections. Traveling through adjacent neighborhoods on Scripps Highland Drive, Trail Crest Drive, Winding Ridge Road and Farmingdale Street is not only less convenient but would force a driver through a circuitous routing on winding local residential streets. Additionally, the actual distance traveled via Scripps Highland/Trail Crest is 3.04 miles while traveling via Spring Canyon/Scripps Ranch Boulevard is 3.42 miles. Based on the higher speed limits and travel speeds prevalent on the Major/ Prime Arterial roadways, the travel time survey results are consistent with the results which would be expected.	The shorter travel time of the Spring Canyon Road/ Scripps Ranch Boulevard route are consistent with the travel patterns shown in the TIA. It is unlikely that residents adjacent to Scripps Highland Drive, Trail Crest Drive, Winding Ridge Road and Farmingdale Streets would experience significantly higher traffic volumes in their neighborhood as a result of the project. However, these residents would experience increased traffic on Scripps Poway Parkway and adjacent to the project site as shown in the TIA and summarized in the EIR.	<b>G-9</b> Comments noted. The TIA prepared for the project includes an assumption for truck traffic based on the classification of the roadways affected and the typical amount of truck traffic those roads carry. Therefore, truck traffic is included in the traffic generation for the project, as well as part of the traffic volumes in the surrounding community.	

Response to Letters of Comment – Page 28 September 2013



			TERS OF COMMENTS AND RESPONSES
G-10	Comment #3 In section 5.2, Transportation/Traffic Circulation/Parking, 5.2.2 Impact Analysis, Issue 6	G-10	This comment re-states the Issue 6 question, presented in Section 5.2, <i>Transportation/Traffic Circulation/Parking</i> of the EIR. No response is necessary.
	Would the project conflict with adopted policies, plans, or programs regarding public transit, bicycle, or pedestrian facilities, or otherwise decrease the performance of safety of such facilities?	G-11	As shown in the TIA and presented in Section 5.2, Transportation/Traffic
G-11	(3a) The added vehicle volume (ADT = 18,000/22,000), and loss of capacity and operating conditions, at several project intersections (LOS E/F) increases the hazards for bicycle and pedestrian users in the neighboring areas.	0.1.1	<i>Cirvalation/Parking</i> , of the EIR, the project would result in a significant increase in traffic at several locations. However, with respect to intersections, the project would mitigate all impacts. Additionally, the
G-12	Comment #4 In section 5.2, Transportation/Traffic Circulation/Parking, 5.2.2 Impact Analysis, Issue 7 Would the project result in: An increased demand for off-site parking? Effects on existing parking?		project would be adding a non-contiguous project would increase traffic, it is Scripps Poway Parkway. Although the project would increase traffic, it is not anticipated that any pedestrian and bicycle hazards would result based on unacceptable intersection delay or LOS.
G-13	(4a) The project proposes a movie theater onsite. Some customers might opt for parking along Scripps Highland Drive or Trails Crest Drive to avoid the mass exit from the parking structure after show times.	G-12	This comment re-states the Issue 7 question, presented in Section 5.2, Transportation/Traffic Circulation/Parking, of the EIR. No response is necessary.
G-14	(4b) Does the project have a drop-off and pickup area movie theater customers? Onsite parking circulation could be impacted and/or off-site parking areas could become drop- off/pickup locations.	G-13	The proposed project would meet the City's parking requirements, and adequate parking would be provided for all proposed uses. Additionally, as with most theaters, movie times would vary, and patrons would come
G-15	Comment #5 In section 5.3, Visual Effects and Neighborhood Character, 5.3.1 Impace Analysis, Issue 3 Would the project result in: Substantial change in the existing landform? The creation of a negative aesthetic site or project?		and go as movies begin and end. The theater would be located proximate to the parking structure, where parking would be provided to serve movie patrons. While street parking may be available, moviegoers would need to walk across the entire development to access any available street parking.
	Project bulk, scale, materials, or style which would be incompatible with surrounding development? Substantial alteration to the existing or planned character of the area, such as could occur with the construction of a subdivision in a previously undeveloped area? Note: For substantial alteration to occur, new development would have to be of a size, scale, or design that would markedly contrast with the character of the surrounding area.		It should be noted that, although Scripps Highlands Drive forms the project site's eastern border, Trails Crest Drive is located farther east of the project. Walking distance between the proposed theater and available street parking on Trails Crest Drive is about 0.3 mile.
G-16	(5a) With the (Area A) project size projected to be 600,000 square feet of retail/office buildings in an area of 23 acres, what is the consideration for hard surfaces versus vegetated surface (HS vs. VS)? Do neighboring SR retail/office sites have similar HS vs. VS ratio values? If not this could restare an actived for character with the surrounding acres.	G-14	There will be a drop-off area for the theater on the lid of the parking structure on the east side of the office building in this location.
		G-15	This comment re-states the Issue 3 question, presented in Section 5.3, Visual Effects and Neighborhood Character, of the EIR. No response is necessary.

	Response to Letters of Comment – P	Septembe
	The Watermark	Final Environmental Impact Report

<b>G-16</b> As presented in Section 3.0, <i>Project Description</i> , of the EIR, the proposec project would provide for an approximately-1.7 acre plaza space designed to accommodate community gatherings and events. This space encompasses a grand lawn/event area, edged with paving and background planting (Figure 3-11, <i>Primary Public Plaza Landscape Design</i> ). Seating opportunities, as well as a children's play area, water feature, foca elements, and thematic elements, would be located surrounding the granc lawn/event area.		

project site was surveyed on September 13, 2011, by REC biologists between 10:00 a.m. and 3:00 p.m. The presence of the on-site woodrat species could potentially be the sensitive desert woodrat, according to their range description ....? The sensitive desert woodrat referred to in the footnote of Table 3 is the San Diego Woodrat (Neotoma lepida intermedia), which is listed as a Species of Special Concern by the California Department of Fish and Wildlife. The San Diego Woodrat is not listed as endangered; and the commentor's notation that the report states the "endangered nocturnal desert woodrat, which is protected under the Endangered Species Act 1973" is in error. Although San Diego Woodrats significant. The off-site area adjacent to the proposed parking structure is Caltrans' right-of-way and located between I-15 and the graded pad where the southern perimeter of the project site. Project lighting would be regulated by the City's Municipal Code (Section 142.0740). The purpose by unnecessary illumination. Regulation of outdoor lighting is also minimize light trespass and to direct, shield, and control light to keep it 142.0740(c)(6), on properties which are adjacent to or contain sensitive This comment re-states the Issue 3 question, presented in Section 5.8, Table 3 within the Biological Technical Report identifies that a woodrat was observed on the project site during biological resource surveys. The was observed during these hours. The footnote to Table 3 states "This (daytime) activity near their den sites. No observations were conducted at Indirect impacts to off-site open space areas would not be considered he Watermark project would be located. Open space areas occur along of the City's Outdoor Lighting Regulations is to minimize negative impacts from light pollution including light trespass, glare, and urban sky glow in intended to promote lighting design that provides for public safety and The regulations require lighting fixtures from falling onto surrounding properties, with the goal that zero directbeam illumination leaves the premises. Per Municipal Code Section exhibit nocturnal foraging behavior, they are also known to exhibit diurnal order to preserve enjoyment of the night sky and minimize conflict caused biological resources, any exterior lighting shall be limited to low-level ights and shields to minimize the amount of light entering any identified Biological Resources, of the EIR. No response is necessary. sensitive biological resource areas. conserves electrical energy. night. G-18 G-19 G-17 (6a) The woodrat was observed in the coastal sage scrub chaparral (CSS/C) habitat area directly (6b) As discussed in the Feb 2013 MRNPC meeting minutes, under "What improvements, if any, large parking structure proposed adjacent to the corridor, could impact the wildlife species that Parkway?" measures to reduce darkness/glare/add lighting in this corridor, particularly with a sensitive, or special status species in the MSCP or other regional plans, policies, or regulations, or by the California Department of Fish and Wildlife (CDFW) or U.S. Fish and Wildlife Service (USFWS)? endangered nocturnal desert rat, which is protected under the Endangered Species Act 1973, bordering the project site on the southeast end. It was noted that this species could be the Would the project directly or indirectly impact any species identified as a candidate, will be made to the N-S bike lane right of way adjacent to I-15, south of Scripps Poway according to their range description. Were observations made during the night? In section 5.8, Biological Resources, 5.8.2 Impact Analysis, Issue 1: utilize the CCS/C and SCSS open space areas. Thank you for your time in this matter. SR Schools Committee member MRNPC resident member 11223 Walking Fern Cove Lorayne Burley San Diego, CA 92131 witte@san.rr.com Comment #6 G-18 G-19 G-17

LETTERS OF COMMENTS AND RESPONSES

The Watermark Final Environmental Impact Report

Response to Letters of Comment – Page 32 September 2013

<text><text><text><text><text><text><text><text></text></text></text></text></text></text></text></text>		Shearer-Nguyen, Elizabeth			
<text><text><text><text><text><text><text></text></text></text></text></text></text></text>		From: Jennifer Cesaro [jenniwinston@yahoo.com] Sent: Jensidey, March 28, 2013 4/07 PM To: Desclent@ScrippsRanch.org Subject: The Watermark - Project #180357			
1.1 Can be determined and the determined of t		To Whom It May Concern:			
I demonstrate and special constants. Right non-use place for come together during the forear that are concentered about fraction. Right non-use a place for come together during the set are a concentered about fraction. The synchronic matter is the black concentrated at the work times for all of us. So as far as traffic goas, this is the black concentrated at the work times for all of us. So as far as traffic goas, this is not black concentrated at the work times for all of us. So as far as traffic goas, this is the black concentrated at the work times for all of us. So as far as traffic goas, this is not black concentrated at the work times for all of us. So as far as traffic goas, the special time would be doed a goometh foot menter foot menter. The adverse work is up in 4.5 Remot. That a shows on the far strift goas is the black concentrated at the work of be doed a goomethy should be pacind at the work of be apound to the adverse times. When we notified the sort doed a goomethy should be pacind at the source of the adverse times in the source added a goomethy concent at the work of be pacing at the source and a goomethy and we concent at this work of be pacing to the source and the goometh adverse and for a dofformation. The source add have strated pacing are were the source of the adverse and the source and a round be goometh. There planes paces, as 1 thick great things could come from it. Brank I, located foot that is an plane protection and the term at the strated and a round be again as a littick great things could come from it. Brank I, located by when the strates and from be some codes and take a strate and a round be address and from be some codes and take a strate and a round be address and the protect and take a strate and a round be goometic. Brank I, release and the source address and the pacing at the source address and the source addre	Н-1	I wanted to express my interest in this project. I have been extrem beginning. I think the addition of this high end shopping center wil Community. I believe it will even increase the value of the houses beautiful upscale neighborhood and this addition is just what it nee perception. My husband and I have lived in Scripps for 7 years no always fet like it was missing something. Anytime we want to go to travel to other areas of San Diego. And of course no one ever c neighborhood. Why would they? There's nothing to do here, unle chain or fast food establishments.	ely excited about this from be so great for the Scripps Ranch inthe area. Scripps Ranch is a des to keep and build upon that v. We love it here, but it has out to eat or go shopping, we have omes out to meet us in our st they want to eat at one of the	Т- 1	Comments noted. These comments do not address the completeness or adequacy of the EIR. No responses are necessary.
the other metric motion of the information of the structure of the structu		I even envision this center to be a place that the community will be holidays and special occasions. Right now we don't really have a	able to come together during place for gatherings.		
Iverant to say that one of the first things that peaked my interest with this project was the proposal to adsomicity interest. We have been performed as a surprise close state interest. We have been performed as the advective performance in the state and the exclusion is a unit of states. The matter state close state is a more such that are started and the state interest. Now it was performed to a major state performance in the state is a more close state in the analytic resolution about its part of the state interest. Now it would be advected the performance in the state interest into a state into a material in our registromore done of the state interest. The state event has a support exploration and resident and interest into a value be advected. Name, which is part of the state into a resolution and interest into a state in		For those that are concerned about traffic, my thought is, with a sh dispersed throughout the day. If they were to put office buildings i concentrated at the worst times for all of us. So as far as traffic go	opping center, traffic will be the space, traffic would be es, this is the better option.		
Lastly, I look forward to having some non-chain or possibly higher end chain restaurants in my area that I could enjoy without having to transing the plans, it sounds the saving to transing the transing to transing to transing the arounds with water features and nice architecture. Why husband and I really hope that these plans pass, as I think great things could come from it. Sincerely, Jennifer Cesario		I want to say that one of the first things that peaked my interest will add a gourmet food market. That alone would be absolutely fanta close by. I believe the closest we have is Jimbos, which is up in 4 grocency shopping and we only shop there if we are in the area. To something similar in our neighborhood would be perfect. I know it is an upper class neighborhood and residents are willing to pay the healthier options. So no doubt about it, this would be a hit.	h this project was the proposal to stic. We have nothing of the sort stanch. That is quite a drive for think that we could have vould do well also. Scripps Ranch higher price tags that come with		
My husband and I really hope that these plans pass, as I think great things could come from it. Sincerely. Jennifer Cesaro		Lastly, I look forward to having some non-chain or possibly higher that I could enjoy without having to travel far. Additionally, from wh plans, it sounds like this is a place that I can go grab a nice lunch stroll around the grounds with water features and nice architecture	end chain restaurants in my area at I understand after hearing the ind maybe some coffee and take a		
Sincerely. Juntier Ceearo		My husband and I really hope that these plans pass, as I think gre-	at things could come from it.		
Jennier Cesaro		Sincerely,			
		Jennifer Cesaro			
		1			

Response to Letters of Comment – Page 33 September 2013

Comments noted. These comments do not address the completeness or adequacy of the EIR. No responses are necessary.	
<u> </u>	
From: Cesaro, Michael [Audatex - Americas] [mailto:Michael.Cesaro@audatex.com] Sent: Sunday, April 14, 2013 4:13 PM To: DSD EAS Co: Presidentes Co: Presidentes Co: Presidentes Subject: The Watermark - project #180357 To Whom It May Concern: To Whom It May Concern: If mwriting to express my interest in the Watermark Project. I believe this development with the gournet foor market, upscale shops, movie theater and restaurants, is just what Scripts Ranch needs. If these proposed plans pass, it will be a benefit to the community, i'm really excited. My only complaint is that 3 years sounds to long to wait! Michael Cesaro	Sant from my Phones
Ξ	

Response to Letters of Comment – Page 34 September 2013

LETTERS OF COMMENTS AND RESPONSES		up the hill (at the end of bavelived here since       J-1       Comments noted. These comments do not address the completeness or until the Medimpact         adequacy of the EIR. No responses are necessary.       adequacy of the EIR. No responses are necessary.         te walking breaks up our ive, and more traffic is il to stop a the stop sign at bur comer who has given       J-1       Comments noted. The EIR evaluates traffic impacts associated with the protect in Section 5.2. Transportation/Traffic Circulation/Parking. and	e proposed site and reased noise level at that size of the proposed         concludes that the project would result in significant direct and cumulative impacts on roadway segments and intersections in the community. Mitigation measures would reduce all project-related traffic impacts to below a level of significance except for roadway segments on common 15 and common 2000.	ill undoubtedly occur posedexit at Scripps o our reighborhood. I - 3 Please see response no. G-9, above. h our neighborhood watch w more since then.	The Watermark project would be served by two access points. One access is backed up from the left point is a direct right-in/right-out off Scripps Poway Parkway. A second or the patrons to enter the patrons to enter the patrons to enter the patrons to enter the size of the buildings by and the size of the buildings by and to handle such a size of the buildings by and to handle such a size of the buildings by and to handle such a size of the buildings by and to handle such a size of the buildings by and to handle such a size of the buildings by and to handle such a size of the buildings by and the size of the buildings by and to handle such a size the provided from Scripps Poway Parkway to the Scripps Highland Drive. All of these access points into the project have been designed to operate efficiently with a minimum of queuing. Unlike the Mira Mesa Boulevard shopping center, right turns. The distance from Scripps Highland Drive. In a minimum of queuing. Unlike the Mira Mesa Boulevard shopping center, right turns. The distance from Scripps Highland Drive to the I-15 NB off-ramp is approximately 950 feet, and the expected queue would be approximately 790 feet.
	From: Clements, Kevin [mailto:kclements@corelogic.com] Sent: Friday, April 12, 2013 3:05 PM To: DSD EAS Cc: President@scrippsranch.org Subject: The Watermark - Project #180357"	We are residents of the Tiempo housing development which is just Scripps Highland and Trailcrest) from the MedImpact buildings. We 2002and the community has always been fairly quiet and peaceful, buildings opened a few year ago. Now we have employees that tak hill will some take smoking breaks at the top of Scripps Highland Dr around our neighborhood. In recent months, several drivers will fa the top of Trailcrest and we have often had a policeman posted at o out tickets to these drivers.	During the month of October there is usually a pumpkin patch at th parking will often extend up Scripps Highland Drive. There is an inc time as well. The size of this pumpkin patch is just a fraction of the Watermark site and should be an indication of the increased traffic as a result of this proposed retail center.	I have several concerns, one of which is the increased traffic that w around our neighborhood and through our neighborhood. The pro Gateway and Scripps Highland will exit into the street that goes into Although planners may think that this is a solution to route traffic th congest and cause more traffic through our neighborhood. Throug we heard of several attempted break-ins over the holidays, and a fe	While there are traffic reports being done, a comparison can be don Mira Mesa Blvd, where on a Friday or Saturday evening, the traffic at ou turn lane into the exit lane on the freeway. This type of traffic at on such a large shopping complex that does not have adequate room f and exit. Another cause for concern is that the developer wants to increase t more than 30 feet above the 60 foot limitation. The area is not me
			J-2	с. Г	

Response to Letters of Comment – Page 35 September 2013

Permit, to allow for design and architectural elements, and to accommodate the parking garage. The proposed deviation from the 60 foot maximum structure height of the CR-2-1 zone would allow the office space as envisioned in the existing planned industrial development permit (CUP/PID No. 1027). This deviation is necessary to maintain the deviation for structure height on the plaza (proposed Lot 9) enables the cell tower, or other architectural features to designate the gateway to the This would assist in achieving an objective of the Community Plan to develop the Mercy Interchange area as an attractive would allow for the construction of a parking garage to accommodate surface parking, providing for more public amenity open space. The Guidelines Sections 15064(e) and 15131, economic or social effects of a unless a physical change in the environment is caused by the economic or Comments noted. These comments do not address the completeness or As explained in Section 3.0, *Project Description*, and evaluated in Section 5.1, Class A office space approved under the existing Planned Industrial Watermark project to proceed with development of additional Class A existing vested development rights under that approval. The proposed development of a landmark architectural feature, such as a clock tower, gateway to the community. The height deviation for the parking garage parking needs for the project. This would reduce land required for wayfinding, and the creation of a community gateway through landmarks; the height deviation for the parking garage; and the height deviations for project buildings would not represent a significant environmental impact and would not be out of character with adjacent and nearby Relative to diminishing property values, in accordance with CEQA project shall not be treated as a significant effect on the environmental, social effects of a project; or if the economic and social effects resulting from a physical change in the environment create a significant effect. These comments do not address the completeness or Land Use, the project proposes deviations to allow for development of the proposed height deviation for the purpose of architectural detail, Neither of theses situations pertains to the proposed project. accuracy of the EIR. No responses are necessary. idequacy of the EIR. Comments noted. developments. community. J-6 <u>۲</u> J-7 Parkway have at least 2 - 3 vacancies in each of these shopping areas. Many businesses have understanding and zoning limitations that the max height of the buildings would be limited to 60 feet and to increase the allowable height by 50% will severely diminish the property values property valuations will have an adverse effect on the collectable tax revenues for the city as movie theater with a bowling alley down the street from our community. This will only serve Baskin Robbins in the Vons shopping plaza that has changed hands a few times and has since closed down. Scripps Ranch does NOT need a high end shopping center, let alone a high end Thank you for respecting the desires of the residents of Scripps Ranch over the developers of The residents of the Tiempo housing development don not want to see an increase in traffic, to increase traffic not only during rush hour, but during evening and weekend hours as well. While the residents realize that something is going into this area, a shopping area will invite unnecessary and troublesome traffic during most of the day and night hours, 7 days a week, versus a office type environment with more traditional business office hours in addition to the residents will undoubtedly have the properties reassessed which will cause the city to on top of the devaluation that results from the shopping center in and of itself. Reducing come and gone because the community could not support them. A prime example is the crime, noise property devaluation and tax revenue loss as because this undesired and ill There are a proposed number of 55 shops, while the shopping areas on Scripps Poway increase. Additionally, we made sizeable investments in our properties under the resident property de-valuations and loss of tax revenues to the city of San Diego. adjust the tax basis and thus reduce collectable monies. Kevin Clements VP Business Development CoreLogic Credco the Watermark Project. thought retail center. Kevin Clements Respectfully, cont. J-5 )-6 7-7

LETTERS OF COMMENTS AND RESPONSES

S		
<b>ISI</b>	l	
õ	l	
SP	l	
Ш	l	
	l	
N	l	
S	l	
F	l	
ΛEI	l	
M	l	
0	l	
	l	
ō	l	
RS	l	
Ë		
F		
<u> </u>	1	

		<b>K-1</b> Access from Scripps Ranch (east of the project) to the project wo provided at the intersection of Scripps Poway Parkway/Scripps Hi Drive. Currently, there exist two left-turn pockets with over 250 space available for cars waiting to make a left-turn. It is expected the storage provided by these turn pockets would be adequate in the with the project. The anticipated level of service at this intersection expected to be acceptable with the project and with proposed mitga	
Shearer-Nguyen, Elizabeth	From: Bmfleids@aol.com Sant: Sunday, March 24, 2013 8.21 PM To: DSD EdA March 24, 2013 8.21 PM DSD Eda March 24, 2013 8.21 PM DSD Eda March 24, 2013 8.21 PM Subject: Vvatemark EIR	What is being done to increase access from Scripps Poway Parkway coming from Scr minimize the impact of traffic through the Scripps Highlands residential community ab possible to increase the length of the left hand turn lane turning from Westbound Scrip Scuttbound Scripps Ranch Resident Brad Fields, Scripps Ranch Resident	-
		Х-1-	

Response to Letters of Comment – Page 37 September 2013

RESPONSES
AND
NTS
MME
ပ္ပ
ЧO
ERS
Ē

<text><text><text><list-item><text><text><text></text></text></text></list-item></text></text></text>		L-1 Comments noted. These comments do not address the completeness or adequacy of the EIR. No responses are necessary.	L-2 Comments noted. These comments do not address the completeness or adequacy of the EIR. No responses are necessary.	<b>L-3</b> Comments noted. These comments do not address the completeness or adequacy of the EIR. No responses are necessary.	
- 0 0	From: Jeff He [mailto:jhe@accelagen.com] Sent: Sunday To: DSD EA C: President@ScrippsRanch.org Subject: Watermark project in Scripps Ranch To Whom It May Concern: To Whom It May Concern: To Mhom the two components of the Watermark Project: bowling alley and movie theater for the folowing reasons:	1. From business point of view, it is a bad idea to have a bowling alley and a movie theater in Scripps Ranch. A big bowling alley, Mira Mess BowL is less than a miles away in Mira Mess; another ons, Poway Fun BowL, is only 4,5 miles away. Those bowling alleys are located in busier commercial areas and, therefore, successful in attracting customers. Scripps Ranch, being a residential area, known for its 'country living', is not an ideal location for such business, and the business will unlikely be successful. Likewise, it is a terrible idea to have a movie theater in Scripps Ranch. Three other rovie theaters are within 5 miles proximity, in particular, the very popular Edwards Chema is less than 3 miles away. What makes Edwards so popular is not only its central location in very busy Mira Mesa area, but also its close proximity to restaurants, shops etc. The parking capacity also makes it possible	2 to host a large number of customers. In summary, bowing alley and movie theater will not be successful in Scripps Ranch. <ol> <li>Forn resident host of views, it relies serious safety and traffic concerns to have bowing alley and movie theater in Scripps Ranch of not or steps ranch of one so not attract with row to make the resident traffic devices the residents love Scripps Ranch because it is a community seclided from the novie theater in Scripps Ranch and movie theater in Scripps Ranch of one so that archer the residents to the sole on the represent the resident traffic devices the residents love Scripps Ranch of one not represent the runder of patoms. Bowling alleys and movie theaters, on the other hand, can only survive on a large number of patoms from outside of Scripps Ranch the proposed Watermark project is located in less than 0.2 mil from a very quiet Scripps Highland community which in past years had already seen house or ar burgiales, and bowling alleys, and burging network will my dute the runder the runder as seen in area around movie theaters or bowling alleys. Exciting Ranch, the proposed Watermark project is located in less than 0.2 mil from a very quiet Scripps Highland community which in past years had already seen house or car burgiales, and burging from the aronsed locater means will any dome schore, howling ralleys and movie theaters or bowling alleys.</li> </ol>	1 Ido not oppose a shopping center at the proposed location. However, I strong believe only neighborhood I do not oppose a shopping center at the proposed location. However, I strong believe only neighborhood orrientated businesses such as restaurants, coffee shop, medical offices, child care center etc would benefit the local residents. I think the proposed project should include a neighborhood park with play area for children, field for ball games etc.	Juff He, Ph.D. Accelagen, Inc. 6044 Cornerstone Ct W, Ste C San Diego, CA 92121 T: (858) 678-8618 x 112 F: (858) 678-8628

Response to Letters of Comment – Page 38 September 2013

			<b>TTERS OF COMMENTS AND RESPONSES</b>
		M-1	The City of San Diego and San Diego Police Department are responsible
	From: sriko4@aol.com [mailto:sriko4@aol.com]		for monitoring and adjusting street signage and regulating on-street
	Sentt Sunday, April 14, 2013 3:58 PM To: DSD FAS	-	parking. The proposed project provides for adequate parking on-site to
	Cc: dustinstelener@hotmail.com; Mitte@san.rr.com; ctodd@wingertlaw.com; msorens1@san.rr.com Subject: Watermark DEIR - #180357		serve the proposed uses, and no significant parking impacts have been
	Dear E. Shearer-Nguyen:		identified. Incretore, no mitigation measures relative to parking are
	Below are comments from the Scripps Ranch Civic Association on the adequacy of The Watermark Draft Environmental Impact Report.		10-Juncon
	Sincerely,	M-2	People leaving the Watermark project via the driveway adjacent to
	Bob Ilko, President Scripps Ranch Civic Association		Kestaurant Pad A would be limited to right-in/right-out movements. There is currently a raised median and signage preventing motorists from
N-1	1. There is one No Parking sign on Scripps Highlands Drive (SHD) and no red curbing at all. If the restaurant goes in Pad A then we can expect many users will want to use SHD for parking anther than use the Medimpact parking structure. It will be much the stored as a parking structure it will be much easier to grads a space on the street then in the structure. Parking on the street also avoids having to drive through the structure to easier to grad a space on the street then in the structure. Parking on the street also avoids having to drive through the structure to easier to grad a space on the street then in the structure. Parking on the street also avoids having to drive through the structure to easi and the structure to easier structure to easier and the network of the rest of vacuum south or the curbing the structure to also narrows south of the Pad Advinway. We predict patrons will pack on SHD south of the driveway this making it ureated for cars going south or SHD. The street can the nether still pack on SHD south of the driveway this making it ureated to also provide the median neading northbound is narrow and should be signed No SHD. The street and the other still pack on SHD south of the driveway this making it ureated for cars going south or SHD. The street of the median neading northbound is narrow and should be signed No maximactions.		turning left out of this driveway. The opportunity to make a u-turn is provided at the south end of Scripps Highland Drive before turning to go up the hill. Illegal u-turns are subject to enforcement by the San Diego Police Department. Also, please see response no. G-9, above.
	2. There is a raised conter median on SHD at the Pad Adriveway which was designed to prohibit cars making a left onto current to median on SHD at the Pad Adriveway which was designed to prohibit cars making a left onto current to median on SHD at the Pad Adriveway which was designed to prohibit cars making a left onto	M-3	Parking for the proposed restaurant on Pad A will be provided on the first
M-2	And, there is a right unit off sign rounding an vencee soming use on waying use out on Shot. Since sets a regard, turn Ony sign rounding right want to get the relation, othered will make llegal u-turns. In properation of these comments to the DEIR, it was writessed that it happened (and that it will only get worse). People dropping or picking up at the restaurant will chose to come back on the drivway as it will be faster than driving through the particip structure. To solve that problem, the drivway strough be one way only (entrance only, no astrocapt the particip structure. To solve that problem, the driveway structure have only the articip the particip the		level of the parking structure, where designated areas will be provided for restaurant patrons. Additionally the restaurant will provide valet parking. Valet-parked cars would be parked in a valet-designated area of the parking garage. Signage at the restaurant would indicate that valet parking
C M	<ol> <li>Where in the parking structure will customers of Pad A restaurant be located? -Will the parking structure have assigned</li> </ol>		is provided and where self-parking for restaurant patrons is located.
0-IN	spaces (herdimpact employees, medimpact visitors, Fact A restaurant)? If the restaurant parking is not adjacent to the testaurant or limited to the back or upper levels; more customers want to park on SHD rather than use the parking structure.	M-4	The gates will be moved back farther into the garage past the parking area
M-4	<ol> <li>The Medimpact parking structure has gates blocking access to the structure for Medimpact employees only. With the restaurant going on Pad A; will those gates be removed? If so, how will spaces be apportioned?</li> </ol>		allocated for restaurant patrons. Please see also response no. M-3, above.
R-5 ح	<ol><li>Will there be exit direction signs at the Pad A restaurant drop off pick up area informing drivers to go through the parking structure in order to exit onto SHD?</li></ol>	M-5	The narking garage will be signed and nosted directing restaurant natrons
M-6	6. Will there be a new raised median on SHD from the Scripps Poway Parkway (SPP) intersection towards the south or will it remain a painted median? Given the expected traffic volumes leaving the site on SHD turning west onto SPP, the left turn lane should be maximized similar to what SRPG/City did for the double left turn lane on SR Blvd northbound onto MM Blvd.		and employees to drive through the garage or, more likely, the access road between the garage and MedImpact parking. Please see also response no.
	<ol> <li>Proposed triple left turn from SHD onto SPP. Two left turns now with one through lane and one right turn lane. Re-stripe to a triple left turn and a shared through lane. There are only two through lanes on SPP to I-15 SB. The number 3 lane on SPP</li> </ol>	, ,	M-3, above.
M-7	becomes a right turn only to I-15 NB. The triple left wortt work if people in the number 3 left turn lane want to go anywhere other than I-15 NB. Otherwise they will cut in the merge into the number 2 lane towards Mercy Road and LYSB making it a gogantic bottlenecki CPP is three lanes going westbound but haif way from SHD to H-15 NB on ramp it turns into two lanes. To reduce the problem: The number 3 lane on SPP that turns into the right turn only lane for I-15 NB should be a right turn only lane from	M-6	It is anticipated that the painted median on Scripps Highland Drive just
	SHD to the I-15 (add signage to SHD that the number 3 left turn lane is I-15 NB only). This would make the merge begin earlier on SPP for westbound vehicles. By moving the SPP merge further east, will it impact Scripps Summit Drive intersection? The		south of Scripps Poway Parkway would remain in the future. There are currently no plans to install a raised median. Comment Noted.



Response to Letters of Comment – Page 40 September 2013

			TERS OF COMMENTS AND RESPONSES
	current am and pm peak is pretty bad. Atternatively, could SPP from SHD to Mercy Road be three lanes where the number 3 lare is a through lane to Mercy and a right turn to 1-15 forthbound? The number 3 left turn lane would still need to be specifically marked so that drivers don't try to merge to the number 1 and 2 lanes to access 1-15 southbound.)		This configuration differs from what previously existed. This improvement would create additional storage and allow for smoother traffic flow. Additional signage would be provided as appropriate when
M-8	8. Propose shifting westbound SPP over a lane to the right to add queue length to Southbound I-15. Will there be two lanes turning Southbound I-15? Need to know how many cars fit in the existing queue and compare that to how many they expect in the new queue without the project and not sure how much of an introvement it will really be for anything peak and of peak hours. It aqueue without the project and not sure how much of an improvement it will really be for am/pm peak and of peak hours. It aqueue short the project and not sure how much of an improvement it will really be for am/pm peak and of peak hours. It avoid shorten the signal dealy for all releacions by having 3 turning lanes rather than 2 (more cars in less time <b>assuming</b> they have a place to move once they turn onto SPP).	M-8	this mitigation measure is implemented. Comments noted. See also response no. M-7, above.
M-9	9. The traffic study area excluded any traffic impacts to the residential community above The Watermark. We are very concerned about this omission. There are community concernes that drivers will take a cut through the community to avoid SPP and +15 instead try to get to Scripps Ranch Blvd to Mira Mesa Blvd to access the I-15. What signage can be added or other mitigation measures can be used to deter cut through traffic?	M-10	Please see response no. G-9, above.
M-10	<ol> <li>What are the ADT's of the site if built as currently zoned I-P? How does that compare to the ADT's of the proposed project? Not only total ADT's but am/pm peak impacts?</li> </ol>		Project / Development Under Existing Approvals – evaluates full development of the project site as allowed under the existing MedImpact approvals, which
M-11	<ol> <li>Are the signals at I-15 NB off ramp and on ramp synchronized with the signal at SHD? Would that improve or impede traffic flow?</li> </ol>		would occur under the existing IP-2-1 zone. Based on the discussion
M-12	12. How could a double left turn not be preferable to a triple left turn from SHD to SPP? It may be counter intuitive to residents that allowing "X" number of cars period. How would it that allowing "X" number of the set period. How would it improve traffic flow? Residents believe that there is a real and critical problem with the 1-15 SB ramp meter. Mejority of residents who travel in the area come from the east of SHD. If it is extremely difficult to get through the series of lights now on SPP, how by adding 20,000+ additional ADT's not deteriorate to an unacceptable level of service?		presented in the EJIK, 0,240 cumulative AD1, with 700 trips in the AM peak hour and 750 trips in the PM peak hour, are associated with that alternative. As concluded in the EJR, the No Project/Development Under Existing Approvals alternative would generate 15,525 less cumulative ADT
M-13	13. Confirm that the right turn exting the project onto SPP vehicles are prohibited from merging to the number one lare of SPPIo make a u-turn is avoid exting SHD. How will the acceleration lane striping line up with the number one left turn/ u-turn lane approaching the SHD intersection?		than the proposed project, with 271 less AM peak hour trips and 1,414 less PM peak hour trips.
M-14	14. What are the retail hours of the food market?		The EIR also addresses a Prime Industrial Lands alternative (Alternative 3).
M-15	15. How many more vehicles will be added to I-15 NB off ramp to SPP? How much longer of a delay?		That alternative could also develop with the existing IP-2-1 zone. Based
M-16	<ol> <li>Why the need to subdivide from 9 lots to 16 lots? Will they be sold individually?</li> </ol>		on me uscussion presenteu m me EAR, trainc associateu win me Frine Industrial I ands alternative would be the same as the No Prinist/Development
M-17	17. Did the Developer and the City explore the possibility of traffic access directly from the I-15 off-ramp into the parking structure?		Under Existing Approvals alternative, as similar land uses could develop.
M-18	18. How will the heating and air conditioning units be screened for noise and esthetics?	M-11	At the time of the existing conditions for the TTA the signals at the LTE/
M-19	<ol> <li>What is the potential proportional traffic count will enter the project using the new access point on Scripps Poway Farkway versus Scripps Highlands?</li> </ol>	-	Scripts Poway Parkway Interchange were not interconnected with the
M-20	<ol> <li>What is the potential proportional traffic count will exit the project using the new access point on Scripps Poway Parkway versus Scripps Highlands?</li> </ol>		signal at octripps rhighlands Drive. The project applicant would construct the infrastructure needed to synchronize the signals. Signal
M-21	21. Will the dock tower chime? How loud and how often?		synchronization is the City's and Caltrans' responsibility. In the future, with the reconfiguration of the interchange as promoved by the project
M-22	22. Provide information about new mass transit bus stop on Scripps Poway Parkway.		Caltrans and the City may be able to better synchronize traffic signals,
M-23	23. Can there be a connection to the project from the existing blke path that is parallel to 1-15 other than at Scripps Poway Parkway? Could access be created some place near or in parking structure by the proposed theater via constructing a pedestrianblike bridge?		which would improve the traffic flow. Please refer to the TIA for more information.
Tho Wot	21 - C - C - C - C - C - C - C - C - C -		Decension to Lottone of Commont Decension

Response to Letters of Comment – Page 41 September 2013

	ITERS OF COMMENTS AND RESPONSES
2 2 2	As presented in the TIA and the EIR, the additional traffic from the Watermark project is expected to cause a deterioration of the existing level of service. However, the project's proposed improvements to the interchange would help alleviate some of the congestion that is anticipated. Part of these improvements is the reconfiguration of the Scripps Poway Parkway/ Scripps Highland Drive intersection to provide a triple left-turn from northbound to westbound. This improvement is necessary to address the anticipated future demand from the Watermark project and would mitigate the level of service at this intersection.
2- 2- 2-	Improvements to Scripps Poway Parkway and the anticipated future striping for this area are shown in Figure 1-1 ( <i>Scripps Poway Park / 1-15 Raonfiguration</i> ). As shown, a raised median would be extended from the channelized right-out from the project access on Scripps Poway Parkway. This raised median would delay the merge of traffic into the thru lanes on Scripps Poway Parkway to prevent those vehicles from crossing the thru lanes and making a left/u-turn at Scripps Highland Drive.
A-14	The hours of the food market have not yet been determined. However, it is anticipated that the hours for the market would be $7:00 \text{ AM} - 11:00 \text{ PM}$ .
A-15	As shown in Figure 3-4 of the TIA, up to 243 PM peak hour additional vehicles are expected at the I-15 NB off-ramp and Scripps Poway Parkway from the Watermark project. These additional vehicles are expected to add up to 4.5 seconds of additional delay to the intersection when project traffic is added to the existing condition in the PM peak hour (see Table 1-4).
M-16	The project would be subdivided for financing purposes so that separate uses can be financed separately and/or lots can be owned by separate entities. This comment does not address the completeness or accuracy of the EIR.
M-17	Access from the I-15 ramp into the Watermark project was explored by the applicant and found to be infeasible due to Caltrans policy and practice.

Response to Letters of Comment – Page 42 September 2013

Walchinaik project in the event in 1.9 extends bus service to initiality		
again in 2011 and 2012. MIS has consistently stated that no service of Scripps Poway Parkway is being considered in current MTS plans. Th City requested that the project locate a bus stop as part of th		
when the project applicant rias been coordinating with M15 start, initially when the project was first submitted to the City for review (2010), and among in 2011 and 2012 MTS has consistently stated that no service of	77-INI	
It is not anticipated that the clock tower would chime.		
0		
As shown on Figure 3-4 of the TIA, up to 154 pm peak hour trips ar expected to exit the project using the new right in/out access on Scripp Poway Parkway and 872 pm peak hour trips are expected to exit vi Scripps Highlands Drive. This represents up to 15% of the project traffi	M-20	
Scripps Highlands Drive. This represents up to $25\%$ of project trafficentering the project.		
As shown on Figure 3-4 of the TIA, up to 245 pm peak hour trips ar expected to enter the project using the new right in/out access on Scripp Poway Parkway, and 733 PM peak hour trips are expected to enter vi	M-19	
that is architecturally integrated with the primary building on the premises The purpose of this regulation is to ensure that rooftop mechanics equipment is screened from views. Therefore, the project would no result in impacts relative to aesthetics of roof-top equipment.		
cumulatively - would exceed the City's property line standards. No impacts would occur and no mitigation is required. Relative to aesthetics in accordance with Municipal Code Section 142.0910, all heating and ai conditioning units must contained within a completely enclosed structur that is architecturally integrated with the minaw building on the memisse		
Noise impacts associated with HVAC systems are analyzed in Section 5.7 Noise. As presented in the EIR, no noise levels - either directly o cumulatively - would exceed the City's property line standards. No	M-18	

Response to Letters of Comment – Page 43 September 2013

LETTERS OF COMMENTS AND RESPONSES	M-24 As part of the proposed project, reclaimed water lines would be installed, and reclaimed water would be available for use in landscaped areas of the project site.	The proposed project would not result in a substantial increase in water use As presented in Section 5.15 <i>Public Utilities</i> of the EIR Public water	facilities are located in Scripps Highlands Drive and Scripps Poway Parkway. The size and capacity of these existing utilities would be adequate to serve the proposed project. No new systems or alterations to the existing utilities would be required, and no impacts to water facilities	would occur. Additionally, the Ciy has prepared a Water Supply Assessment (WSA) for the project. As presented in Section 5.15, based on the results of the WSA, the project would not result in any	unanticipated demands and would not result in significant impacts associated with water supply.	M-25 As concluded in Section 7.0, <i>Effects Found Not to Be Significant</i> , of the EIR,	an increase in local or regional population would not occur as a direct result of project implementation. As the proposed project would not	result in an increase in population, the project would not increase the use of recreational facilities such that substantial physical deterioration of the	facility would occur. The proposed project would not result in the potential to impact parks and recreation. See also response no. G-4, above.	<b>M-26</b> As stated in the comment, traffic counts for the TIA were completed in late 2008. Counts were completed at that time in order to commence environmental analysis for the project. Early discussions with the City, Planning Group, and Caltrans preceded the completion of the first draft TIA, which was based on the 2008 counts.	In 2009/10, the recession was particularly deep and periodic counts showed volumes which were significantly lower than 2008. For example, according to Caltrans count data, freeway volumes on I-15 at post mile 17.311 (Mercy Road) showed a "Back AADT" (i.e. ADT) of 256,000 in 2008, 240,000 in 2009 and 248,000 in 2010. The effects of the recession caused economic activity and traffic to decrease. Therefore, counts from	2008 more appropriately reflect "typical" traffic conditions for the area.
		M-24 24. Will this project use? "Purple pipe" reclaimed water? If not, why not? How will the Developer miligate what is likely to be a huge increase of water use? M-25 25. Where will retail, commercial and office employees go to recreate when not at work? Will they visit parks in Scripps Backard? These previous are already stretched by overuse and buddet cuts. Has this immach been	studied and what does the Developer propose in way of mitigation? 26. "Accepting that the DEIR set forth a rather imposing picture of the traffic impact of the project, the DEIR still does not even begin to carrately detecte the problem with the traffic cutuation. The traffic impact of the project, the DEIR still does not even the freeway on ramp has become with the traffic cutuation. The traffic impact of the project, the DEIR still does not even the freeway on ramp has become with the traffic cutuation. The traffic free was analyzed in the estimate the freeway on ramp has become more compated by the 5 years after the study was completed. In essence, the traffic impact is exponentially greater than DEIR set out. The on ramp wait time exceeds 20 minutes as raffic backs up all the way to 5 corps bornmit. And the tai without the extra ADT's contempleted by the report." Day of Bergin at a the way Traffic courts summarized on this flave ware completed in October and November 2008.)	27. It defines logic and common sense to assume that the evaluated intersections and on-ramps hawn't become more congested since 2008? If not, what data/information do you have to support such a conclusion? If so, should the DEIR be revised according the use of outdated traffic information/data to be a fatal flaw in the DEIR and request an updated traffic study using current that projected on the most validated model. To repeat, it is a disservice to residents and City Council to base traffic projections on stale data.	M-28 28. A more recent, more accurate traffic count is a prudent measurement to use rather than something 5 years old? Shouldn't the resident and City's decision makers have up to date information?.	W-29 you have to support such a condustor? If yes, should the DEIR be revised accordingly?	WI-50 you have to support such a conclusion? If yes, should the DEIR be revised accordingly? 31. A study of the freeway ramp delays today would be significantly better to use than counts from 2008? We believe thatthe MI-31 DEIR should be revised accordingly.	M-32 32. "I fail to see the need for this project. There is so much vacant office space now, why do we need more office buildings? I haven't heard about not enough hele rooms. There is phenty of retail and movie theatres now, and certainly enough traffic. I prefer to preserve what little greenery we have now." Nan Morris	33. "We are hoping to see a different approach than just another strip-mall (like the New Vons mall or Poway's Community Drive Poway Rd mail or anne two of hundreds of similar mails around sam Delego Courty). We initivital Scripps Rauch and a court of the New Vons mall or Poway's Community dessrves and would benefit from a novel, better and potentially more distinctive design approach. In that regard, key features that we would support in its design (and a first giance i lokas) is in a novel of the new Vons mail below) would be statistical reads around statist of the set index in the site mary of these might a leady be in the with your email below) would support in its design (and a first giance are with parking on the unside surrounding its chart aftraping weaking distance. This would make in different from the typical "drive in, go to one! wo nearby places and drive out" that typings on many spanwing on the outside of the project, visitors would see many approxing atom would appending the New Your small. By making on the outside of the project, visitors would be support indis, including the New Your small. By making on the outside of the project visitors would be support indis.	Park Inter face note, waik moto the mail and be surrounded by dimeter styching distance. Such moto the mail and be surrounded by dimeter styching distance. Such certain and source of approximation structure and sourced by the strip mail configurations. We've all seen many of the periphenis shops/ restaurants in the New Vons area struggle and many have closed, potentially contributed by a lack of routine foot traffic. It makes a more economic, communy and environmental sense to noy have to park on one in the New Vons area struggle and many have closed, potentially contributed by a lack of routine foot traffic. It makes a more economic, communy and environmental sense to noy have to park one of the periphenis chock area in the mail. At first granter, the seems that the parking structure you mentioned may at second state its. A contract commune area that can be used as a gathening place, including a playsgruud for kides and possibly ta strated area that the staurants and structure and possibly at the mark of the structure on the used as a gathening place, including a playsgruud for kides and possibly ta strated area that the staurants and strate and the strate and any area that can be used as a gathening place, including a playsgruud for kides and possibly ta strated area that the staurants and strate and possibly ta strated area that the staurants and strate strate and any and keep our resident nght-life lowers at Ballast Point Somewhere for chrise, live music might second much the community and keep our resident nght-life lowers - in Scripps Ranct. The structure structure and possibly at the plane of the structure structure and busited as the structure structure and the structure structure and any area that the structure struct	34. We live in the community just up the hill from this project and we have experienced more traffic since the opening of the Wedimpact building. We also have employees walking around our neighborhood, smoking, and wandering up Traitcrest and into the housing areas during their breaks. While this may seem like a wonderful opportunity to create jobs, in my opinion, it is an the housing areas during their breaks. While this may seem like a wonderful opportunity to create jobs, in my opinion, it is an the housing areas during their breaks. While this may seem like a wonderful opportunity to create jobs, in my opinion, it is an the nortice of the transmission of the providence of the transmission.	

	LETTERS OF COMMENTS AND RESPONSES
	Subsequently, with construction at the interchange that began in 2008 and was completed in 2011, counts could not be re-taken. However, in order to account for regional growth, counts were increased with traffic data from other projects as they were completed to illustrate "Near-Term" and traffic increases have been projected to show long term trends as well. Therefore, any increase in traffic volumes is captured in at least one of the three time horizons (i.e. existing/Near Term/Long Term) that were evaluated as part of the TIA.
	<b>M-27</b> As discussed in response number M-26 above, traffic counts were completed in late 2008 during peak traffic conditions at the I-15/ Scripps Poway Parkway interchange and surrounding areas. Ramp volumes from Caltrans count data shows that in 2006, the Southbound on-ramp had 15,400 ADT. In 2007, traffic conditions peaked at 16,300 ADT and subsequently declined slightly to 16,200 ADT in 2010/11. For the Northbound on-ramp, the difference is similar with 10,200 ADT in 2010, even attached). Therefore, it can be seen that traffic conditions were relatively stable with a modest peak occurring around 2007. The existing counts were taken in this time-frame and traffic conditions were monitored to make sure no significant changes occurred.
	Also, as discussed in response M-26, traffic conditions were temporarily impacted during construction making more recent counts difficult. For example, when ramp meter rates were requested from Caltrans in March 2010 as part of ongoing monitoring of traffic conditions while the traffic study was being completed, Caltrans noted "temporary conditions during construction". These temporary conditions led to significantly higher ramp meter rates than is typical and impacted both observed conditions and measured values based on temporary signal timing and lane configurations.
	Therefore, "pre-construction" conditions from 2008 were deemed most appropriate in evaluating impacts from the Watermark project not only because counts completed during this time frame represent a recent peak in traffic conditions but because they most accurately represent "typical" traffic conditions for the area not influenced by construction activity.
	<b>M-28</b> Please see response no M-26 and M-27, above.
The Watermark	Response to Letters of Comment – Page 45

ponse to Letters of Comment – Page 45 September 2013

S	
ISE	I
0	I
SР	I
Ш́	l
Г	l
Z	l
A Ω	I
Ě	I
Π	I
ΔĮ	I
Ō	I
0	I
Q	I
SS	I
Ш	I
F	
	11

LETTERS OF COMMENTS AND RESPONSES	<b>1-34</b> Comments noted. Also, please see response no. G-9, above.	<ul> <li>1-35a Comments noted. These comments do not address the completeness or accuracy of the EIR. No responses are necessary.</li> <li>1-35b The proposed project would not affect the existing Class I bike path that course complete the the transfer of the proposed project would be that is broaded within Column? with of the proposed project would be that is broaded within Column? with of the proposed project would be provided with the proposed project would be provided with the proposed project would be provided with the provided with the</li></ul>	way. As described in Sections 4.0, <i>History of Project Changes</i> , and 5.2, <i>Transportation/Traffic Circulation/Parking</i> , the project proposes the construction of a separated bike lane within a multi-use path facility	atong scripps Foway Parkway to reduce the potential for bicycle/motorists conflicts and to improve bicycle and pedestrian safety/access along the project's frontage on Scripps Poway Parkway.	<b>1-36a</b> Please see response nos. J-4 and M-13, above.	<b>1-36b</b> Comments noted. These comments do not address the completeness or accuracy of the EIR. No responses are necessary.	<b>1-36C</b> Comments noted.	<b>1-36d</b> Comments noted. These comments do not address the completeness or adequacy of the EIR. No responses are necessary.	1-36e Comments noted.	<b>1-36f</b> Comments noted. The applicant anticipates that the market would be a specialty market on a smaller scale than a full-size grocery store.	<b>1-37</b> Comments noted. These comments do not address the completeness or accuracy of the EIR. No responses are necessary.	<b>1-38</b> Comments noted. These comments do not address the completeness or accuracy of the EIR. No responses are necessary.
		opening for more traffic and people to start rearring our neighborhood. We are also part of the neighborhood watch and theths are more rampant in recent months, especially during the holdays. I vahiolehartedly oppose this project on the basis of what I am acading in the overflow of traffic going in and and our neighborhood. We are also people and people are solved to the overflow of traffic going in and and our neighborhood. What news the traffic and the traffic going in a basis of what I am these concents on stronghout the evening and right as well. Hould like to know who the appropriate off concil people are to bring as a stropping opportunity, it threatens to add more frame and cover areas. "Deamna Wong as a stropping opportunity, it threatens in the and mere barrene to care to the and the solution opportunity.	M-35a 35. "I'm really concerned about the increase in traffic1 I don't see a way it can be dimensioned adequately. In fact, access to the freeway (southbound) from Scripts Poway Parkway is already slow at tush hour. The modification uder the bridge didn't help muchand with the way and tush hour. The modification uder the bridge didn't help additional mall. We have all the services offered in the pan within less then 2 miles score, the planthere are no effect of the planthe modification uder the bridge didn't help additional mall. We have all the services offered in the plan within less then 2 miles from the planmed location (Edward cinemas in Mu-35b Maranesa, along with shopping malls and restaurants, 2 holes across the planmed location the new Yons scored from its accounted to the the larks the current biles carbot by 1 don't splan The new Yons scored from its accounted to the the larks the current bile to the rest. Now we don't restaurants, the new Yons scored from its accounted to the plan and shops, Now we don't need the current biles carbot bile the plans The second is not well and more second to the plane current biles carbot biles accounted to the restaurants.	M-36a 36. "I) the extra entrancelexicit directly on to Scripps Powery Parkway. Wy concern is that this feed at will limit the ability of people to make a right turn on the red light exting the 15 freeway north and turning on to Scripps Powery Parkway. If cars can't make that right turn on a red light because of a back up of cars with transces, will will an an a stacking of cars with that exit anno and polaritally all the way on to the feedway. At that hours, this is a such at attransce, will will spot. Treatise can't and be and polaritally all the way on to the feedway. At that hours, this is already a dry tapol. If realize and the exit annotated polaritally all the way on to the feedway. At that hours, this is already a dry tapol. If realize and the exit people Holphard Drw, but that mild inconvenient means that the avail and ready and the article statement will not not work the article statement will be availed and and polaritally all how work that make the avail convenient will be availed and an and an and the availed to the availed and an and the availed to the availed t	M-36b bits of the second secon	perturgs a less tray neighborhood than some of us would like to believe it is. Further, foorther subgests that the retail shops with the irrel at your of the end want to be in the same space as a bowling alley. Putting a bowing alley there, suggests that the retail shops will be a 99 cent store and the like-all of which have there are in the some space as a bowling alley. Putting a bowing alley there, suggests that the retail stops will be a 99 cent store and the like-all of which have there in the same space as a bowling alley. Putting a bowing alley there, suggests that the retail stops will be a 99 cent store and the like-all of which have there in the intervent design M-36C. It is alley assisting station that the retail stores will have the advect of the current design of cont states which both are deviced are of stores and intervio book of stores in the advect. All have than a data the current design for the current design with both are deviced are of stores and intervio book of stores with parking in the avers and water for a state state and intervio book of stores with parking network of the current design for the current design with a parking interviewent. All have the avers and water a fine.	M-36d measures measure incomparing the former location or approximation throw measures measures or measures and a measure income measures or location in the measures o	M-36f biological contractions of Yami's & Fingelia. That means semitting other than chains and stat casual. A bakery would be delightlike. Perturbas the 3rd Corner owners would open a new restaurant here. They have one in pain Desert and the model of the state of the contract seery what wind of market. She may any of restaurant we need: 9) MM-46 provide the state of	M-37 31. "Although a new theater, restaurants, and a small number of shops may be nice and add to empoyment possibilities in the Scrops Ranzh reas, overalt he jobs that houses and retail restaultsments provide are low verti and with the plots that house it the Scrips Ranch area is to add traffic to Scrips Ranzh-verti the Scrips Ranch area is to add traffic to Scrips Ranzh-verti the Scrips Ranch area is to add traffic to Scrips Ranzh-verti the Scrips Ranch area is to add traffic to Scrips Ranzh-verti the Scrips Ranch area is to add traffic to Scrips Ranzh-verti the number of house real evel of tranship and pay barely information may would like in non-more and house the script script and the number of house real evel of proposed development.	massive development that will increase traffic in an area that has overloaded and crowded access to the 15 freeway corridor; these are surface streets we use on a daily basis, and count thems are antered vocassive. Scripts Pavery Parkwey carriers traffic to the freeway (from Poway and Deyrod; it can't handle more cars without adverse affect all driver who must use the noad. Let's be reasonable about the density of develop and limit the types of growth necessary in this area. We are a family of three generations of addit women, and we live in Scripts Ranch because it of its "live ability" and people-firerly neighborhoods and services. The proposed Watemark development does not fill with our vision of our community" Call Rchmond, Magaret	Samuelson Walker and Kelly Walker 38. "When I think that McMillan paid \$2 million for Scripps Ranch HS and \$1 million for our library, aud still made millions in M-38 profit, while these Sudberry folks are possibly building an on-ramp from a freeway that looks like a parking lot each moming and	-W

Response to Letters of Comment – Page 47 September 2013

			TERS OF COMMENTS AND RESPONSES
	-		
		M-39	Comments noted. These comments do not address the completeness or accuracy of the EIR. No responses are necessary.
	each afternoon, it makes me wonder at the tementy of the Sudberry Properties outfit. To whom do I send this remark?" David DeBus	M-40	The number of shops is not known at this time. The buildings are
M-39	39. This sounds great, Perhaps a little heavy for that area 1 wonder if the population will support "highend" relatil 1 also worder if they provide the set of the		designed in a manner that smaller or larger shops can be created, depending on the needs of the shop owner.
M-40	if given a chance." Natalia Moorhead 40. What is the proposed number of retail stores as shown in the plans made available to the community?	M-41	The project does not include a food court. It is anticipated that there will be smaller mich-serve food services such as coffee shore
M-41	<ol> <li>Is there a food court in the project?.</li> </ol>		will be summer, quick-serve rood services, such as conce surply, sandwich shops, etc.
M-42	42. If 75% of the vehicles enter the Scripps Highlands driveway, are the two left turn larses long enough on westbound Scripps Powey Parkway to handle the queuing traffic?		Diana adente ana M 12 abarro
M-43	43. If 85% of the vehicles leave using the Scripps Highlands driveway, should it be signalized?	1/1-42	riease refer to response M-13, above.
M-44	44. What is the road designation for Scripps Highlands Drive (residential; collector; arterial)?, What will be if the project is built?	M-43	The intersection of Scripps Highlands Drive/ Scripps Gateway Court
M-45	45. What is the ADT capacity and actual traffic numbers for Scripps Highlands Drive (ADT) in 2013?		(access to the project) is planned to be signalized as part of the
M-46	<ol> <li>Is the left turn queue long enough on northbound Scripps Highlands to handle westbound turning traffic onto Scripps Powery Parkway?</li> </ol>		Watermark project.
	47. What is the level of service of intersection existing versus proposed project of Scripps Poway Parkway and Scripps Highlands?	M-44	Scripps Highlands Drive currently operates as a four lane collector,
M-47	48. What is the level of service at 1-15 and Scripps Poway Parkway? What impact does the project have on ADT, delay and level of service? Keep in mind there are two signals at the 1-15 (one on each side). Note: The left turn queue going south on 1-15 is for each diverse and we have a service in the area on a side in what hards.		between Scripps Poway Parkway and Scripps Gateway Court, and is classified as a two lane Collector in the Miramar Ranch North
IVI-40	49. What is the total number of ADT's for the no project alternative in comparison to the proposed development?		Community Plan.
M-49 M E0	50. What are the changes in ADT's for the ampm peak periods between the no project alternative versus proposed?	M-45	Please refer to Fioure 4-1 of the TTA for the City of San Dievo level of
M-51	<ol> <li>The history of the site is confusing in the Master Plan Guideline. Please consider doing an overview of the Scripps Gateway development and how Medimpact and now Watermark are part of that development in one concise location in the Final EIR.</li> </ol>		service and capacity table. A four lane collector has a capacity of 30.000 ADT According to counts from October 2008 Services
M-52	53 Has the Treffic Immod Andraics (TAA) as astilize revenued channed?		UV/UV ADI. ACCOTUNIS U COUNTS ITOM OCODET 2006, SCHIPPS
	2 Thas the Intermined charaysis (1x) as samile presented charaged? Right In, right out access from the site heading asil on SPP. Weating of SPP from 1:51 to Scrippa Highland Drive: Meaning of SPP from 1:51 to Scrippa Highland Drive: Longer left tim pocket for the SPP to Tartific to the SB 1:15 ramp; Signal miligation at 1:51 contribution and southbound intersections and Scrippa Highland intersection; Right miligation at 1:51 contribution and southbound intersections and Scrippa Highland Intersection. Intercharage Improvements at north ramps, acuth ramps, and Scrippa Highland Drive/SPP intersection.		Highland Drive had 4,099 ADJ. Please refer to Appendix B of the TIA for additional traffic count information. Counts used in the TIA are from 2008. A discussion of traffic conditions from 2008 to 2011 is found in response to comments M-26 and M-27. Traffic data from 2013 was not published or available at the time the traffic study for the
M-53	53. Describe what additional lanes under the I-15 will be. Through lanes to Mercy Road and/or turn lanss to I-15?		Watermark project was completed or at the time environmental analysis
M-54	54. Describe how the left turn pocket for westbound SPP to I-15 south will be made longer.		commenced.
	55. Describe the improvements to I-15 north ramps and I-15 south ramps.		
сс-M М-56	56. The DEIR says that Medimpact buildings will be 350,743 square feet on page 2-5 but the traffic stucy page 7-1 says it is 324,774 square feet. Which one is correct? If the DEIR is correct, the traffic study should be revised?		
The Wate	ermark		Response to Letters of Comment – Page 48

	TERS OF COMMENTS AND RESPONSES
M-46	It is anticipated that, with the reconfiguration shown in the Scripps Poway Parkway / I-15 Reconfigurations figure above (also Figure 1-1 of the TIA), there will be enough storage for left turning vehicles from Scripps Highland Drive to Scripps Poway Parkway. This intersection is projected to operate at an acceptable level of service with these improvements as shown in Table 1-13.
M-47	The existing level of service at Scripps Poway Parkway/ Scripps Highlands Drive is "B" in the AM peak hour and "C" in the PM peak hour. The level of service for Scripps Poway Parkway/ Scripps Highlands Drive is "C" in the AM peak hour and "D" in the PM peak hour in the existing with project condition. For more information please refer to tables 1-4 thru 1-6 and table 1-13 of the TIA.
M - 48	The level of service at Scripps Poway Parkway and I-15 NB and SB ramps without project for Existing, Near Term and Horizon Year 2030 during the PM peak hour are "C", "D" and "D" respectively. During the AM peak hour, the level of service is "B" for all conditions without the project. The level of service at the Scripps Poway Parkway and I-15 SB ramps with the project for Existing, Near Term and Horizon Year 2030 scenarios during the AM peak hour, the level of service is "C", "D" and "D" for the Existing, Near Term and Horizon Year 2030 scenarios with the project respectively. The level of service at Scripps Poway Parkway and 'D" for the Existing, Near Term and Horizon Year 2030 scenarios with the project respectively. The level of service at Scripps Poway Parkway and 1-15 NB ramps with and without the project for Existing. Near Term and Horizon Year 2030 scenarios with the project respectively. The level of service at Scripps Poway Parkway and 1-15 NB ramps with and without the project for Existing. Near Term and Horizon Year 2030 scenarios with the project respectively. The level of service at Scripps Poway Parkway and 1-15 NB ramps with and without the project for Existing to n Figure 3-2 and Figure 3-3 of the TIA, it is anticipated that 13,120 ADT and 547 peak hour trips will be contributed to the interchange area by the proposed project. This will contributed to the interchange area by the proposed project. This will contributed to the interchange area by the Proposed project. This will contributed to the interchange area by the Proposed project. This will contribute to an increase in both the NB ramps and the SB ramps were evaluated in the TIA with associated level of service changes. Capacity deficiencies have been noted, and mitigation has been proposed where feasible. Please refer to Section 1.0 of the TIA for more details.

Response to Letters of Comment – Page 49 September 2013

S	
Ш	L
Ž	L
õ	L
S	L
Ш	L
ц С	L
	L
A	L
S	L
Ξ	L
1Π	L
Ϋ́	L
ō	L
$\mathbf{O}$	L
Ц	L
0	L
R S	
Ш	
F	
Щ	

LETTERS OF COMMENTS AND RESPONSES	<b>1-53</b> Figure 1-1 (re-printed below as <i>Scripps Poway Parkway / 1-15 Reonfigurations</i> with response no. M-7) and page 1-7 of the TIA describe improvements to the 1-15/Mercy/Scripps Poway Parkway Interchange. As described in the TIA on page 1-7 and shown in Figure 1-1, the proposed reconfiguration would narrow the median and shift the westbound through lanes on Scripps Poway Parkway to the north and provide additional queuing length for westbound traffic on Scripps Poway Parkway to the Interchange. The existing "back-to-back" left turn lanes would be eliminated, and additional queuing for westbound 1-15 would be provided. The additional westbound lane in the median would be turns onto the 1-15 southbound ramp. Please refer to Figure 1-1 for details.	<b>1-54</b> Please refer to comment response M-53, above.	<b>1-55</b> Please refer to comment response M-53, above.	<b>1-50</b> As presented in Section 10.2.2, Alternative 1 – No Project/Development Under Existing Approvals Alternative, the No Project/Development Under Existing Approvals alternative would generate 15,525 less cumulative ADT than the proposed project, with 271 fewer AM peak hour trips and 1,414 fewer PM peak hour trips.	<b>1-51</b> The history of the approvals in effect on the project site is presented in Section 2.3, <i>Project History</i> , and Section 3.1, <i>Project Background</i> , of the EIR.	<b>1-52</b> The TIA has not changed since the DEIR was released. Information regarding mitigation can be found in Figure 1-1 and on page 1-7 of the TIA. Information regarding pedestrian and bicycle access as well as associated improvements and the right in/out access can be found in Section 12.0 of the TIA.	<b>1-56</b> Both the DEIR and traffic study are correct. Of the 350,743 square feet built and approved on the Med-Impact site, only 324,274 square feet is office space which generates traffic. The remaining square footage is dedicated to on-site amenities for Med-Impact employees only and does not generate any traffic.
LETTER	M-53 Figure with re with re via the T the T the T reconfiguration of the parkw. Would left figuration onto the provide the figuration of t	M-54 Please	M-55 Please	M-50 As pr Exist Appr the p fewer	M-51 The h-Section	M-52 The T regaring TIA. assoc Sectio	M-56 Both built office dedic not g

Response to Letters of Comment – Page 51 September 2013

LETTERS OF COMMENTS AND RESPONSES	M-57 Please see response no. I-49 above.	<b>M-58</b> Trip generation used in the EIR and TIA are based on the City's Trip Generation Manual. The generation rates are calculated by multiplying a "Trip Rate" factor by the development intensity. Peak Hour trips are calculated based on a percentage of the average daily traffic (ADT).	Trip generation for MedImpact only is calculated by multiplying the square footage of MedImpact Buildings 1 and 2 by a generation rate of 10 trips per 1,000 square feet + the future development of MedImpact	allowed under the existing entitlements multiplied by a generation rate based on the logarithmic formula of "Ln(T)= $0.756$ Ln(x) + $3.95$ " for multi-tenant office space. Because the No Project / No Build Alternative	is build-out under the existing MedImpact entitlements, the trip generation calculation is the same for MedImpact only and the No Project / No Build Alternative.	Trip generation for the proposed project is presented in Table 5.2-6,	Watermark Proper Lraftw Generation, of the EJR. For MedImpact Buildings 1 and 2, the same formula as expressed above is used; however, the	building square footage is slightly different. (See 1 able 5.2-6.) In addition to MedImpact Buildings 1 and 2 and multi-tenant office space, the	proposed project includes retail uses, a movie theater, and hotel use. The trip generation for retail uses is calculated by taking the square footage for	the retail space and multiplying that by the logarithmic formula of	Lu( $y$ ) $x$ , Lu( $x$ )+32 . I trip generation for the movie theater is calculated based on the square footage of the movie theater multiplied by	80 trips per 1,000 square feet. The hotel's trip generation is calculated by multiplying hotel rooms by 10 trips per room.		<b>M-59</b> The target office development for the project is presented in Table 3-1, <i>Proposed Project Development Intensity</i> . As shown in that table, a target development intensity for office use is 502,112 square feet, which includes 350,743 square feet of office use for MedImpact. The Site Plan for the project shows 132,369 square feet of mixed-use office space. The EIR and TIA are conservative in their evaluations, assuming a higher amount of office use than currently proposed, which results in a greater amount of parking.
		M-57       57. What is the comparison of no project build out driveway counts versus project for driveway counts?         M-58       58. How exactly are the ADT's calculated for:         -       Montpact only?         -       No project alterburd?         -       Project alterburd?         -       Project alterburd?         Show, the comparison for calculated for:       -	M-59 59. DEIR on page 5.1-17 says office total is 502.112 square feet (including Medimpact of 350.743 square feet) thus leaving 151.369 square feet of office in the project. Parking calculations says 132.369 square feet of office. Which is correct? Which should be revised? . Should the traffic study be revised accordingly? M-60 80. What is head DT's leaving the project to go South on Scripps Highlands Drive? What is AMPM peak ADT's and other than ADT's leaving the project to go South on Scripps Highlands Drive? What is AMPM peak ADT's and other than AMPP peak ADT's leaving the project to go South on Scripps Highlands Drive? What is AMPM peak ADT's and other than ADT's peak ADT's leaving the project to go South on Scripps Highlands Drive? What is AMPM peak ADT's and other than ADT's Drive?	M-61     61. What is the ADT capacity of Scripps Powery Parkwary?       M-62     62. What is the ADT capacity of Scripps Highlands Drive?       M-63     63. How many ADT's will be added to Scripps Highlands Drive?	M-65 65. When will the emulations for the Project recreate in the community of the emulation of the community of the communit	66. In the DEIR analysis of the Prime Industrial overlay, the text discusses vacancies at the Northridge business park (ie, Nokia no longer there). That may have been the case in 2008, but since then Intel has moved in thus challenging the corclusion that Prime Industrial overlay is not even an Audartial and is a casro resource that requires very thoughfut managoment.	67. According to industry experts, the industrial vacancy rate is dropping and look to speculative construction in 2014 and 2015. See http://www.utsandiago.com/news/2013/apr058/industrial-vacancy-colliens/ How can the Prime Industrial overlay be removed if there is expected new industrial space construction for the next two years?	M-68 Rs. In Appendix C (Traffic Analysis) regarding traffic mitigation page 14-14 references Figure 1, but there is no Figure 1. Please Enroyde.	M-69 The traffic indigation discussed in Appendix C page 14-14 says that the "back" tell turn larses will be eliminated and additional queuing will be added to Southburdh 1-15. Exactly how many feet of queuing and approximately how many vehicles, and what proportion of the projected increases will this accomposate?	70. The traffic unplaced in Appendix C page 14-14 says that the median modifications to increase road width on Scripps Poway Parkway will be required. How exactly will the median be modified, how many feet and approximate number of vehicle queuries will be added?	M-71 71. In the DEIR, the theater is listed as being 43,917 square feet on pages 5,1-17 and 10-5 but in Appendix O it is listed as 4,500 square feet. Which one is correct?	72. In the DEIR, the threater is listed as being 43.917 square fact on pages 51-17 and 10-5 but in the perking calculations it is prised as being 35.917 square fact. Which one is correct? If the DEIR is correct then the parking calculations needs to be corrected?	M-73 73. Without the benefit of reviewing the Conditional Use Permit for the theater, we cannont ascertain which square footage is used in that permit and how does it compare to the DEIR text and parking calculations square footages?	
	LETTERS OF COMMENTS AND RESPONSES													
-------------	---													
	This allows for the potential that the ultimate mix of uses may differ from what is shown as the target development intensity and that more office uses could occur in the project, provided that the trip generation and AM/PM peak hour trips presented in the TIA and evaluated in the EIR are not exceeded. Parking would be provided based on the actual use and in accordance with City parking regulations.													
	<b>M-60</b> Figure 3-1 of the TIA shows the expected trip distribution percentages from the Watermark project. These percentages are based on a SANDAG Select-Zone traffic model as discussed on page 4-2 of the TIA. Based on this model, 3% of project traffic with 557 ADT and a maximum of 31 peak hour trips in the peak direction were shown to travel south on Scripps Highlands Drive from the project site (see Figure 3-5 of the TIA). Please also refer to response no. G-9.													
	<b>M-61</b> As shown on Table 5-1 of the TIA, the ADT capacity of Scripps Poway Parkway between the I-15 ramps and Scripps Highland Drive is 60,000 ADT.													
	M-62 Per Table 5-1 of the TIA, the ADT capacity of Scripps Highlands Drive is 30,000 ADT along the four-lane portion of the road.													
	M-63 Please refer to Figure 3-7 of the TIA (Figures 5.2-4a and 5.2-4b in the EIR) to see the traffic assignment from the project. Up to 17,209 daily project trips are anticipated to use Scripps Highlands Drive.													
	<b>M-64</b> Please refer to Tables 1-1 thru 1-3 and Sections 6.0, 9.0 and 11.0 of the TIA for this information. As shown on Table 6-1, in the Existing with project condition, only Scripps Poway Parkway between I-15 and Scripps Highlands Drive is expected to be over capacity in the existing with project condition along the six lane portion of Scripps Poway Parkway. Also as shown in this table, the segment of Scripps Poway Parkway between I-15 and Scripps Poway between I-15 and Scripps Highlands Drive, capacity is expected to be exceeded by 10,733 ADT in the existing with project condition and 12,711 ADT in the Near Term with project condition (see Tables 1-1 and 1-2).													
The Michael	Docession to Lottore of Commont Darie E3													

Response to Letters of Comment – Page 53 September 2013

Response to Letters of Comment – Page 54 September 2013	<b>'atermark</b> nvironmental Impact Report
Please see response to comment M-69, above. The median would be narrowed by 12 feet for a length of approximately 450 feet.	M-7
Approximately 300 feet of actual queuing space will be provided through the additional lane to be provided in the median as shown on Figure 1-1 (reprinted above as "Scripps Poway Parkway / I-15 Reconfigurations"). However, together with the whole improvements shown on Figure 1-1, approximately 750 feet of additional queuing space will be provided versus the existing condition examined in the TIA. This additional queuing space will be provided through a narrowed median, as well as lane reallocation from the previously existing thru lane to feed the left turn lanes. However, since the signal timing of the future interchange after mitigation has not been established, it is impossible to quantify how much this will improve the existing and future queue with the project. Table 1- 13 of the TIA shows the expected improvement in level of service at I- 15/ Scripps Poway Parkway with project mitigation.	ρ-Μ
<b>3</b> Page 14-14 of the TIA references Figure 1-1. The figure can be found in Section 1.0 of the TIA found in appendix C to the EIR.	M-6
Comments noted. See also response no. M-66 above.	M-6
Comments noted.	M-6
Please see response no. M-25 above.	M-6
On the segment of Scripps Poway Parkway between Spring Canyon Road and Scripps Creek Drive, capacity is expected to be exceeded by 1,960 ADT in the existing with project condition and 2,479 ADT in the near term with project condition (see tables 1-1 and 1-2). As discussed on page 1-7 and shown in Figure 1-1 of the TIA, an additional lane would be added to the segment of Scripps Poway Parkway between I-15 and Scripps Highlands Drive as mitigation to augment roadway and intersection capacity. A peak hour arterial level of service analysis on Scripps Poway Parkway between Spring Canyon Road and Scripps Creek Drive shows an acceptable level of service in peak hours for the existing and near term with project conditions (see tables 1-1 and 1-2).	
	Ī
<b>TTERS OF COMMENTS AND RESPONSES</b>	

LETTERS OF COMMENTS AND RESPONSES	M-71 There is a typo in the <i>Prime Industrial Lands Criteria Analysis</i> contained in Appendix O to the EIR. The size of the movie theater should have read "45,000 square feet", a rounding and approximation for the size of the movie theater as proposed by the project and shown on project plans. This error has been corrected. This typo does not affect the conclusions of the <i>Prime Industrial Lands Criteria Analysis</i> or the conclusions in the EIR.	M-72 The EIR and the TIA are based on the initial project submittal, which showed a larger theater. In this manner, the EIR provides a worst case analysis. The EIR and TIA show adequate parking based on the target development intensity, which included a larger theater.	<b>M-73</b> Please see response no. M-72, above. Parking for the theater would be provided in accordance with City requirements and at a rate that corresponds to the actual square footage for the theater when building permits are submittal for City approval.	

Response to Letters of Comment – Page 55 September 2013

S
ш
S
Ζ
$\overline{\bigcirc}$
ň
йí
$\overline{\mathbf{v}}$
$\Box$
Ζ
$\overline{\triangleleft}$
$\hat{\mathbf{n}}$
С С
5
Ξ.
Ξ
$\leq$
$\geq$
Ο
()
Ë
O
S
Ŕ
ш
F
H
ш

I

D-18 of the DEIR it states the project will provide "balance of traffic". How exactly does the prevent disingentuous when the luticrum of the "balance" resists on data collected in 2008 and signestication during provide "balance of traffic". How exactly does the project will provide "balance of traffic". How exactly does the project and a provide "balance of traffic". How exactly does the project and a provide "balance of traffic". How exactly does the project and a provide "balance of traffic". How exactly does the project provide "balance of traffic" when the project does over 15,000 ADTs versus no project and common sense, and more importantly is not supported by the data.	M-74 As a predominantly residential community, Miramar R minent majority of the currently experiences high peak hour/directional traffic. A retail development, the Watermark project will add traffic du PM and weekend periods and directional traffic that directional and peak hour traffic of the community. For exa primarily closed during the morning peak hour of traffic. traffic is balanced in the PM peak hour between directions. directional traffic helps spread traffic to use other elements network which do not conflict with the high residential pe Additionally, by serving retail needs within the community length of vehicle trips may be reduced. Please refer to Table of the TIA for detailed information.	<b>M-75</b> The "balance" of trips referenced in the EIR refers to commercial retail uses generate traffic during off-peak throughout the day, whereas, office uses create directiona occurs primarily as traffic entering the site during the AM p leaving the site during the PM peak hour, as employees arr work.
74. On page 1 projected ADT got 75. How does This affronts reaso	On page 10-18 of the DEIR it states the project will provide "balance of traffic". How exactly does the p appears odd, even disingmuous when the induction of the obtaince" rest or data collected in 2008 and sis addADT generation during non-peak hours that are approximately 20% of Scripps Poway Parkway traffic Hourd does the project provide "balance of traffic" when the project adds over 15,000 ADT's versus no pr afforts reason and common sense, and more importantly is not supported by the data.	

The Watermark Final Environmental Impact Report

Response to Letters of Comment – Page 56 September 2013

	1 Comments noted. These comments do not address the completeness or adequacy of the EIR. No responses are necessary.	
Shearer-Nguyen, Elizabeth From: Catherine Kastner (odk@kastner.com) Sent: Triday, Abol (06, 2013 7:50 AM Sent: DSD EAS Co: DSD EAS Co: DSD EAS Co: Watermark is a Great Idea	I received the call regarding the survey for the grocery store Lazy Acres - it will be a great addition to the adjoint to the use of this Project. We need this Project. The free data is the store of t	I

Response to Letters of Comment – Page 57 September 2013

ETTERS OF COMMENTS AND RESPONSES			Comment Noted. Please refer to response no. J-4 for an explanation of the project access.	Comments noted. These comments do not address the completeness or adequacy of the EIR. No responses are necessary.	Comments noted. Fire safety is addressed in Section 5.13, <i>Health and Safety</i> , of the EIR. As described in Section 5.13, the project site is bordered on the south by existing open space area. Two brush management zones (one	of 65 feet and one of 35 feet) buffer the open space area from project development. The requirements for these brush management zones are in addition to those mandated by California Public Resources Code in Section 4291 of the City of San Diego Landscape Technical Manual. These brush management zones would minimize the risk of exposure of people or buildings to potential wildland fires that may occur in surrounding open space areas and would effectively minimize exposure to	wildland fire risk. Project impacts associated with wildland fires would not be significant.
<u> </u>			0-1	0-2	0-3		
	From: Ruriko Kusumoto [mailto:rsakurauchi@san.rr.com] Sent: Saturday, April 13, 2013 11:43 PM To: DSD EAS Cc: President@ScrippsRanch.org Subject: The Watermark - Project #180357	I live in Scripps Highlands and I drive by the lot proposed for this Watermark project almost every day. After reviewing the report on this new retail center, I have some major concerns and am opposing the current proposal.	<b>D-1</b> First of all, even without any commercial retail, the traffic in the moming commute hours makes it difficult to make a left turn from Scripps Highlands onto Scripps Poway Parkway. I also notice a similar situation in the afternoon commute hours. The proximity of the project to the 15 freeway will probably impact commuters who exit the freeway onto Scripps Poway Parkway. Even with the proposed extra right turn lane into the development, especially during the commute hours, I could see that there may be a backup of traffic exting the freeway. So, this development will affect both the commutes entering the freeway as well as those exiting the freeway.	Secondly, I am concerned about the safety and well being of our neighborhood. Recently, O-2 there has been an increase in break-ins in our neighborhood, and I am concerned that bringing in extra retail may invite people who do not live in the area to wander into our neighborhood. Although most of the employees of Medimpact have been respectful, I do see an increase in people walking in our neighborhood or using the small community parks and benches that we as a homeowner's association pay to maintain. Multiply that by the number of people who shop and patronize the proposed shops. As a homeowner, I	need to insure that the money I put into maintaining the common areas are beingused for us. Perhaps, a security gate needs to be placed so that only people that live in the neighborhood have access to our common areas.	Thirdly, I have a concern about fire safety. Scripps Ranch was impacted by a devastating fire in 2003 that destroyed many homes in the Pomerado Rd. area. From a personal friend whose home was destroyed, she explained that fire crews were not allowed into her area. Here is a portion of her email The 2003 Cedar Firestorm burned down our house and 5 others in our little cul de sac. The next street street over is Pinecastle and Walnutdale which is a long huge cul de sac	

			TERS OF COMMENTS AND RESPONSES
		0-4	Comments noted. Please see response no. O-3 above.
	with 40-50 homes with only 1 entry/exit street. Those 40-50 homes ALL burned down because San Diego Fire Dept. will not permit a fire truck to enter a dead end cul de sac. The fire dept. does not want to take a chance for injury to firemen and equipment from	0-5	Comments noted. Please see response no. O-3 above.
	the flames and explosion of homes. Also if the fire was at the entry of the cul de sac, there was no other escape exit.	9-0	Comments noted. Noise impacts associated with the project are addressed in Section 5.7 Naice of the FTR To order to evaluate portential noise
O-4	In 2007 when we had to evacuate our home again. We have 2 exits out of our neighborhood but the line of cars/traffic to get out on Pomerado Road to freeway 15 took foreveran hour!! If it was a real fire in the SR, I think we would be toast.		impacts associated with the project, a <i>Noise Analysis</i> was prepared and is included in Appendix F of the EIR.
0-5	With such a high density of stores, movie theater, hotel, and parking structure, on a hillside with a lot of brush, fire safety should be a major concern.		As concluded in Section 5.7, the proposed Watermark project would not result in significant noise impacts associated with construction and/or "".
9-0	On a fourth note, I am concerned about the increased noise levels that we will experience with increased traffic in our area. When the pumpkin patch is open during the month of October, I do hear the extra noise into the evening hours of not only people at the pumpkin patch, but also with those people driving into our neighborhood.		operation. The project would not result in exposure of people to noise levels that exceed the City's adopted noise ordinance and would not be in conflict with the City's noise guidelines and/or the Noise Element of the City's General Plan. No significant noise impacts would result from the proposed project.
0-7	Lastly, sadly with the economy, there are already many vacancies in the current shopping centers of Scripps Ranch. Even in the Vons shopping center, we recently lost Baskin Robbins which should be a fixture in almost every community. If our community cannot even support a Baskin Robbins, do we need another shopping center with high end stores and restaurants? I've scen restaurants come and go in the Trader Joe's shopping center as well.	0-7	Comments noted. The purpose of the EIR is to evaluate the potential for significant impacts on the environment resulting from the proposed project. CEQA identifies a significant effect on the environment as "a substantial, or potentially substantial, adverse change in any of the
	Thank you for taking our concerns and comments. Please reconsider building the Watermark project. I vote "No," for this project.		physical conditions with the areas affected by the project." (CEQA Guidelines Section 15382.) The EIR addresses the physical changes to the environment that would result from implementation of the proposed
	Sincerely, Ruriko Kusumoto		project. Additionally, in accordance with CEQA Guidelines Sections 15064(e) and 15131, economic or social effects of a project shall not be treated as a significant effect on the environmental, unless a physical change in the environment is caused by the economic or social effects of a project; or if the economic and social effects resulting from a physical change in the environment create a significant effect. Neither of theses situations pertains to the proposed project.

Response to Letters of Comment – Page 59 September 2013

	<ul> <li>P-1a Comment Noted. Please refer to response no. M-74 and the 'IIA for information regarding the anticipated project impacts and mitigation.</li> <li>P-1b Comments noted.</li> <li>P-1c Comments noted.</li> <li>P-2 Comments noted.</li> </ul>
Shearer-Nguyen, Elizabeth From: VALORIE MCCLELLAND [valmac0214@shoglobal.net] Sent: Sunday, April 07, 2013 12.22 PM To: DSDEAS Cc: president@scrippsranch.org Subject: The Watermark - Project #180357	<ol> <li>1.3 Elservigue</li> <li>1.4 Elservigue</li> <li>1.5 Elservigue</li> <li>1.6 Elservigue de plans for the vareaux shoping complet 1 have configue to the variant of the plans for the normal for</li></ol>

Response to Letters of Comment – Page 60 September 2013

<ul> <li>3.1 diversion of the variant integration of the control model of the control model and the control model of the control model and the control model of the control model and the control mode</li></ul>	7.0.1. Moseman@adicoml Entris Study April 14, 2013 7:45 AM Tor DS EAK Tor
--	--

l

Second, while I concede that public safety and crime are not areas creaming blocking whether to approve or disapprove this proposal. Applicably, the DEIR does not appear to acknowledge the uniserate 1) Northbound, nor the important fact that this pathway is interstate 1) Northbound, nor the important fact that this pathway is interstated by SPF. 11 Applicit this regionally that rules are allocated and an interstate 1, Northbound, nor the important fact that this is negative to a the new personally with the rules are allocated interstates 1) Northbound, nor the important fact that this pathway is interstated by SPF. 11 Applicit this region as exceptionally hazardous any sister and 1 have personally with the lake the downward sloping of the blockley deterrian pathway, has lead to these domarcous interstates 1, and intersection would be needed to access the Maternark formers. I may witness a the statistic for the ground of the blockley deterriant pathway. The market are addi- to the ground additionally. Thave learned that businesses placed near to this area, I may witness and businesses blaced near areas worse bodily injury accelents involving whicles and gedestrians. Additionally. There learned that businesses placed near to this area, I may witness and other hybrid complex, which would include a bloce Nuch instant (Mathing to this area, I may vitters and while fact the formed areas access locations and "quick getaways. I mantion to this areas, I may vitters and and a fact the to the statistics of the conveniently positioned treeway on and off remains are diversite fact to the ground and and to the intering of merida as avait to this areas. I may vitters and while fact the statistics of the block balance and while fact the and addition the statistics of the conveniently positioned treeway on and off remains and while fact that the SDP and addit reports sought the public's assistance in finding the factor deduced to the project planning, I point out the many vacancies and place is extremely questioned. Inthe fact	<b>2-3</b> Comments noted. See also response nos. M-35b and P-1b.	<b>2-4</b> Comments noted. As part of the EIR review process, the City's Fire-Rescue and Police Departments were contacted. The result of this coordination is presented in Sections 2.6, <i>Public Infrastructure and Services</i> , and 5.14, <i>Public Services and Facilities</i> , of the EIR. As concluded in Section 5.14, the proposed project would not have significant impacts on these services or require construction of new public facilities.	<b>2-5</b> Comments noted. See also response nos. M-33 and P-2, above.
2-3	3-3 Second, while I concede that public safety and crime are not areas formally assessed in the DEIR, I feel they are paramount factors to consider when deciding whether to approve or disapprove this proposal. Applicably, the DEIR does not appear to acknowledge the long standing bicycle/pedestrian pathway that runs parallel to Interstate 15 Northbound, nor the important fact that this pathway is intersected by SPP. I highlight this region as exceptionally hazardous saw sister and I have personally withessed two near accidents involving bicyclists and vehicles here. I believe the downward sloping SPP exit off the Interstate 15, along with the lack of commanding signage of the bicycle/pedestrian pathway, has lead to these dangerous situations. In both events, the car drivers seemed to look solely to their left when making their right hand turns on to SPP and failed for section provisits, and show and the none of the none of the bicycle bedestrian pathway is situations. This events, the car drivers seemed to look solely to their left when making their right hand turns on to SPP and failed for section bicyclists. As they show and the none of the none of the section for the section failed for solely the for section the interstate to see the bicyclists.	<ul> <li>Project: I fear if approval is given for the mixed use shopping complex, which would include a hotel with transitory guests unfamiliar complex, which would include a hotel with transitory guests unfamiliar complex, which would include a hotel with transitory guests unfamiliar complex, which would include a hotel with transitory guests undamiliar complex, which would include a hotel with transitory guests undamiliar complex, which would include a hotel with transitory guests undamiliar to the ground or even worse boily injury accidents involving vehicles and pedestrians.</li> <li>2.4 even worse locations and "quick getaways." I mention that both banks on SPP, Bank of America and Wells Fargo, less than 1/4 mile from the proposed project, have been robbed several times as media reports sought the public's assistance in finding the suspects. I believe that the San Diego Police Department (SDPD) and other law agencies can confirm the crime statistics of the conveniently positioned targets</li> </ul>	<ul> <li>b. The standard of the standard set o</li></ul>

<b>Q-6</b> Comments	
"Nokia" has left The Flaza at SN on Scripps Summit Drive, while The Finnacle Medical Building on These are just a few These are just a few These are just a few asamples that offer prompt opportunities for corporate and retail businesses, without the need for any degradation examples that offer prompt opportunities for corporate and retail businesses, without the need for any degradation to the emitromment (air, water, land) and numerous mandates for starting new developments. I might add that if these vacancies are eventually filled, traffic congestion will grow exponentially as the "Nokia" Duilding in itself can employ hundreds of employees if not more. Tastly and in closing, I respectfully urge all Gity Council Members and decision-making authorities of this project, to seriously conremplate the identified might help authorities of this project, to seriously conremplate the identified and any alternatives presented. I truly hope after such careful review of the complete information that all will vote to deny The Watermark Project and thus conclude its negative impacts consideration. Sincerely. Christina Mosema	

		R-1	Comment noted.
	From: jmoseman@san.rr.com [mailto;jmoseman@san.rr.com] Sent: Sunday, April 14, 2013 9:19 PM To: DSD EAS Cc: president@scrippsranch.org Subject: DEIR	R-2	Comment noted. The Reduced Intensity – 17 Percent Reduction in Trips Alternative would include the existing MedImpact facilities on previously approved Lot 1, the future approved development on Lot 2, and development of the remainder of the site in a manage similar to the
	Dear Environmental Planner,		proposed project but without development of the hotel (approximately
R-1	After reviewing the DEIR, I voice strong objection to the change in zoning of the proposed Watermark Project.		90,540 square feet) and the office building (approximately 132,007 square feet). The Site Plan for the proposed project shows the hotel located in
R-2	Even after eliminating the hotel and commercial business – rezoning proposals this would only decrease the predicted 'increase trips' by 17%. This means the remaining 're zoned Entertainment' proposal brings the wallop of increase traffic.		the southwest corner of the project site (Building L). This alternative would eliminate the hotel and provide surface parking where the hotel
R-3	Yet the mitigation measures in the report describe adding left turn lanes with queuing modifications. This does not account for the fixed freeway access and perhaps even increased freeway traffic. Getting you faster on a ramp does not mean faster access to freeway especially during rush hours or Friday and Saturday nights. This quickly will back unto the surface streets as those who live here already experience on Mira Mesa Blvd just west of the I-15.		would have occurred. Under the proposed suc rian, onlice uses would occur on floors two through 6 of Building M. Because this alternative would eliminate office uses in Building M, Building M would change from a six-story retail/office building to a two-story retail building. With the
R-4	Also they did not take into account southbound traffic from the I-15 turning east on Scripps Poway except to reduce medians- again a very limited measurement with rapid filling.		elimination of these elements, this alternative would result in approximately 17 percent less total trips than under the proposed project
R-5	Remember Entertainment usage carries a different mind set with its users. They are not going to work but to play, they can be less cautious and more distracted.		(15,341 with this alternative compared to 18,552 resulting from the proposed project).
R-6	The analysis did not include or mention the low profile but heavily used pedestrian and bicycle X-ing at the Northbound 1-15 exit at Scripps Poway. This connects to the Knott memorial and the Fransquitos preserve. I have observed numerous close calls and read of several collisions from vehicle right turns unto Scripps Poway vs. crosswalk users. Actually, motor vehicles have a large visual impairment from the freeway superstructure at this location that impairs both west and east bound turns.	R-3	Comment Noted. Increased freeway traffic was projected and is shown in Tables 1-7 thru 1-12 of the TIA. Mitigation for the project is shown on Figure 1-1 (re-printed below as <i>Scripps Poway Parkway / 1-15 Reconfigurations</i> , with resonce no. M-7) and discussed on page 1-7 of the TIA. Mitigation
R-7	To significantly increase the traffic is truly asking for safety problems –1 am sure the City does not want to be liable for poor planning and structural inadequacies. Both Police and Fire are predicting increase response times if such a project is approved. I chuckled when the report mentioned to keep the speed limit to 15 mph to prevent noise pollution. Good luck with law enforcement.		is further discussed in Section 14.7 of the TIA. As discussed in Section 5.2 of the DEIR, significant project impacts are expected to remain after project mitigation.
R-8	Also there are Artesian springs at the hairpin turn of Scripps Highland. I have seen foxes, bobcats, kites and osprey in the vicinity. These are grassland inhabitants. For some reason the report does not consider these creatures significantly impacted. Direct downhill construction can definitely affect such a unique habitat and water supply. I would hate to see the rodent population go unchecked.	R-4	Southbound traffic from I-15 to eastbound Scripps Poway Parkway is shown on Figure 3-4 of the TIA as well as subsequent intersection AM/PM peak hour traffic figures in the TIA. Southbound traffic was fully accounted for in the analysis.
R-9	It seems this analysis was hermitically sealed without adequately envisioning practical human usage.		
	The least impact would be to keep the current zoning.	R-5 O	Comment noted.

ETTERS OF COMMENTS AND RESPONSES	Please see response to comment B-13 and B-16. As discussed in these comments the existing pedestrian/bicycle pedestrian refuge crossing Scripps Poway Parkway at the I-15 northbound ramps will remain and a pedestrian/bicycle pathway on Scripps Poway Parkway will be constructed with the project. See also response to comment B-3. The turn lane into the project will not begin until approximately 100 feet east of the intersection to avoid conflicts with bicyclists.	Comments noted. See also response no. Q-4, above.	During the environmental review for the project, site surveys were conducted, and an evaluation of all sensitive wildlife species with the potential to occur onsite was performed. Foxes, bobcats, kites, and osprey were not observed on the project site nor were they found to have the potential to occur on the project site. Furthermore, no grassland habitat or artesian springs occur on the project site. The only habitats of value that occur on the project site include Coastal Sage Scrub/Chaparral and Seeded Coastal Sage Scrub, for which the project would result in no impact.	It should be noted that the project abuts an existing open space preserve, which would not be impacted by the proposed project. Although it is possible the wildlife species noted in this letter by the commentor occurred in this off-site open space area, the open space supports coastal sage scrub and southern mixed chaparral, not grassland habitats. Furthermore, compliance with existing regulations and mitigation measures identified throughout the EIR ensure that on-site construction activities will not impact off-site areas, and no impacts were identified.	Comments noted. These comments do not address the completeness or adequacy of the EIR.	
	R-6	R-7	R-8		R-9	

Response to Letters of Comment – Page 65 September 2013

NSES	
ESPOI	
ND R	
NTS A	
MMEN	
PF CO	
ERS C	
L L L L	

Response to Letters of Comment – Page 66 September 2013

			TTERS OF COMMENTS AND RESPONSES
	Shearer-Nguyen, Elizabeth		
	From: mintchip@san.r.com Sent: Tuestedy, April 02, 2013 11:19 AM To: DSDEdds, April 02, 2013 11:19 AM Co: president@scrippsranch.org Subject: The Watermark - Project #180357	۲ ن	
S-1	As a 77-year resident of Scripps Ranch Villages (Miramar Ranch North), it is quite distressing to read in The Watermark Daff EIR that the proposed Watermark Project will, for the next 17 years and byond (2030 is the year used in the draft EIR), severely impact traffic flow to unacceptable levels (E and F). It is inconcivable to me that if the City of San Diego has defined levels of service ((LOS A+F) related to traffic flows, and turther sipulated which levels are unacceptable (E constrained in the City of San Diego or the community of Scripps Ranch Villages (Miramar Ranch North) would even consider moving (orward with such a project as The Watermark that is defined as having an effect that is "cumulatively significant and unitypated impacts associated with traffic" (page 10.33), the result of which will be to not only incorvenience buf endanger the safety of motorists and pedestrians by moving abied with. The Watermark Project as proposed.	-	Comments noted. Traine impacts associated with the proposed project are addressed in Section 5.2, <i>Transportation/Traffic Circulation/Parking</i> of the EIR. In order to assess the traffic impacts of the proposed project, a <i>Traffic Impact Analysis</i> was prepared and is included in Appendix C of the EIR. As presented in the <i>Traffic Impact Analysis</i> and concluded in Section 5.2, the Watermark project would result in significant direct and cumulative traffic impacts. Mitigation measures are proposed to reduce
S-2	Of particular concern is the draft EIR's complete disregard for traffic patterns on the section of road that is Eastbound Scripps Poway Parkway between the Mercy Road underpass and Scripps Highland Drive. To create an entrance and exit to the proposed Watermark Project at this location definitely does create an increased hazard which is not assessed in the draft EIR. Motorisis exiting the 1-15 morthbound at Scripps Poway Parkway already cause accidents by turning right (eastbound) at this intersection. Adding an entrancedual along that section of road on eastbound Scripps Poway Parkway will surely cause a treemedous increase in accidents and injuries, and potentially deaths. The draft EIR defines this entry/oxit point as a "solution" to the question of energency access, but it does not address the hazard it compounds at that section of eastbound Scripps Poway Parkway.		impacts; however, traffic impacts would not be reduced to below a level of significance and a significant, unmitigated traffic impact would remain. CEQA Section 15093(b) requires that the decision-maker adopt a Statement of Overriding Considerations when the decision-maker decides to approve a project that would result in the occurrence of significant impacts which cannot be avoided or substantially lessened. The
S-3	The draft EIR in section 10.0 Alternatives, fails to define the resulting LOS associated with each Alternative, therefore it is not possible to adequately assess the transportation/traffic circulation of each alternative. The current proposed Watermark Project SHOULD NOT proceed.		Statement of Overriding Considerations provides the specific reasons to support approval of a project even when significant impacts cannot be fully mitigated.
S-4	Based or current draft EIR, certainly Alternative 1 is the most attractive and favored for consideration in this resident's optinion. With any of these proposals, how is the catch phrase of "country living" truthfully touted on license plate frames in our	S-2	Please refer to response nos. R-6, B-13, B-16 and B-3, above.
	communy? Sincerely, Tina Nelsen 1141 Ann Arbor Lane San Diego, CA 92131 San Diego, CA 92131	S-3	The EIR includes an evaluation of traffic associated with each alternative, including total cumulative, driveway, and peak hour trips, which provides sufficient information "to allow a evaluation, analysis, and comparison" of each alternative with the proposed project, as required by CEQA Guidelines Section 15126.6(d).
		S-4	Comments noted. These comments do not address the completeness or adequacy of the EIR. No responses are necessary.

		<u>ا</u> ت	TTERS OF COMMENTS AND RESPONSES
		1-1	Comments noted. As presented in Section 3.0, <i>Project Description</i> , of the EIR, the project would result in the target development intensity of
	From: Mark Sampson [mailto:marksampson@sbcglobal.net]		approximately 204,112 square reet of office space. This development intensity includes the 350,743 square feet of office space currently
	Sent: Sunday, April 14, 2013 5:59 PM		approved for and partially developed on Lots 1 and 2 of MedImpact.
	To: DSD EAS		The arrange of an DID is not to identify the broadfite of a surject for a
	Cc: President@ScippsRanch.org		The purpose of an ELIN is not to identify the benefits of a project for a particular area community or region Instead CEOA requires that the
	Subject: The Watermark - Project #180357		EIR analyze the potential for significant environmental impacts to result
T-1	Thank you for the opportunity to comment upon the draft environmental impact report for the proposed Watermark mixed-use development planned for the intersection of 1-15 and Scripps Poway. Parkway. There are many elements of the planned evelopment that appear to have heartif for the horth City communities and consumers. A review of the EIR gives rise to a number of quastioning the proposed development project. The issues and related		from the proposed project. Therefore, the project's benefits are not addressed in the EIR. CEQA Section 15124(b) requires that an EIR include a statement relative to the objective sought by the project. For
	questions are noted below: questions genoreals allow for the construction of up to 350,000 square feet of office space. Approximately 150,000 square feet has been built and is occupied. Table 3.1 <i>Proposed Project Development Intensity</i> of the draft Watermark is outline as a proposal for the development of in a ceese of 2000 square feet of Office space or 150,000 square feet more than originally approved. Additionally, there is a total targeted development intensity that approaches		the Watermark EIR, the <i>Project Objectives</i> are presented in Section 3.2, <i>Purpose and Objectives of the Proposed Project.</i>
	nearly one million square feet in a relatively small footprint. What benefit should the North City communities expect in exchange for increases in the developer entitlement? The Watermark drapt FIR contains is its cardinal 3.2 <i>Pursoses And Objectives of the Project</i> the objective of	T-2	
T-2	"Implement design guidations community the provision structure in Provision and Section 2014 and 2014	l	The project includes Design Guidelines that would be approved as part of the Planned Development Permit. The intent of these discretionary
	Privators setuptes reaves; each setup encode contract containing the setup end and accurate for the senate What assumpts are available that Watemmark would truly be a landmark for the community and offer high quality design and aesthetics similar to what exist in other high end retail and mixed-use centers such as Santana Row in San Dose, CA, page Yaterei In Emeryville, CA, or The Growy, Los Angeles, CL, thew does the community arears that the project as illustrated in architecturate readerings is actually the development that is placed in the ground?		actions is to ensure that development occurs in the manner that is shown on the plans included in the approval package for the project.
T-3	Additionally, Section 3.2 Purpose And Objectives of the Project cites the goal of providing a community guther place. "Provide quasi-public space for community use in the form of a pedentrian placa as a focal point for the project, which		The Site Plan, Landscape Plan, Vesting Tentative, and associated maps
T-4	would function as incly gathering place for visions, employees, and neighbors, we have a pedestrian plaza? What exactly are the proposed civic benefits? Is the sole public space for community use a pedestrian plaza? What amenifies may become awalke to service as a point of attraction for both adult and juvenic community members? The proposed protect will serve as a gateway to Martamar Ranch Neuh, Scripts Ranch and Devay communities. Additionally, Watermark, File thut, will be viewed by hundreds of thousands of communes weekly from north- and		In exmons become apart of the approval package and would be used by staff in reviewing and approving future building permits for the project. If building permits result in substantial changes from what is approved,
	southbound lanes of 1-15. How can the North City communities ensure that the Watermark development is in fact a welcome. attractive addition that is relactive of community values. Athracter and expectations. especially in view of the requested building height enstroidon waives and proposed land use intensity? Thank you for the opportunity to comment on the proposed Watermark project and for your considerations of my		subsequent review by staff would be required and modifications to project approvals may be necessary, which would initiate new discretional actions,
	concerns. Sincerely, Mark Sampson 10636 Frank Daniels Way		reviews, and approvals.
	Sun Diego. CA 92131 marksampson@sbegiobal.net	T-3	As presented in Section 3.0, <i>Project Description</i> , of the EIR, the proposed project would provide for a large (approximately 1.7 acres) plaza space
			designed to accommodate community gatherings and events. This space
			planting (Figure 3-11, Primary Public Plaza Landscape Design). Seating
			elements, and thematic elements, would be located surrounding the grand
			lawn/ event area available for public use.

Response to Letters of Comment – Page 68 September 2013

+ +	As discussed in Section 5.3, <i>Visual Effect and Neighborhoal Charater</i> , of the EIR, the Miramar Ranch North Community Plan Design Element provides specific recommendations for the development of property within the Marcy Interchange Gateway, which encompasses the project site. This section of the Community Plan calls for " <i>the sentitive development of the Mary interchange Gateway, which encompasses the project site.</i> This section of the Community Plan calls for " <i>the sentitive development of the Mary interchange Gateway, which encompasses the project site.</i> This section of the Community Plan calls for " <i>the sentitive development of the interchange area at an attractive gateway into the community.</i> " Sensitive development and multi-family developments located to the northwest of the interchange area. To implement the goal of creating a coherent and signature design statement at this gateway to the Miramar Ranch North community, the office building the use of natural stone, to achieve compatibility in the implementation of the office campus. The modern design is achieved through an extensive use of galss with metal and accents. Building attriculation employs the subtle use of offsets and curves to provide relief from standard rectangular building stere of offsets and curves to provide an offset view from the freeway and the pindary drive, as well as shield the view of the parking structures that serve each building. (See Figures 5.3-2.5.5.2.b), and 5.3-3.b). Additionally, the project has been reviewed by City's General Plan. See also response T-2, above.

		<b>V-1</b> Comments noted. Also see response no. S-1, above.					
Shearer-Nguyen, Elizabeth From: Connie Simonsen (dstimonsen@san.r.com)	Sont: Saturday, March 30, 2013 7.42 PM To: DSD E-AS, presiden@scrippsranch.org; Kersey, Mark Ksimonsen@san.rr.com Subject: Watermark draft EIR	I am writing regarding the Watermark project after having reviewed the draft EIR. While I understand the desire to develop the site, I am very troubled by the conclusions in the draft EIR regarding transportation/traffic criculation/partial. After having seen the back-up of traffic on Scripps Poway Parkway, particulary weatbound vehicles turning left from Scripps Poway Parkway onto the I-15 South, the fact that this project is going to greatly increase traffic flow is unacceptable. The following language from the report should raise a huge red flag. Having seen a similar growth the interscence of Westview Parkway and Mira Mesa Boulevard as a result of the still-increasing development, and the interscence of Westview Parkway and Mira Mesa Boulevard as a result of the still-increasing development, and the extremely dysfunctional level of traffic there, I do not feel that it is in the community's best interest to similarly invite high-traffic retail uses immediately adjacent to the freeway, when there is no or insufficient mitigation to accompany it.	I would very much like to hear your thoughts on the matter. Thank you.	Kevin Simonsen ksimonsen@san.r.com	Mitigation measures are proposed (Section 5.0) to reduce all P significance, with the exception of <b>Transportation/Traffic Ci</b> and unmitigated impacts relative to <b>Transportation/Traffic C</b> on several street segments of Scripps Poway Parkway. The att Appendices document the basis for the above Determination.		-

Comments noted.			
- × - ×			
Shearer-Nguyen, Elizabeth           From:         Annette Sorensen (SorensenMarkeling@att.net)           From:         Funder, March 20, 2013, 308 PM           From:         Funder, March 20, 2013, 308 PM           For the Sorensen (Sorensen/SorensenMarkeling@att.net)         Funder, March 20, 2013, 308 PM           Font:         Funder, March 20, 2013, 308 PM           For the Sorensen (Sorensen (Sorensen/Markeling@att.net)         Funder, March 20, 2013, 308 PM           For the Sorensen (Sorensen (Sorensen)         Funder, March 20, 2013, 308 PM           For the Sorensen (Sorensen)         Foresteinen(SocripsRanch.org)           Co:         President of Scripps Ranch or 30, years and this project. Will bring many much needed services to our arrea that in market and dring. Scripps Ranch or 30 years and this project will bring many much needed services to our arrea that in the toroid be considered fine dring. Scripps Ranch in a could be considered fine dring.	While I'm aware that a number of people who live next to the proposed shopping center are concerned about traffic, the traffic studies that Suderry Properties has provided should alleviate these concerns. I am hopeful that this project can move forward and provide Scripps Ranch residents with much needed services and entertainment.		
M-1			

Response to Letters of Comment – Page 72 September 2013

l

		Comments noted.		Comments noted.	
		X-1		X-2	
Sullivan Family 950-130 Scripps Vista Way San Diego CA 92131 April 9, 2013 To: The City of San Diego Development Services Department	Attn: E. Shearer-Nguyen Ervironmental Planner City of San Diego DSC 1222 First Avenue MS 501 San Diego, CA 92101 SuBJECT; draft E.I.R. <b>The Watermark</b> , Project No. 180357/SCH No.2010091079	Dear Planners, This is in response to the draft E.I.R. for the Watermark project located in the northwest area of Scripps Ranch in District 5. Currently a bicycle path parallels the east side of 1-15 connecting the Scripps Miramar Ranch and the Miramar Ranch communities. The bike path joins with bike lanes on Erma Rd. at the south and Scripps Poway Parkway on the North. The planned	waternark project is bocated on the nort end of use norwaw with a significant entimeder of families residing in the multi-family residential communities located on the south end at the Erma Rd, access-less than 1 mile from the project. I have lived in Scripps Townhomes, a 164 unit, multi-level-resort style townhome community located adjacent to the bike path since 1998. I have served as Board President for much of the past decade and have advocated on behalf of the numerous communities in this area to preserve our high quality of life and reduce the negative impacts of the ongoing development of our region. This has included mitigating the addition of lanes to the freeway by installing a pollution barrier, providing community input for the location of the anticipated DAR as well as numerous Water Department relining project improvements. Working closely with Caltrans, the City and local community for many been successful in improving many areas of concern that will serve this community for many vears to come. As a result of my ongoing commitment to this community of the proposed of the proposed project 1 believe I have a direct interest in the Watermark project as proposed.	After attending the MRNPC Workshop on the project in Feb 2013, followed by discussing the project with local leaders and meeting on-site with the Caltrans Bicycle and Pedestrian <i>Coor</i> ditator, I have concluded that potential problems exists that are not being considered, as well as opportunities to significantly improve the use of non-motorized transportation to enter/exit the project ultimately reducing the already anticipated negative impacts. The DEIR concludes that "the project would result in significant environmental impacts to the following areas: <b>Transportation/Traffic circulation/Parking, and Air Quality</b> , Obviously, the most effective way to reduce all of these impacts is by reducing the total amount of vehicles actually entering the project with the goal of still being able to provide easy and safe access alternatives. I believe specific improvements in bike and pedestrian access are solutions that should be added to this project prior to approval by the City.	1
		X-1		X-2	

The Watermark Final Environmental Impact Report

Response to Letters of Comment – Page 73 September 2013

and exprising structure. This subject of the problem of the probl
e from Erma Ru to the center of toxic throughout that both exercised on the outper proposed on Scripps segred to proposed on Scripps area and la bilities. This access point at the proposed for the editional through through the project that the proposed for the distance of the edit point intrues. Without an access from and loger that would appeal to the distance that would be an opportunity to the analytic that the project.
a rearr of the project. In the trip on ame location in the project will believe the additional time and A, Terraces HOA, Scripps Westview A, Terraces HOA, Scripps Westview A, Terraces HOA, Scripps Westview A, Terraces HOA, Scripps Mestview A, Terraces HOA, Scripps Vestview A, T

			TERS OF COMMENTS AND RESPONSES
	reach of food and entertainment opportunities. The City of San Diego has stated this to be their goal. Caltrans has policies in place and initiatives to "Promote alternative transportation" also providing additional guidelines in the PDPM for proposals after freeway construction. It appears well understood promoting pedestrian use is ultimately best for all. Now all we need is the well west and our limited resources are construct or any ego addition are guidelines in the address to any construction.		
X-5	An extraordinary amount of time, money and energy is being spent on traffic analysis and considering how to get the most vehicles in and out of this project but I see very little energy in developing and encouraging non-motorized travel to this project. The Bikeway along this project is the only reasonable and safe place to develop high volume non-motorized use. This the only real solution of adding customers to the project that has zero negative consequences to the traffic and parking problems, yet it is all but ignored in the development plan. My hope is that the addition of a pedestrian access/ground level valkway from the bike path through the parking structure is given thoughtful consideration and pressure is placed on the developer to create a plan that is as good, or better than what I have proposed to encourage more non- motorized access.	X-5 (	Comments noted. Please see response no. X-4, above.
	My specific comments/concerns:	7 7	
X-6	<ol> <li>Why has a pedestrian/bike access from the bikeway to the proposed parking structure not been included in this project?</li> </ol>	- -	lease see response no. A-4 above.
X-7 X-8	<ol> <li>Page 2-9 Figure 2-5 Surrounding Land Uses - This map shows the significant Geographical Land Uses in the areas for multiple miles to the East and portions of the south. However, the numerous multi-family housing communities to the south, less than a mile away (at the south end of the bikeway of Erma Rd.) have not been included. Why were these communities with thousands of residents onitted from the report?</li> <li>Wouldn't the residents off Erma Rd. benefit the most from a pedestrian friendly street, site</li> </ol>	X-7	There was no intent to omit land uses located farther south of the project site outside the Miramar Ranch North community. The purpose of Figure 2-5 was to show the land uses adjacent to and in the immediate vicinity of the proposed project.
	and building designt Most of the other commuttees and are faced with safety issues and the physical challenges of the terrain that will ultimately discourage non-motorized access. Why are there no plans to encourage pedestrian traffic from this quadrant?	8-X	The figure on the following page shows the location of residential
X-9	4. The Intersection at the off ramp of 1-15 and Scripps Poway Parkway is already a known hazard to pedestrians and cyclists. The addition of a driveway in such close proximity to the off ramp will only add to the problems with distracted drivers and increase the risk to non-motorized travelers. With movile theaters, dining and possibly a bowling center it should be assumed children and young adults will travel to the project either by foot or bicycle. Why has the developer not presented solutions to allow easier and safer access for young or disabled people in this area? It makes no sense to funnel pedestrians into a main entrance that is sure to be busy and at times very congested. Wouldn't an access point prior to the traching that is sure to be busy and at times very congested. Wouldn't an access point prior to the reching that is sure to be busy and at times very congested. Wouldn't an access point prior to the reching that is sure to be busy and at times very congested. Wouldn't an access point prior to the proving data to the proving the adviser to a strength the second period of the second period.		development along Erma Road in the Scripps Miramar Ranch community and the relationship with the proposed project. Residential development along Erma Road is approximately 3.5 miles from the MedImpact site, if accessed using existing public streets, and about 1.5 miles if using the bike path along the I-15 freeway. The project does not discourage pedestrian traffic from the Erma Road area. See also response no. X-4.
	5. Regardness II an access is added to the proposed project, there exists a LUO Claritants sectoack between the bikeway along the freeway and the project limit lines. This is currently a drainage area with why more handscape and no real "design plan" in place. My concerns for	, ,	
X-10	this area include; a) What plans are in place to incorporate the setback area into the aesthetic of the project? Obvious much though has been placed on the amearance of the actual	4-X	Please see response no. X-4 and also responses M-35b and P-1b, above.
	project as you exit the free weavy (see; Site-West Elevation, Conceptual design for project as you exit the free weavy (see; Site-West Elevation, Conceptual design for illustrative purposes- Figure 3-7a Project Elevation. Page 3-16), however I have seen nothing that addresses the specific improvements to this area. Will this simply become 3	X-10	The area addressed in this comment is a combination of slopes within the project site boundaries and Caltrans right-of-way. The proposed project would be responsible for maintaining its slopes, and Caltrans has the responsibility of maintaining its rights-of way.

Response to Letters of Comment – Page 75 September 2013



<b>X-11</b> The project applicant is not proposing a path through the steep the western border of the project site or within Caltrans' righ Neither is the project encouraging unauthorized access throu areas. The developer will be responsible for maintenance of the	Stope and security for the project. Calitatis has responsionity to of-way. X-12 The developer will be responsible for maintenance of the on-site	<ul><li>x-13 No FBA fees are required for the project.</li></ul>	X-14 Comments noted.					
<ul> <li>a 100-foot Caltrans "dead zone" between the freeway/bikeway and the project? The conceptual design shows a clean and landscaped exterior outside the project walls. Is this landscape improvement included in the proposed project or is this a rendering that includes the existing Caltrans setback (that currently looks nothing like the concept drawings) and will see no change?</li> <li>b) Due to this setback and with a lack of a pedestrian access it should be assumed a path/trail would eventually be formed down the slope and into the project (behind or near the hotel?) to cut the travel distance. Will the Project developer encourage and allow this, ignore it or try preventing 17 Wy concern is that an unsafe situation will be unseed to be preventing to the preventing will be unseed to be been to a preventing the project (behind or near the hotel?) to cut the travel distance.</li> </ul>	created due to the lack of anternatives available. c) With numerous eateries, employment/work opportunity and items of value being thrown in the trash link it is highly likely that if this area is ignored the south portion may develop into encampments of sorts. This is regularly seen just to the north in the canyon on the other side of Scripps Poway Parkway. What plans are in place to prevent this type of negative activity from entering our community?	Due to the change in land use designation from Industrial/Business Park to mmercial/Residential Prohibited will any additional fees be collected from the developer to commodate this dramatic change in land use? Dyviously the change in uses has a significant pact on the local area. What was once going to be business people coming and going to work training, has now developed into a proposal that would drive tourists, shoppers and uners to s area. A portion of which I assume will use our local parks, litter our streets and want to local facilities. Which far wa difficual fees or local innovements like a hikeway arcress have	developer offered in order to mitigate this additional burden on our community?	upport smart and responsible development in my community. I believe with some odifications to the proposed Watermark project this could add significantly to the quality of e of Scripps Ranch residents. Obviously these improvements will require a cooperative team or to the tween the developer. Caltrans and the City. Because it's almost impossible to go back d correct even a minor problem in an efficient manner, I leel that working together now and tting the best project before the City Council is well worth the effort.	ank you for your consideration of my concerns.	ncerely,	mes Sullivan ripps Ranch Resident esident-Scripps Townhomes Owners Association	*
		6. or in a co	the	I s miff eff an pu	μĽ	Sil	Jan Sc Pr	

Response to Letters of Comment – Page 77 September 2013

		<ul><li>Y-1 The maps attached to this commentor's e-mail are intended to be used in reference to previous comment letter "X". No responses are necessary.</li></ul>										
Shearer-Nguyen, Elizabeth	From:         James Sullivan (jimsully@gmall com)           Sent:         Usednesday, April 10, 2013 11:27 AM           To:         DSD EAS           To:         DSD EAS           President@scrippsrach.org; Janet Wagner, Dustin Steiner         Basiner           Cc:         President@scrippsrach.org; Janet Wagner, Dustin Steiner           Re:         Project Name: The Watermark Project No. 180357           Attachments:         The Watermark Project- maps showing difference in travel from bikeway.docx	City Planners- If this document can be included in my comments on the draft E.I.R. it may be helpful to those trying to understand my concerns and the benefits of having an alternative access included in the project. Thank you- James Sullivan	On Wed, Apr 10, 2013 at 10:35 AM, James Sullivan <jimsully@gmail.com> wrote: Dear Sir or Ma'am,</jimsully@gmail.com>	Attached are my comments regarding the draft E.I.R. for the Sudberry project located in District 5.	Project name: The Watermark Project Number: 180357	If possible, please confirm receipt.	Thank you,	James Sullivan 9950-130 Scripps Vista Way San Diego, CA 92131 <u>858-437-3520</u>			1	

Response to Letters of Comment – Page 78 September 2013

Response to Letters of Comment – Page 79 September 2013



В	
SN	l
РО	l
ШS	l
2	l
Z	l
S S	l
Ł	l
Ш	l
MC	l
Ŭ	l
ЧO	l
SS	l
Ē	
НШ	
	1

Form:         Christopher W. Todd (Good@MingetTawn Veedmestay, April 03, 2013 11:18 AM Subject:           The Watermark - Project No. 180357 inport:         The Watermark - Project No. 180357 Subject:           2-1         Places accept these comments concerning the referenced project development was released for review February 5, 2013. The DE impacts, "bespite Subterry"s apparent affors to mitigate. Over additionally, Subferry has not included any information concer- ation of the subject of the voignal office building projected to be worse, particularly along Scripps Poway Parkwa Additionally, Subferry has not included any information concer adjoining. These stress are logical overlows for the projected view for the realities englinearing in proven accessibility to bus n does not address this sufficiently.           2-2         Subferry has anticularly along Scripp Fowary Parkwa Additionally, Subferry has anticularly along Scripp Fowary for adjoining. These stress are logical overlows to the projected view for the formage with MTS to improve accessibility to bus n does not address this sufficiently.           2-3         Subferry with MTS to improve accessibility to bus n does not address this sufficiently.           2-3         Subferry way to provide greater visibility and astery for problem at of condemotive grows visibility and astery for production of the stress subferry with MTS to improve scissibility to bus n does not address this sufficiently.           2-3         Subferry with MTS to improve scissibility to bus n does not address this sufficiently.           2-4         Parkwary in foroviding provide greater to stress and to work with MTS to mark with MTS to improver subleberry with all to 1 <sup>-12</sup> . View mark stores. Subferry wo	Wingertiaw.com] 1:18.AM . 180357 . renced project. to allow this dissolvement resolved community sumserf. so here		
Z-1     Please accept these comments concerning the referenced poject An earlier request to amend the Community Plan to allow this d as there would have been anticipated if the original office building development was released for review Fehruary 5, 2013. The DE inposed." despite subderry's apprent efforts on mirgure. Over what would have been anticipated if the original office building pojected to be works, particularly along Scripps Powy Parkwa Additionally, Sudberry has not included any information concer Highland Dive, Scripps Ranch Boulevard, Spring Canyon Rae, Durber traffic engineering study should be required, with furthe Curberry has articulated no special plans for bus service. Sudbo collaborative efforts with MTS to improve accessibility to bus n does not address this sufficiently.       Z-3     Sudberry has articulated no special plans for bus service. Sudberry way to provide greater visibility to the service. Sudbo collaborative efforts with MTS to improve accessibility to bus n poway parkway, the frontage will be widered with construction sudberry will build a 10°-12°-wide multi-modal path behind a 6 does not address this sufficiently.       Z-4     Highland Dive, Scripps Summit Drive, Spring Canyon Road, No Sudberry will build a 10°-12°-wide multi-modal path behind a 6 does or Cypress Canyon Road/Village Ridge Road, all of whis scatters are allow to address this.       Z-4     Highland Dive, Scripps Summit Drive, Spring Canyon Road, No Sudberry kas attractive norisone to vandals at an in- stant continger on address this.       Z-4     Highland Dive, Scripps Summit Drive, Spring Canyon Road, No Sudberry take noting to intered or an or subscript stant to address this.       Z-4     Highland Diver, Scripps Summit Drive, Spring Canyon Road, No subscript stanter to address this.       Z-5	renced project. to allow this devoluement received community sumoor so lone		
<ul> <li>Z-1 An earlier request to amend the Community Plan to allow this development was released for review February 5. 2013. The DE impacts," despine Sudberry's apparent affors to mitigate. Over projected to be worse, particularly along Scripps Poway Parkwa Additionally, Sudberry has not included any information concerning projected to be worse, particularly along Scripps Poway Parkwa Additionally, Sudberry has an included any information concerning projected to be worse, particularly along Scripps Poway Parkwa Additionally, Sudberry has an included to special plans for the world have traffic enginear over at adjoining. These strends are logical overflows to the projected with furthe Tarffic engineering study should be required, with furthe Tarffic engineering study should be required, with furthe Tarffic engineering in proven and adverse this sufficiently.</li> <li>Z-3 Sudberry with will al 0.1-2.1 with the Tarffic engine in Poway Parkway, the frontage will be widemed with construction oncerning to improve visibility to bus no subterry with build a 10-12.</li> <li>Z-4 the tarffic engineering in proven thatsets for the North-South Parkway to provide greater visibility to the existing crossing and the source of states of the North-South Powary Parkway, the frontage will be widemed with construction oncernition Deck of Crypres Submit Dive, Scripps Summit Dive, Scripps Summit Dive, Scripps Poway Parkway. This darkened area o and south or Scripps Poway Parkway. This darkened area to the corridor, with heavy equipment, residents request the reprosed development includes public garding to with a night metadom and the source of the North-South Stotes. Subberry with all a 10-12.</li> <li>Z-4 the Park way to provide greater visibility to the existing crossing taffit.</li> <li>Z-4 the Park way. The reason was the source of a south or Scripps Powary Parkway. This darkened area o and south of Scripps Powary Parkway. This darkened area o subset of an attractive misunce to vandus or futoe with a large prime resource and t</li></ul>	to allow this development received community sumort so long		
<ul> <li>Z-2 Sudberry has articulated no special plans for bus service. Sudb does not address this sufficiently.</li> <li>Sudberry is evaluating in-provement flashers for the North-South Parkway to provide grater visibility to the existing consensing in Provement flashers for the North-South Parkway the fromage will be widemed with construction Southerry will be a 10-12.</li> <li>Z-3 Sudberry si evaluating in-provement flashers for the North-South Parkway to in provise grater visibility and safety for pedestrian cross Highland Drive, Scripps Summit Drive, Spring Canyon Road/ty Ridg Ridd.</li> <li>Z-4 Recent CalTrans mitigation has left a nighttime darkened area to and south of Southerry should be required to mitigate increasing taffi DEIR does not appear to address this.</li> <li>Z-4 Recent CalTrans mitigation has left a nighttime darkened area to and south of Southerry should be required to mitigate increasing taffi, lowing a domin of southerry states. Subberry attractive proposed dispution to the corridor, with a large parting structure proposed disputed for the home of already existing public gratering pader. Ho</li> <li>Z-5 The proposed development includes public grathering agace. Ho increased use of already existing public grathering agace. Ho increased development includes public grathering agace. Ho increased use of already existing public grathering agace. Ho increased use of already existing public space (parks, school pla other public grathering spaces). The DEIR hould address Sudber Tank you for your consideration.</li> </ul>	The Daries are severeported in the event of community of participants of the community of t	Z-1	Comment Noted. Please refer to comment response G-9 for information about traffic on neighborhood streets and accessing other freeway interchanges. Please refer to tables 1-1 thru 1-6 for information regarding impacts to intersections and street segments within the project study area including Scripps Highland Drive.
Z-3     Sudberry is evaluating in-pavement flashers for the North-South Pavkwy to provide greater visibility to the existing construction poway Parkwy, the fromage will be widemed with construction Sudberry will build a 10–12- wide multi-modal put behind a 6       Z-3     Sudberry will build a 10–12- wide multi-modal put behind a 6       Parkwy, the fromage will be widemed with construction Sudberry will build a 10–12- wide multi-modal put behind a 6       Z-4     Recent CalTrans mitigation has left a nightline darkened area o and south of Scripps Poway Parkway. This darkened area to and south of Scripps Poway Parkway. This darkened area to and south of Scripps Poway Parkway. This darkened area to with a large parking structure proposed adjecent to the proposed development, residents request the perograed devel with a large parking structure proposed adjecent to the contridu- with a large parking structure proposed adjecent to the contridu- with a large parking structure proposed adjecent to the contridu- with a large parking structure proposed adjecent to the contridu- with a large parking structure proposed adjecent to the contridu- with a large parking structure proposed adjecent to the contridu- with a large parking structure proposed adjecent to the contridu- vition base of an attractive nuisance to vandals or the hom The proposed development includes public gathering space. Ho increased use of altready existing public space (parks, school Jah other public gathering spaces). The DEIR should address Sudb. Thank you for your consideration.       J-5     Thank you for your consideration.       -6     Chris Todd	service. Sudberry should be required to present results of ibility to bus routes for Scripps Ranch residents. The DEIR	Z-2	Please see response no. M-22, above.
<ul> <li>Z-4 Recent Call Trans mitigation has left a nighttime darkened area o and south of Scripps Powy Physerson. B darkened area to and south of Scripps Powy Physerson. B darkened area to and south of Scripps Powy Physerson. Tab. Proposed disput variables. Trans, graffith, lo with a large partiming structure proposed disput variables and heavy corridor. With heavy cupinment, residents request that the bikeway corridor. With heavy cupinment, residents request that the bikeway corridors with heavy cupinment, residents request that the bikeway corridors. The proposed development includes public gathering space. Ho increased use of already existing public spatering spaces. Ho increased use of</li></ul>	e North-South bike lane parallel to 1-15 crossing Scripps Powary ug crossing in that very busy area. For Eastbound Scripps th construction of retaining wall. With the increased frontage, ath behind a 6' planted traffic break. However, Sudberry has pedestrian crossings across Scripps Poway Parkway at Scripps anyon Road/Yry Hill Drive, Scripps Creek Drive/Village Kidge and, all of which provide pedestrian access to schools and retail creasing traffic risks at these intersections at a minimum. The	Z-3	Please refer to response no. G-12. Other than the locations and improvements discussed in the comment, no changes to the existing bicycle or pedestrian facilities are planned as a result of the project. Bike lanes and non-contiguous sidewalks exist on Scripps Poway Parkway
Z-5 The proposed development includes public gathering space. Ho increased use of already existing public gathering space. Jo ther public gathering spaces). The DEIR should address Sudb. Thank you for your consideration. –Chris Todd Scripps Ranch/Miramar Ranch North resident. Mr Ble   Websile   vCard	arkened area on the N-S bike lame right of way adjacent to 1-15 and area is mear the Erms Road residences at the South end of roposed development, community members request that sh, garffit, joitering, and dawhesis in this corridor, particularly to the corridor. Since Subserby will be an the immediate vicinity sikeway corridor be made more assibilitiedly pleasing, less of an also or the homeless. The DEIR makes no reference to this issue.		providing safe facilities for both pedestrians and bicyclists. Striped crosswalks are provided at all signalized intersections discussed in the comment to provide safe pedestrian/bicycle crossing of Scripps Poway Parkway as well as safe access to schools.
Thank you for your consideration. -Chris Todd Scripps Ranch/Miramar Ranch North resident <u>Mr Bio</u>   <u>Website</u>   <u>vCard</u>	ing space. However, the DEIR does not appear to address rks, school playgrounds, trails, picnic spots, Miramar Reservoir, address Sudberry's mitigation of increased use of those spaces.	Z-4	Please see response no. X-4, above.
Chris Todd Scripps Ranch/Miramar Ranch North resident Mr Bte   Website   vCard			
Mv Bio   Website   vCard		Z-5	Please see response no. M-25, above.
	*1		

Response to Letters of Comment – Page 80 September 2013

	ure of the addressess only. E-mull is by privileged. The information contained texpense and the capies of this e- couble for delivering this e-mult to a on distribution, expring a forwarding to recorrect this e-mult in error. Thank		
r W. Todd solvensker & Juskie LLP codway, Suite 1200 CA 2010 131 131 131 131 133 133 133 133 134 134	for if it this e-mult and any associated intuchments is intended for the confidential for the Electronic Communication Privacy Acts (13, LS, C, S) (23, 24), and is legan e analyset to the automy-dense privalge and or may be automy work-product. It field: accessfue for each LS (10) are not an antiferesce or an automotic ad guern respo- dience accession prior for a strain and antiferesce or an automotic ad guern respo- denses or an automotic ad guern response disseminated diseases, you thuse testeled this c-mult in error. Any further review, disseminated is articly prohibited. Please notify us immediately at (61.9) 322-813.1 (2 you that is articly prohibited. Please notify us immediately at (61.9) 322-813.1 (2 you that		N

Response to Letters of Comment – Page 81 September 2013

	Shearer-Nguyen, Elizabeth		
	From: Dearma Wong (deenwong@aot.com) Sent: ThursdaA March 28, 2013 11:26 PM To: DSD Eds Resident@scrippsranch.org Cc: The Watermark - Project #160357		
	Dear E. Shearer-Nguyen,		
AA-1	We are residents of the Tiempo housing which is just up the hill (at the end of Scripps Highland and Traincest) from the MedImpact buildings. We have lived here since 2001 and the community has always been fairly quiet and peaceful, until the MedImpact buildings opened a few years ago. Now we have employees that take walking breaks up our hill while some take smoking breaks at the top of	AA-1	Comments noted.
AA-2	Scripps Highland Drive, and more traffic is around our neighborhood. In recent months, several drivers will fail to stop at the stop sign at the top of Traincrest and we have often had a policernan posted at our corner who has given out tickets to these drivers. During the month of October there is usually a purmpkin patch at the proposed site and parking will offen extend up Scripps Highland Drive. There is an increased noise level at that time as well. The size of this purpwise patch is just a fraction of the size of the proposed Watermark site and should be an indication of the increased traffic and noise that would be in niceases in noise will not be noted, but we as residents of this neighborhood know that there will definitely be an increase in noise will not be noted, but we as residents of this neighborhood know that there will definitely be an increase in noise will not be noted to but we as residents of this neighborhood know that there will definitely be an increase in noise will not be noted.	AA-2	Comments noted. Please see response no. O-6, above.
AA-3	our nomes. I have several concerns, one of which is the increased traffic that will undoubtedly occur around our neighborhood and through our neighborhood. The proposed exit at Scripps Gateway and Scripps Highland will exit into the street that goes into our neighborhood. Although planners may think that this is a solution to route traffic this way, it just serves to congest and cause more traffic through our	AA-3	Comments noted. Please see response no. and G-8, above.
	neighborhood. Through our neighborhood watch we heard of several attempted break-ins over the holidays, and a few more since then.	AA-4	Comments noted. Please see response no. O-3 and O-4. above.
AA-4	An addition concern is if there were a fire, that the Scripps Poway Parkway is our closest exit to the freeway. In the mornings, there is already traffic backed up so that we can often not turn left onto Scripps Poway Parkway because the freeway is backed up to Scripps Highland. During the cedar fires, homes and entire neighborhoods burned in areas that had only one exit or entry point. We will be plugging up our closest exit point to the freeway and out of our neighborhood.	L	
<b>AA</b> -5	Another cause for concern is that the developer wants to increase the size of the buildings by more than 30 feet above the 60 foot limitation. The area is not meant to handle such a size increase. An immediate comparison for traffic can be done to the 15 south exit at Mira Mase Blvd, where on a Friday or Saturday evening, the traffic is backed up from the left turn lane into the exit lane on the freeway. This type of traffic at our exits forseable with such a large shopping complex that does	G-AA	Comments noted. Please see response nos. G-9 and J-5, above.
AA-6	not have adequate room for the patrons to enter and exit. There are a proposed number of 55 shops, while the shopping areas on Scripps Poway Parkway (where the Vons is located and also where the Santana's and Sombrero's is located) have at least 2. 3 vacancles in their shopping areas. Many businesses have come and gone because the community could not support them. A prime example is the Baskin Robbins in the Vons shopping plaza that has changed hands a few times and may ite thesater with a bowling alley down the street from our community. This will only serve to increase traffic not only during rush hour, but during evening and	9-6A	Comments noted. Please see response nos. M-33 and P-2, above.

7 Comments noted. These comments do not address the completeness or adequacy of the EIR. No responses are necessary.	
AA-7	
<ul> <li>weekend hours as well. While the residents realize that something is going into this area, a shopping area will invite traffic during most of the day and night hours. 7 days a week, versus a business building that is open Monday through Friday with business hours. Watemark thinks that the community needs more shopping, we have sufficient shops and stores already in place. There are vacancies in all of these shopping one trans already.</li> <li>Ve do not want to see an increase in crime, noise, traffic, and a devaluation of our property because of this retail center.</li> <li>Deanna Wong</li> </ul>	2
АА	

The Watermark Final Environmental Impact Report

Response to Letters of Comment – Page 83 September 2013

Image: Solution of the solution		<ul> <li>BB-1 Please see response no. X-4 and response no. M-22, above.</li> <li>BB-2 Please refer to the TIA for information regarding project traffic and impacts. The intersection of Scripps Poway Parkway/Pomerado Road was evaluated and the Watermark project is not anticipated to have any impacts at that location. As shown on Figure 3-1, up to 2% of the Watermark project traffic is expected to use Pomerado Road. This would reflect up to 342 ADT on any one segment of Pomerado Road. No significant impacts to Pomerado Road are anticipated.</li> </ul>	
She From Semi Semi Semi Semi Semi Semi Semi Semi	Shearer-Nguyen, Elizabeth Sinearer-Nguyen, Elizabeth From: Elizabeth Sent: Thuraday, Wood (CWood@taxcomp com) Sent: Thuraday, Model (CWood@taxcomp com) Sent: Thuraday, Model (CWood@taxcomp com) Sent: Thuraday, Model (CWood@taxcomp com) Sent: Thuraday, Non-Sent Section CC: Presiden((ScrippeR-Rent), or Com) Presiden((ScrippeR-Rent), or Com) Subject: Work and the complex have a direct link to the bike path that is adjacent to the property along interstate I would deally like to see this complex have a direct link to the bike path that is adjacent to the property along interstate I would deally like to see this complex have a direct link to the bike path that is adjacent to the property along interstate I would deally like to see this complex have a direct link to the bike path that is adjacent to the property along interstate	1. The number of any space must need with a base house interaction of the transportation of must reason and the space and the space of the transportation of the space of the transportation of the space of the transportation of the space of the transport of the space o	

## LIST OF ACRONYMS AND ABBREVIATIONS

AB	Assembly Bill
ac	acre
ADD	Assistant Deputy Director
ADT	Average Daily Traffic
AEOZ	Airport Environs Overlay Zone
AF	acre-feet
AFY	acre-feet per year
AHM	Acutely Hazardous Materials
AIA	Airport Influence Area
ALS	advanced life support
ALUC	Airport Land Use Commission
ALUC Plan/ALCUP	Airport Land Use Compatibility Plan
AM/a.m.	morning
AMSL	above mean sea level
APCD	Air Pollution Control District
ARB	Air Resources Board
ASHRAE	American Society of Heating, Refrigerating, and Air-Conditioning
	Engineers
BAT	Best Available Technology Economically Achievable
BAU	business as usual
ВСТ	Best Conventional Pollutant Control Technology
BEIGIS	Biogenic Emissions Inventory Geographic Information System
BI	Building Inspector
BMP(s)	Best Management Practice(s)
СА	California
CAA	Federal Clean Air Act
CAAQS	California Ambient Air Quality Standards
CAC	California Administrative Code
CAFE	Federal Corporate Average Fuel Economy standard
CalEEMod	California Emission Estimator Model
CalEPA	California EPA
Caltrans	California Department of Transportation
CAPCOA	California Air Pollution Control Officers Association
CBC	California Building Code
CCAP	California Climate Action Plan
CCR	California Code of Regulations
CEC	California Energy Commission
CEFS	California Emission Forecasting System
CEFS	criteria pollutant forecast system
CEIDARS	California Emission Inventory Development and Reporting System
CEQA	California Environmental Quality Act

CFC	chlorofluorocarbons
CFS	cubic feet per second
CGS	California Geologic Survey
CH <sub>4</sub>	methane
Class A Building	Office building classification representing the highest quality building in their market. They are generally the best looking buildings with the best construction, and possess high quality building infrastructure. Class A buildings also are well-located, have good access, and are professionally managed. As a result of this, they attract the highest quality tenants and also command the highest rents.
CMP	Congestion Management Plan
CNEL	community noise equivalent level
CO	carbon monoxide
CO <sub>2</sub>	carbon dioxide
CO <sub>2</sub> e	$CO_2$ equivalent
CR-2-1	City of San Diego Commercial – Regional zone
CSVR	Consultant Site Visit Record
CUP	Conditional Use Permit
001	
dB	decibel
dB(A)	A-weighted decibel
DEH	County Department of Environmental Health
0	degrees as in degrees Fabrenheit
DIE	Development Impact Fee
DIP	City of San Diago Development Services Department
030	City of San Diego Development Services Department
EAS	City of San Diego Environmental Analysis Section
EDU	equivalent dwelling unit
EIR	Environmental Impact Report
EPA	Environmental Protection Agency
ESD	Environmental Services Department
et sea	and the following
et seq.	and the following
FAA	Federal Aviation Administration
FBA	Facilities Benefit Assessment
FEMA	Federal Emergency Management Agency
FHWA	Federal Highway Administration
ft.	feet
FSC	Forest Stewardship Council
100	
g/bhp-hr	grams of particulate matter per brake horsepower hour
ĞCĊ	global climate change
GCP	General Construction Permit
GHG	greenhouse gas
g/l	gram per liter
GWP	global warming potential

HAPs	Hazardous Air Pollutants
HCFC	hydrochlorofluorocarbons
HCM	Highway Capacity Manual
HFC	hydrofluorocarbon
HFE	hydrofluorinated ethers
HMMD	Hazardous Materials Management Division
H <sub>2</sub> S	hydrogen sulfide
H&SC	California Health and Safety Code
HUD	Federal Department of Housing and Urban Development
HVAC	heating, ventilation, and air conditioning
HVAC&R	heating, ventilation, air conditioning, and refrigerating
I-	Interstate, as in I-15
IAQ	indoor air quality
IEPR	Integrated Energy Policy Report
Inc.	incorporated
IPCC	United Nations Intergovernmental Panel on Climate Change
IPM	Integrated Pest Management
IP-2-1	City of San Diego Industrial Park zone
ISO	California Independent System Operator
IWRP	Integrated Water Resources Plan
К	Kindergarten
kg	kilogram
"kit of parts"	Term used to describe elements that are common amongst the
	buildings of a multiple building project that create a unified
	architectural character within the project. Included in these elements
	can be various types of building massing and articulation, forms and
	materials of entries, windows, awnings and other architectural
	treatments. The use of similar, but not identical, elements on
	multiple buildings creates a unified character within the project.
kV	kilovolt
kWh	kilowatt hour
lb	pound
LCFS	Low Carbon Fuel Standard
LDC	City of San Diego Land Development Code
LDR	Land Development Review
LEED	Leadership in Energy and Environmental Design
Leq	equivalent continuous sound level
LID	Low Impact Development
LOS	level of service
MCAS Miramar	Marine Corps Air Station Miramar
mgd	million gallons per day

$\mu g/m^3$	micrograms per cubic meter
$mg/m^3$	milligrams per cubic meter
MHPA	Multi Habitat Planning Area
M-IP	City of San Diego Manufacturing - Industrial Park zone
MMC	Mitigation Monitoring Coordination
MMT	million metric tons
MMRP	Mitigation Monitoring and Reporting Program
mph	miles per hour
MSCP	Multiple Species Conservation Program
MT	metric tons
MMT	million metric tons
MW	megawatt
MWh	megawatt hour
MWD	Metropolitan Water District of Southern California
MWWD	Metropolitan Wastewater Department
NAAOS	National Ambient Air Quality Standards
NB/nb	northbound
NFa	nitrogen trifluoride
NOC	Notice of Completion
NOI	Notice of Intent
NOP	Notice of Preparation
No	number
NO	nitrogen oxide
NOv	avides of pitrogen
NOA	nitrogen dioxide
NDDES	National Pollution Discharge Elimination System
NI DES	national i oliution Discharge Emmitation System
$N_2O$	Introds Oxide
0.	ozone
OCA	off-site consequences analysis
OHP	California Office of Historic Preservation
OPR	The Governor's Office of Planning and Research
01 K	The Governor's Office of Flaining and Research
Pb	lead
PDFs	Project Design Features
PDP	Planned Development Permit
PFC	perfluorocarbon
PFFP	Public Facilities Financing Program
PID	Planned Industrial Development
PM/n m	afternoon
PM.	particulate matter less than 2.5 microns in diameter
PM	particulate matter of 10 microns in diameter or smaller
	parts per million
PRC	Public Resources Code
DVC	nonucine coloride
	poryvinyr unonuc
RAQS	Regional Air Quality Strategy
----------	---
RCP	reinforced concrete pipe
RE	Resident Engineer
RMPP	Risk Management and Prevention Plan
ROG	Reactive Organic Gas
RPS	California's Renewable Portfolio Standard
RSA	Regionally Significant Arterial
RTP	Regional Transportation Plan
RUWMP	Regional Urban Water Management Plan
RWQCB	Regional Water Quality Control Board
SANDAG	San Diego Association of Governments
SB	Senate Bill
SB/sb	southbound
SCAG	Southern California Association of Governments
SCAQMD	South Coast Air Quality Management District
SCH	State Clearinghouse
SCS	Sustainable Communities Strategy
SDAB	San Diego Air Basin
SDAPCD	San Diego Air Pollution Control District
SDCGHGI	San Diego County Greenhouse Gas Inventory
SDCRAA	San Diego County Regional Airport Authority
SDCWA	San Diego County Water Authority
SDFD	San Diego Fire-Rescue Department
SDG&E	San Diego Gas and Electric
SDPD	San Diego Police Department
sec.	second(s)
$SF_6$	sulfur hexafluoride
SIP	State Implementation Plan
SMACNA	Sheet Metal and Air Conditioning National Contractors Association
SOx	sulfur monoxide
$SO_2$	sulfur dioxide
SR	State Route, as in SR-76
SRRE	Source Reduction and Recycling Element
SWRCB	State Water Resources Control Board
SWPPP	Storm Water Pollution Prevention Plan
TAC(s)	Toxic Air Contaminant(s)
Tc	time of concentration
TIA	Traffic Impact Analysis
TLV-STEL	Thresholds Limit Value – Short Term Exposure Limit
TLV-TWA	Threshold Limit Value – Time Weighted Average
TMDL	Total Maximum Daily Load
TNM	Traffic Noise Model

UBC	Uniform Building Code
UFC	Uniform Fire Code
UNFCCC	United Nations Framework Convention on Climate Change
U.S./US	United States
USAI	Urban Systems Associates, Inc.
UWMP	Urban Water Management Plan
v/c	vehicle to capacity ratio
VMT	vehicle miles traveled
VOC	Volatile Organic Compounds
VTM	Vesting Tentative Map
WARM	Waste Reduction Model
WMP	Waste Management Plan
WSA	Water Supply Assessment
WQTR	Water Quality Technical Report

# **EXECUTIVE SUMMARY**

This Environmental Impact Report (EIR) has been prepared for the Watermark project, a private development project located in the Miramar Ranch North Community Plan area. This document analyzes the potential environmental effects associated with implementation of the project (including direct and indirect impacts, secondary impacts, and cumulative effects). Prepared under the direction of the City of San Diego's Environmental Analysis Section, this EIR reflects the independent judgment of the City of San Diego.

# PURPOSE AND SCOPE OF THE EIR

This EIR has been prepared in accordance with, and complies with, all criteria, standards, and procedures of the California Environmental Quality Act (CEQA) of 1970 as amended (PRC 21000 et seq.), State CEQA Guidelines (CAC 15000 et seq.), and City of San Diego's EIR Preparation Guidelines. Per Section 21067 of CEQA and Sections 15367 and 15050 through 15053 of the State CEQA Guidelines, the City of San Diego is the *Lead Agency* under whose authority this document has been prepared. As an informational document, this EIR is intended for use by the City of San Diego decision-makers and members of the public in evaluating the potential environmental effects of the proposed Watermark project.

This EIR provides decision-makers, public agencies, and the public in general with detailed information about the potential significant adverse environmental impacts of the proposed Watermark project. By recognizing the environmental impacts of the proposed project, decision-makers will have a better understanding of the physical and environmental changes that would accompany the project should it be approved. The EIR includes recommended mitigation measures which, when implemented, would provide the Lead Agency with ways to substantially lessen or avoid significant effects of the project on the environment, whenever feasible. Alternatives to the proposed project are presented to evaluate alternative development scenarios that can further reduce or avoid significant impacts associated with the project.

The Watermark project proposes a General Plan Amendment to remove the Prime Industrial Land identification from a portion of the project site, an Amendment to the Miramar Ranch North Community Plan to change the current land use designation from Industrial/Business Park to Regional Commercial for a portion of the site, a Rezone for a portion of the site from IP-2-1 to CR-2-1 (residential prohibitedCommercial - Regional) for a portion of the site, Vesting Tentative Map (VTM), Planned Development Permit (PDP), Conditional Use Permit (CUP) to allow a movie theatre use, <u>Vacation of a Public Utility Easement</u>, and a Street Vacation to vacate Scripps Gateway Court. It is intended that this EIR, once certified, serve as the primary environmental document for those actions. According to Section 15162 of the CEQA Guidelines, when an EIR has been certified for a project, no subsequent EIR shall be prepared for that project unless the Lead Agency determines, on the basis of substantial evidence in light of the whole record, one or more of the following:

- (1) Substantial changes are proposed in the project which will require major revisions of the previous EIR due to the involvement of new significant environmental effects or a substantial increase in the severity of previously identified significant effect;
- (2) Substantial changes occur with respect to the circumstances under which the project is undertaken which will require major revisions of the previous EIR due to the involvement of

new significant environmental effects or a substantial increase in the severity of previously identified significant effects; or

- (3) New information of substantial importance, which was not known and could not have been known with the exercise of reasonable diligence at the time the previous EIR was certified as complete, shows any of the following:
  - (A) The project will have one or more significant effects not discussed in the previous EIR;
  - (B) Significant effects previously examined will be substantially more severe than shown in the previous EIR;
  - (C) Mitigation measures or alternatives previously found not to be feasible would in fact be feasible, and would substantially reduce one or more significant effects of the project, but the project proponents decline to adopt the mitigation measure or alternative; or
  - (D) Mitigation measures or alternative which are considerably different from those analyzed in the previous EIR would substantially reduce one or more significant effects on the environment, but the project proponents decline to adopt the mitigation measure or alternative.

In accordance with CEQA Guidelines Section 15082(a), a Notice of Preparation (NOP), dated September 24, 2010, has been prepared for the project and distributed to all Responsible and Trustee Agencies, as well as other agencies and members of the public who may have an interest in the project. The purpose of the NOP was to solicit comments on the scope and analysis to be included in the EIR for the proposed Watermark project. A copy of the NOP and letters received during its review are included in Appendix A to this EIR. In addition, comments were also gathered at a public scoping session held for the project on January 13, 2010. A transcript of the public scoping meeting is included in Appendix B.

Based on an initial review of the project and comments received, the City of San Diego determined that the EIR for the proposed project should address the following environmental issues:

- Land Use
- Transportation/Traffic Circulation/Parking
- Visual Effects and Neighborhood Character
- Air Quality
- Global Climate Change
- Energy
- Noise
- Biological Resources

- Historical Resources (Archaeological Resources and Historic Resources)
- Geologic Conditions
- Paleontological Resources
- Hydrology/Water Quality
- Health and Safety
- Public Services and Facilities
- Public Utilities
- Cumulative Impacts

Based on the analysis contained in Section 5.0, *Environmental Analysis*, of this EIR, the proposed project would result in significant impacts to: Transportation/Traffic Circulation/Parking (direct and cumulative), and Air Quality (direct relative to construction). Mitigation measures have been identified which would reduce direct and cumulative impacts to below a level of significance for all significant impacts except: Transportation/Traffic Circulation/Parking (direct and cumulative).

# **PROJECT LOCATION AND SETTING**

The regional and local setting of the project is discussed in Section 2.0, *Environmental Setting*, of this EIR. The proposed Watermark project is located in the Miramar Ranch North community of the City of San Diego, within San Diego County. The Watermark project site is located in the southeast quadrant of I-15 and Scripps Poway Parkway. Situated south of Scripps Poway Parkway, east of I-15, a distance north of Mira Mesa Boulevard, and west of Scripps Highlands Drive, the Watermark project site encompasses approximately 34.39 acres, with 22.42 acres developed as a mix of office and retail (Area A) and 11.97 acres remaining as the MedImpact office complex (Area B). (For a full description of Area A and B, please see Section 3.0, *Project Description*.) Single-family residential development within the Scripps Highlands neighborhood occurs east and south of the project site at elevations above the project site. Steep slopes vegetated in native habitat and preserved through an open space easement separate the Watermark site from the Scripps Highlands residential neighborhood on the south and east. North of the project is a small neighborhood commercial center (with hotels and restaurants) and office buildings are located to the northeast of the project site. (For a detailed discussion of the project setting, please see Section 2.0, *Environmental Setting*.)

# **PROJECT BASELINE**

CEQA Guidelines Section 15125(a) guides the discussion of the environmental setting for the proposed project and advises in the establishment of the project baseline. According to CEQA, "[a]n *EIR must include a description of the physical environmental conditions in the vicinity of the project, as they exist at the time the notice of preparation is published*[...]. This environmental setting will normally constitute the baseline physical conditions by which a lead agency determines whether an impact is significant." The following discussion explains the baseline for the Watermark project.

The proposed Watermark project site encompasses the entirety of the approved 2001 Scripps Gateway project amendment for the MedImpact corporate headquarters (see Section 2-3, *Project History*). Under that approved amendment, the MedImpact project site was divided into six lots. Lots 1 and 2 (referred to as Area B in this EIR) of that previous approval include the existing approximately 155,000 square feet of office space (Lot 1) and the yet-to-be constructed 195,743 square feet (Lot 2) of approved corporate office development for MedImpact development. Lot 1 (4.16 acres) includes the recently completed office building and parking structure. Lot 2 (5.68 acres) is entitled as part of the previous approved MedImpact project to construct an office building, parking structure, and ancillary building. While no construction has occurred on Lot 2, building permits can be issued and construction can proceed in accordance with the existing, vested approvals. When completed, Lots 1 and 2 will result in 350,743 square feet of corporate office space. The remainder of the project site (referred to as Area A in this EIR) encompasses 21.13 acres of previously graded but undeveloped pads and 3.42 acres of open space (Lot A). For purposes of this EIR, the existing development on MedImpact Lot 1, the vested development for MedImpact Lot 2, and the remainder of the graded and undeveloped project site establish the existing conditions and baseline for the project. (See Table ES-1, *Baseline Conditions.*)

	Area	Existing Conditions
Area A	21.13 Acres	Undeveloped/Graded Pads
	3.42 Acres	Open Space Lot A
Area B		
MedImpact Lot 1	4.16 Acres	155,000 square feet corporate office space
MedImpact Lot 2	5.68 Acres	195,743 square feet corporate office space
TOTAL	34.39 Acres	<ul> <li>21.13 acres - Undeveloped/Graded Pads</li> <li>3.42 acres - Open Space Lot A</li> <li>9.84 acres - 350,743 square feet corporate office space</li> </ul>

#### Table ES-1. Baseline Conditions

## **PROJECT HISTORY**

The Watermark project site was a part of the larger Scripps Gateway project site (LDR No. 92-0466). Approved in July 1998, the Scripps Gateway project resulted in the subdivision of the original 242.1-acre property and zoning the property for residential, commercial retail, and industrial park uses through the approval of a General Plan/Community Plan Amendment, Tentative Map, Planned Development Permits, and associated actions. Consistent with the original approvals, residential development has occurred east and south of the Watermark site, and retail commercial and office uses have occurred to the north.

The Watermark portion of the Scripps Gateway project is identified as the location of the industrial park uses, originally approved as a Planned Industrial Development (PID) permit, and was zoned M-IP (now the IP-2-1 zone) as part of the original approvals. A Final EIR for the Scripps Gateway project (dated July 16, 1998) was certified for the existing approvals/previous project (LDR No. 92-0466; SCH No. 92101036).

The PID portion of the Scripps Gateway project (i.e., the entire Watermark project site) was subsequently amended in 2001, granting approval for MedImpact to construct its corporate campus to be comprised of seven buildings for use as office, employee training, a cafeteria, exercise facility, and childcare facility (for employee use only) for a total of 658,456 square feet. Public improvements and mass grading was completed at the MedImpact site in 2002. In early 2008, construction began on previously approved Lot 1 of the MedImpact site. Current project approvals included an approved CUP, PID Permit, PDP, and Extension of Time (CUP/PID No. 99-1027; CUP No. 174323/PDP No. 174234 Extension of Time).

## **PROJECT DESCRIPTION**

To implement the Watermark project, the project applicant is requesting approval of an Amendment to the Miramar Ranch North Community Plan and associated General Plan Amendment to change the land use designation from Industrial/Business Park to Regional Commercial, a General Plan Amendment to remove the Prime Industrial Lands identification from a portion of the project site, a Rezone for a portion of the project site from IP-2-1 (Industrial-Park) to CR-2-1 (Commercial-Regional), a Vesting Tentative Map, a PDP with Design Guidelines, a Street Vacation for Scripps Gateway Court, and a CUP for a movie theater. The elements of these various project actions are described in detail in Section 3.0, *Project Description*, of this EIR.

Development is occurring on Lots 1 and 2 of the project site in accordance with existing project approvals for the MedImpact development. Existing project approvals include an approved CUP, PID Permit, PDP, and Extension of Time (CUP/PID No. 99-1027; and CUP No. 174323/PDP No. 174234

Extension of Time). An *Addendum to an Environmental Impact Report* No. 92-0466 was prepared for PID No. 99-1027. The existing approvals allow for the construction of two Class A office buildings, totaling 350,743 square feet, and four additional buildings as the new corporate headquarters for MedImpact Healthcare Systems, Inc. The first of the two buildings (approximately 155,000 square feet) and parking structure have been constructed on the site (Area B). Total development approved for MedImpact is 658,456 square feet. The proposed project includes changes to the approved MedImpact PID, which involves constructing a restaurant in the northeast corner of MedImpact Lot 1 and shared use of parking garages on MedImpact Lots 1 and 2.

Table ES-2, *Proposed Project Development Intensity*, shows the proposed development for the Watermark project, including the existing approvals in effect on the site. In order to allow flexibility in the mix of regional commercial office and/or retail uses in a manner that is reflective of market conditions for employment and retail serving uses, the Traffic Impact Analysis (TIA) is based on a "target development intensity." It is the target development intensity that forms the basis of analysis in this EIR. Depending on the needs of the marketplace at the time development occurs, other mixes of office and retail development remains consistent with the TIA for both total traffic generated and the amount of peak-hour directional trips and that the development complies with the Watermark Architectural Design Guidelines. Therefore, the following table includes the target development intensity, as well as the minimum and maximum development intensity range, that could be developed subject to the limitations of the Traffic Impact Analysis.

	DEVELOPMENT INTENSITY		
USE		PROPOSED 1	
	(Lots 1 and 2)	Development Intensity Range	Target Development Intensity
Commercial Office	350,743 sq. ft. <sup>2</sup>	400,000 - 658,456 sq. ft.	502,112 sq. ft.
Commercial Retail		0 – 500,000 sq. ft.	316,000 sq. ft.
Entertainment (Theater)		0 – 45,000 sq. ft.	43,917 sq. ft.
Hotel (130 rooms)		0 – 100,000 sq. ft.	90,540 sq. ft.
		τοται	953,566 sq. ft.

 Table ES-2.
 Proposed Project Development Intensity

<sup>1</sup> Includes approved project of 350,743 square feet.

<sup>2</sup> Constructed on Lot 1 – 155,000 square feet.

The proposed PDP includes the development square footage for the Watermark project and would supplant the existing vested approvals in effect on the project site (see discussion above). For the purposes of the EIR, the approved on-going development approved for Area B will be considered as part of the existing conditions, except where noted.

## SUMMARY OF ENVIRONMENTAL IMPACTS AND MITIGATION

Section 5.0 of this EIR presents the *Environmental Analysis* of the proposed project. Based on the analysis contained in Section 5.0 of this EIR, the proposed Watermark project would result in significant impacts to: Transportation/Traffic Circulation/Parking (direct and cumulative), and Air Quality (direct relative to construction). Mitigation measures have been identified which would reduce significant direct and cumulative impacts to below a level of significance for all impacts except: Transportation/Traffic Circulation/Parking (direct and cumulative). The alternatives identified in this analysis are intended to further reduce or avoid significant environmental impacts associated with the proposed project.

Table ES-3, *Summary of Environmental Impacts and Mitigation Measures*, summarizes the potential environmental impacts of the Watermark project by issue area, as analyzed in Section 5.0, *Environmental Analysis*, of this EIR. The table also provides a summary of the mitigation measures proposed to avoid or reduce significant adverse impacts. The significance of environmental impacts after implementation of the recommended mitigation measures is provided in the last column of Table ES-1. Responsibilities for monitoring compliance with each mitigation measure are provided in Section 11.0, *Mitigation Monitoring and Reporting Program*, of this EIR.

Environmental Impacts	Mitigation Measures		Level of Significance After Mitigation
Transportation/Traffic Circulation/Parking The project would result in significant direct and cumulative impacts to street segments, and intersections. Section 5.2, <i>Transportation/Traffic</i> <i>Circulation/Parking</i> , provides a summary of the project's impacts before and after mitigation to roadways segments, arterials, intersections, metered freeway on-ramps, and freeway segments from project start through Horizon Year.	MM 5.2-1	Prior to issuance of the first construction permit, owner/permittee shall assure by permit and bond the reconfiguration to shift the westbound through lanes on Scripps Poway Parkway to the north and provide additional queuing length for westbound traffic on Scripps Poway Parkway to the interchange. The "back-to-back" left turn lanes will be eliminated and additional queuing for traffic turning left from Scripps Poway Parkway to southbound I-15 will be provided. Reduction in the width of raised median on Scripps Poway Parkway east of the interchange will be required. All work to be done to the satisfaction of the City Engineer. Prior to issuance of the first construction permit, owner/permittee shall assure by permit and bond the provision of a triple left-turn at Scripps Poway Parkway and Scripps Highlands Drive intersection by re-striping the northbound leg to take a thru-lane and make it a shared left-thru lane. The pedestrian crossing on the west leg of the intersection will be removed. Additionally, a northbound right-turn overlap will be provided. All work to be done to the satisfaction of the City Engineer.	All intersection impacts would be mitigated to below a level of significance. Impacts to street segments would remain significant and unmitigated. The project would not result in impacts to metered freeway on-ramps and freeway segments.
Air Quality Project construction emissions of PM <sub>10</sub> are considered significant but temporary.	<ul> <li>MM 5.4-1. Standard dust control measures would be employed during construction. These standard dust control measures include the following:</li> <li>Watering active grading sites a minimum of three times daily</li> <li>Apply soil stabilizers to inactive construction sites</li> <li>Replace ground cover in disturbed areas as soon as possible</li> <li>Control dust during equipment loading/unloading (load moist material, ensure at least 12 inches of freeboard in haul trucks</li> <li>Reduce speeds on unpaved roads to 15 mph or less</li> <li>Water unpaved roads a minimum of three</li> </ul>		Mitigated to below a level of significance.

Table ES-3. Summary of Environmental Impacts and Mitigation Measures

Environmental Impacts	Mitigation Measures	Level of Significance After Mitigation
	times daily	
	These dust control measures would reduce the amount of fugitive dust generated during construction to below a level of significance.	

## POTENTIAL AREAS OF CONTROVERSY

Pursuant to CEQA Guidelines Section 15123(b)(2), an EIR shall identify areas of controversy known to the Lead Agency, including issues raised by the agencies and the public, and issues to be resolved, including the choice among alternatives and whether and how to mitigate for significant effects. The NOP for the EIR was distributed on September 24, 2010, for a 30-day public review and comment period. In addition, a Public Scoping Meeting was held on January 13, 2010. Comments received in response to the NOP and at the public scoping session present issues to be address in the EIR. No areas of controversy were raised in the comments received.

Presented in Table ES-4, *Summary of NOP Responses and Scoping Meeting Comments*, is a summary of the comments received as part of the City scoping process. (Please see Appendix A for a copy of the NOP and letters received during its review, and Appendix B for a transcript of the public scoping session.)

Issue Raised	Response
Department of Fish and Game - October 20, 2010	
<ol> <li>This letter requests:</li> <li>the project address direct and/or indirect impacts to biological resources on or adjacent to the project site; and</li> <li>the environmental document describe why the proposed project, irrespective of other alternatives to the project, is consistent with and appropriate in the context of open space preservation and the City's Subarea Plan and Implementing Agreement.</li> </ol>	The biological report prepared for the project is summarized in Section 5.8, <i>Biological Resources</i> , and includes a complete assessment of flora and fauna within and surrounding the project site, a discussion of the project's impacts on biological resources, and mitigation measures to reduce those impacts. Section 5.2, <i>Land Use</i> , address the project's consistency with the City's open space preservation plans.
Department of Toxic Substances Control – October 26, 2010	
This letter identifies the need for the DEIR to address hazardous wastes/substances at the project site and in the surrounding area, for any investigation to be summarized in the document, and for a regulatory agency to oversee investigations, samplings, and/or remedial actions.	Potential project impacts relative to human health, public safety, and hazardous materials are discussed in Section 5.13, <i>Health and Safety</i> , and mitigation measures are identified.
Department of Transportation – October 2, 2010	
This letter requests a traffic study be prepared for the proposed project that analyzes near- and long-term effects to State facilities and cumulative traffic impacts, and that mitigation measures are included. The geographic area of the traffic study is requested to include as a minimum all regionally significant arterial system segments and intersections. Analysis should include all freeway entrance and exit ramps where the proposed project will add a significant number of peak-hour trips that may cause any traffic queues to exceed storage capacities, as well as a focused analysis for project trips assigned to a State highway facility that is experiencing significant delay.	A traffic impact analysis was prepared for the proposed project and is summarized in Section 5.2, <i>Transportation/Traffic</i> <i>Circulation/Parking</i> , of the Draft EIR. The analysis evaluates existing conditions, Near Term, and Horizon Year (2030). Cumulative impacts were also analyzed. Impacts were identified for project area roadways, intersections, and freeway ramp meters. Mitigation measures have been identified.
San Diego County Archaeological Society - October 10, 2010	
This letter acknowledges receipt of the NOP and requests to be included on the distribution list of the DEIR, as well as to receive a copy of the cultural resources technical report.	A copy of the EIR and all cultural reports will be sent to the San Diego County Archaeological Society, as requested.

## Table ES-4. Summary of NOP Comments and Scoping Meeting Comments

Issue Raised	Response	
San Diego Audubon Society – October 24, 2010		
This letter requests:	The EIR analyses GHG emissions to the fullest extent possible in	
<ol> <li>a forward-looking issue question be added to the EIR to address if the project has implemented all reasonable measures to reduce the GHG emissions of our region and corresponding narrative detail;</li> <li>a new issue question to Biological Resources relative to the project's tendency to degrade nearby and downstream habitat areas; and</li> <li>the EIR study urban runoff, identify measures that would delay runoff to a greater extent than natural to offset the impacts of prior developments, and prohibit release of any sediments with the runoff.</li> </ol>	Section 5.5, <i>Global Climate Change</i> . Impacts to Biological Resources are addressed in Section 5.8, and project hydrology, including drainage and runoff, is addressed in Section 5.12, <i>Hydrology and Water Quality</i> .	
Briggs Law Corporation – October 15, 2010		
This letter requests inclusion of the Briggs Law Corporation – Inland Empire Office on the EIR distribution list.	No environmental issues were raised in this letter. Brigs Law Corporation was included on the distribution list, as requested.	

## SUMMARY OF PROJECT ALTERNATIVES

#### Alternatives Considered But Rejected

The *Alternatives* section (Section 10.0) of this EIR includes a discussion of alternatives which were considered early in the project design process but which have been rejected. The Alternative Location alternative was determined to be *Considered but Rejected*.

#### Alternative Location Alternative

The proposed Watermark project is intended to serve as a regional commercial center and provide shopping opportunities that are not provided by local neighborhood and community retail, with Class A office buildings (MedImpact) designed as a corporate campus. According the *Watermark Fiscal Revenue Generation Study* (The London Group, July 2012), the growth in the I-15 corridor has increased the demand for specialty retailers that is being served by the emergence of lifestyle retail centers. The project's unique nature allows for a site design on a much smaller footprint than the traditional regional center that includes multiple anchor tenants. While the project's strategic location takes advantage of the proximity to I-15 to provide employment and attract users from the I-15 corridor communities and the region at large, the site also provides an optimum location between two major regional centers to serve an emerging clientele and is well situated to serve an increased demand in high quality corporate office to reflect shifts in the economic structure of the region.

There are no other sites areas within Miramar Ranch North or adjoining communities appropriately located and of sufficient size that could develop in a manner similar to that proposed by the Watermark project. Additionally, there are no other sites under the applicant's control along the I-15 corridor of sufficient size to allow development of a mixed-use commercial center that meets the project's objectives.

In accordance with CEQA Guidelines Section 15126.6(f)(2), alternative locations for the proposed project would be considered if "any of the significant effects of the project would be avoided or substantially lessened by putting the project in another location. Only locations that would avoid or substantially lessens any of the significant effects of the project would need to be considered for inclusion in the EIR." Moving the Watermark project to an alternative site in the community or other areas of the City would not avoid or substantially lessen the

project's impact and could result in greater environmental effects. The project is proposed for a graded site that is slated for development in the Miramar Ranch North Community Plan. The site has easy access to public streets and freeways. Given traffic congestion in the City and County, traffic generated from any alternative site could have the potential to result in impacts on circulation segments, intersections, and freeways. A similar level of intensity as the proposed project constructed at another site in the City or County would have the same level of impacts relative to air quality and GHG emissions. However, the project site has a potential advantage over other sites from an environmental resources standpoint, as the project site does not possess sensitive biological or important cultural resources. Other sites in the City or County may contain significant sensitive resources; and development on another site could result in impacts to biological resources and impacts to cultural resources, which would not occur at the proposed project site.

For these reasons, there are no other feasible alternative locations for the Watermark project as proposed that would meet the project's objectives. Therefore, the Alternative Location alternative has been considered but rejected as infeasible.

## Alternatives Considered

Alternatives considered for the Watermark project, including a discussion of the "No Project" alternative, are addressed in detail in Section 10.0, *Alternatives*. Relative to the requirement to address a "No Project" alternative, CEQA Guidelines Section 15126.6(e)(2) states that:

(2) The "no project" analysis shall discuss the exisitng ocnditions at the time the notice of preparario is published . . . as well as what would be reasonably expected to occur in the foreseeavle future if the project were not approved based on current plans and consistent with available infrastructure and community services.

## CEQA Guidelines Section 15126.6(e)(3) states that:

- (A) When the project is the revision of an existing land use or regulatory plan, policy or ongoing operation, the "no project" alternative will be the continuation of the existing plan, policy or operation into the future.
- (B) If the project is other than a land use or regulatory plan, for example a development project on identifiable property, the "no project" alternative is the circumstance under which the project does not proceed.

In accordance with CEQA, a true No Project alternative would not include the approved MedImpact entitlements. However, because the existing MedImpact approvals can proceed if the Watermark project is not approved, this EIR addresses the No Project Alternative as development that could occur under the existing approvals, as directed by Section 15126(e)(3) of CEQA.

Alternatives to the Watermark project discussed in this EIR include the "No Project" alternative that is mandated by CEQA and other alternatives that were developed in the course of project planning and environmental review for the proposed project. Specifically, the following project alternatives are addressed in this EIR:

Alternative 1 – No Project/Development Under Existing Approvals

- Alternative 2 Light Industrial Park
- Alternative 3 Prime Industrial Lands
- Alternative 4 Reduced Intensity 17 Percent Reduction in Trips

## Alternative 1 – No Project/Development Under Existing Approvals

Under the No Project/Development Under Existing Approvals Alternative, the proposed project would not proceed. Instead, the project site would develop under existing approvals <u>consistent with the current Community Plan and General Plan and zoning for the project site</u>. This would, which involves development of the 34.39-acre site as permitted under the approved CUP/PID No. 99-1027. The approved CUP/PID No 99-1027 allows development of corporate office structures and ancillary buildings on the project site, including the construction of seven buildings for use as office, employee training, cafeteria, exercise facility, and child daycare facility for a total of 658,456 square feet. This alternative would not require also assumes that no additional grading, as the project site has been graded in accordance with the Vesting Tentative Map approved in concert with would be required to construct the existing entitlements.

When compared to the proposed project, the No Project/Development Under Existing Approvals Alternative would result in less overall traffic volumes but the potential for increased trips to neighborhood services. This alternative would result in less vehicular emissions, resulting in less impacts associated with air quality and GHG emissions. Significant air quality impacts associated with construction would occur, similar to the proposed project. Additionally, this alternative would generate less noise associated with less trips than would occur with the proposed project. This alternative would result in the same level of impacts as the proposed project relative to Land Use, Visual Quality and Neighborhood Character, Energy, Historic Resources, Geologic Conditions, Paleontological Resources, Hydrology/Water Quality, Health and Safety, Public Services and Facilities, and Public Utilities.

The No Project/Development Under Existing Approvals Alternative would meet some of the project objectives. Specifically, this alternative would create a coherent and signature design statement as a community gateway to Miramar Ranch North and would result in a project of high quality design and aesthetics, creating a landmark for the community. This alternative would not provide quasi-public space for community use in the form of a pedestrian plaza as a focal point for the project, which would function as a lively gathering place for visitors, employees, and neighbors; and would not implement transportation (vehicular, bicycle, and pedestrian) improvements that would improve operations of the current roadway and bicycle network and would encourage pedestrian use. Additionally, this alternative would not provide flexibility in the allocation of commercial office and retail development based on market demand and consistent with limitations established by the TIA prepared for the project and would not provide for retail uses currently unavailable in the surrounding market area.

## Alternative 2 – Light Industrial Park

The project includes a proposed Community Plan Amendment to change the land use designation from Industrial to Regional Commercial and an amendment to the General Plan to remove the Prime Industrial Lands identification from the site. While the EIR concludes that the proposed land use changes would not result in significant environmental impacts, the proposed project would not be in strict conformation with the Miramar Ranch North Community Plan and the City's General Plan. Therefore, an alternative has been developed to evaluate development of the project site with light industrial land uses, which would be in compliance with <u>the Community Plan's recommendation for a range of light industrial and mini storage uses at this location.</u>

#### require an amendment to the Community Plan.

The Light Industrial Park Alternative would include the on-going development of the MedImpact facilities, including the constructed facilities on Lot 1 and the approved facilities for Lot 2, but would develop the remainder of the site with a mix of manufacturing, wholesale distribution, warehousing, and recreational vehicle storage area screened by mini-warehousing facilities as identified in the Miramar Ranch North Community Plan. Under this alternative, this mix of light industrial uses/mini storage uses would occur on the approximately 21 developable acres outside Area B (MedImpact Lots 1 and 2). The Light Industrial Park alternative would include one- and two-story buildings that would be available for manufacturing, wholesale distribution, mini storage, and warehouse uses and would include an open and screened recreational vehicle storage area. For this alternative, it is assumed that, similar to the proposed project, architectural design guidelines would be developed to ensure high quality design of structures within the project.

The project site is currently zoned IP-2-1. The IP-2-1 zone allows a range of light industrial uses but does not include moving and storage facilities (such as mini storage) and warehouses. Therefore, this alternative would include a rezone from the IP-2-1 zone to the IL-2-1 zone to accommodate the range of light industrial/mini storage uses as outlined in the Community Plan and described for this alternative. Additionally, because this alternative would include uses that are not considered Prime Industrial Lands uses, like the proposed project, an amendment to the General Plan would be required to remove the Prime Industrial Lands identification from the project site.

Overall traffic volumes would be reduced under this alternative; however, this alternative would not provide the balancing of traffic that occurs with the proposed project. Because this alternative would not provide retail commercial and restaurant uses at the project site, neighborhood trips to those services could potentially occur outside the community. Therefore, this alternative would result in less overall trips and could create increased trips to neighborhood services.

While visual quality impacts would not be regarded as significant under this alternative, the gateway statement anticipated by the Community Plan for this area of Miramar Ranch North would not be provided. The nature of types of light industrial, warehousing and mini storage uses associated with this alternative would contrast with the high quality office facilities developed for MedImpact. This alternative would create mini storage facilities and open areas for storage of recreational vehicles. The nature of these types of light industrial uses could create an inconsistent architectural style with the office facilities developed for MedImpact. Additionally, large areas of open surface parking would occur under this alternative. Because this alternative would include open storage facilities and a greater amount of surface parking, it could be regarded as less compatible with surrounding residential development to the north, east, and south, and the commercial retail development to the west and would not create the gateway statement anticipated in the community plan. Therefore, visual quality and neighborhood character impacts would be greater under this alternative.

Relative to air quality and GHG emissions, this alternative has the potential to result in greater impacts. Air quality associated with operational impacts (vehicle trips) would be reduced under the Light Industrial Park Alternative. Relative to construction, impacts would be similar. Warehousing and manufacturing uses which would occur under this alternative would involve a greater amount of diesel trucks accessing the site and would include diesel emissions above those that would occur with the

project. Additionally, light industrial uses, and in particular research and development uses, use toxic substances and hazardous materials which would not occur with the proposed project.

Noise impacts under this alternative would be greater than the proposed project. The Light Industrial Park Alternative could result in an increase in noise levels in the immediate environment due to an increase in truck traffic (in particular, heavy trucks) accessing the site. While such increases in noise would be at a nuisance level and are not anticipated to be significant, this level of noise would not occur with the proposed project.

Similar to the proposed project, the Light Industrial Park Alternative would also not result in significant impacts on site hydrology or drainage. However, this alternative could result in increased impacts associated with water quality, due to the increase in impervious surfaces associated with open parking areas and the use of hazardous materials in manufacturing. While required adherence to State and County regulations would ensure that significant impacts are avoided. These impacts would not occur under the proposed project.

The potential for health and safety impacts could increase with the Light Industrial Park Alternative. Light industrial uses, particularly manufacturing and research and development, can employ hazardous materials. Accidental spills and/or release of hazardous materials or the generation of toxic fumes would create a health and safety risk not associated with the proposed project.

The Light Industrial Park Alternative would result in less cumulative impacts associated with traffic and vehicular emissions, but would result in greater impacts associated with cumulative air quality and GHG emissions, due to diesel truck emissions and emissions that could be generated by light industrial uses that use toxic substances and hazardous materials. For these reasons, cumulative air quality and GHG impacts would be considered greater under this alternative than the proposed project.

This alternative would require similar mitigation as is required for the proposed project to reduce impacts associated with public services and facilities (solid waste generation) to below a level of significance. The same level of impacts as the proposed project would result under this alternative relative to Energy, Historic Resources, Geologic Conditions, Paleontological Resources, and Public Utilities. None of those issues areas were found to result in significant impacts with the proposed project.

This alternative would not meet most of the project objectives. Specifically, this alternative would not create a coherent and signature design statement at a community gateway to Miramar Ranch North. It would not provide flexibility in the allocation of commercial office and retail development based on market demand and consistent with limitations established by the Traffic Impact Analysis prepared for the project and would not allow for retail uses currently unavailable in the surrounding market area. Although this alternative would implement a high quality design and aesthetics, it does not have the same potential to create a landmark statement for the community as would occur under the proposed project. This alternative would not provide quasi-public space for community use in the form of a pedestrian plaza as a focal point for the project, which would function as a lively gathering place for visitors, employees, and neighbors.

## Alternative 3 – Prime Industrial Lands

The Prime Industrial Lands Alternative would include low-rise buildings that would be available for research and development, manufacturing, warehouse heavy or light industrial, and research and development uses consistent with the General Plan's Prime Industrial Lands identification. Under this alternative, the on-going development of the MedImpact facilities and Lots 1 and 2 (350,743 square feet of office uses) would occur; and 21 acres of light industrial uses (such as warehouse distribution, heavy or light manufacturing, and research and development) would occur on the project site, outside Lots 1 and 2. For this alternative, it is assumed that, similar to the proposed project, architectural design guidelines would be developed to ensure high quality design of structures within the project, and that light industrial buildings would be compatible with the existing MedImpact office building and parking structure.

This alternative would not require a Community Plan Amendment or an amendment to the General Plan, as land uses described under this alternative would be consistent with the land uses allowed in the Community Plan and in Prime Industrial Lands. This alternative would not require a rezone, as all uses described for this alternative would be permitted in the existing zones for the project site. However, the uses that would occur under this alternative could result in greater impacts associated with air quality, greenhouse gas emissions, noise, use of toxic substances, and hazardous materials.

Overall traffic volumes would be reduced under this alternative; however, this alternative would not provide the balancing of traffic that occurs with the proposed project. Instead, the project site would develop solely as an employment center, with traffic entering the site during AM peak hours and leaving the site during PM peak hours. Because this alternative would not provide retail commercial and restaurant uses at the project site, neighborhood trips to those services could potentially occur outside the community. Therefore, this alternative would result in less overall trips and the potential for increased trips to neighborhood services.

While visual quality impacts would not be regarded as significant under this alternative, the gateway statement anticipated by the Community Plan for this area of Miramar Ranch North would not be provided. This alternative would result in an active industrial park, with predominately low-rise structures, open surface parking, and truck bays for distributing products. This alternative would be less compatible with surrounding development and would not create the gateway statement anticipated in the community plan. As such, the visual effect of this alternative would not be as desirable, and neighborhood character compatibility would be reduced.

Relative to air quality and GHG emissions, this alternative has the potential to result in greater impacts. Air quality associated with operational impacts (vehicle trips) would be reduced under the Prime Industrial Lands Alternative. Relative to construction, impacts would be the same or slightly less. Manufacturing uses which would occur under this alternative would involve a greater amount of diesel trucks accessing the site and would include diesel emissions above those that would occur with the project. Additionally, Prime Industrial Lands uses, and in particular research and development uses, use toxic substances and hazardous materials which would not occur with the proposed project. For these reasons, air quality impacts and the project's contribution to global climate change would be considered greater under this alternative than the proposed project

Noise impacts under this alternative would be greater than the proposed project. The Prime Industrial Lands Alternative would result in an increase in noise levels in the immediate environment due to an increase in truck traffic (in particular, heavy trucks) accessing the site. While such increases in noise

would be at a nuisance level and are not anticipated to be significant, this level of noise would not occur with the proposed project.

Similar to the proposed project, the Prime Industrial Lands Alternative would also not result in significant impacts on site hydrology or drainage. However, this alternative could result in increased impacts associated with water quality, due to the increase in impervious surfaces associated with open parking areas and the use of hazardous materials in manufacturing and research and development. While required adherence to State and County regulations would ensure that significant impacts are avoided, these impacts would not occur under the proposed project, and impacts associated with water quality would be increased under this alternative.

The potential for health and safety impacts could increase with the Prime Industrial Lands Alternative. Light industrial uses, particularly manufacturing and research and development, can employ hazardous materials. Accidental spills and/or release of hazardous materials or the generation of toxic fumes would create a health and safety risk not associated with the proposed project.

The Prime Industrial Lands Alternative would result in less cumulative impacts associated with traffic and vehicular emissions, but would result in greater impacts associated with cumulative air quality and GHG emissions, due to diesel truck emissions and emissions that could be generated Prime Industrial Lands uses that use toxic substances and hazardous materials For these reasons, cumulative air quality and GHG impacts would be considered greater under this alternative than the proposed project.

This alternative would require similar mitigation as is required for the proposed project to reduce impacts associated with public services and facilities (solid waste generation) to below a level of significance. The same level of impacts as the proposed project would result under this alternative relative to Energy, Historic Resources, Geologic Conditions, Paleontological Resources, and Public Utilities. None of those issues areas were found to result in significant impacts with the proposed project.

This alternative would not meet most of the project objectives. Specifically, this alternative would not create a coherent and signature design statement at a community gateway to Miramar Ranch North. It would not provide flexibility in the allocation of commercial office and retail development based on market demand and consistent with limitations established by the Traffic Impact Analysis prepared for the project and would not allow for retail uses currently unavailable in the surrounding market area. Although this alternative would implement design guidelines that would ensure high quality design and aesthetics, it does not have the same potential to create a landmark statement for the community as would occur under the proposed project. This alternative would not provide quasi-public space for community use in the form of a pedestrian plaza as a focal point for the project, which would function as a lively gathering place for visitors, employees, and neighbors. This alternative would be required to implement transportation (vehicular, bicycle, and pedestrian) improvements that would improve operations of the current roadway and bicycle network and would encourage pedestrian use.

# Alternative 4 – Reduced Intensity – 17 Percent Reduction in Trips

A reduced intensity alternative is addressed in the EIR to evaluate a project alternative that would attain most of the project's goals but reduce project traffic to a point where there would be no unmitigated traffic impacts. It was determined that a 17 percent reduction in trips would avoid unmitigated traffic impacts; all traffic impacts that would result under this alternative could be mitigated to below a level of significance.

The Reduced Intensity – 17 Percent Reduction in Trips Alternative would include the existing MedImpact facilities on previously approved Lot 1, the future approved development on Lot 2, and development of the remainder of the site in a manner similar to the proposed project but without development of the hotel (approximately 90,540 square feet) and the office building (approximately 132,007 square feet). (See Table 10.1, *Alternative 4 – Reduced Intensity – 17 Percent Reduction in Trips Development Intensity.*) The proposed *Watermark Site Plan* (see Figure 3-5) shows the hotel located in the southwest corner of the project site (Building L). This alternative would eliminate the hotel and provide surface parking where the hotel would have occurred. Under the proposed Site Plan, office uses would occur on floors two through 6 of Building M. Because this alternative would eliminate office uses in Building M, Building M would change from a six-story retail/office building to a two-story retail building. Figure 10-1, *Reduced Intensity – 17 Percent Reduction in Trips Alternative Site Plan*, shows the Site Plan resulting from this alternative. With the elimination of these elements, this alternative would result in approximately 17 percent less total trips than under the proposed project (15,341 with this alternative compared to 18,552 resulting from the proposed project).

For the most part, the Reduced Intensity – 17 Percent Reduction in Trips Alternative would have similar impacts and require the same mitigation as the proposed project. This alternative would result in similar uses as the proposed project, but at a reduced level. An Amendment to the Miramar Ranch North Community Plan would still be required to change the land use designation from Industrial to Retail Commercial/Residential Prohibited. Like the proposed project, the General Plan would be amended to remove the Prime Industrial Lands identification from the project site. Like the proposed project, this alternative would create a coherent and signature design statement at this gateway to the Miramar Ranch North community and would maintain consistency with the architectural style established with the existing MedImpact campus by maintaining complementary architectural vocabulary for the mixed-use project. Traffic would be reduced under this alternative, and there would be a concomitant reduction in emissions resulting in a slight reduction in impacts associated with air quality, GHG emissions, and noise. While the Site Plan under this alternative would be modified to eliminate the hotel and the office component of a retail/office building, views of the site would not be substantially different than the proposed project, and impacts associated with Visual Quality and Neighborhood Character would be the same as the proposed project. Impacts relative to all other environmental issue areas would be the same as the proposed project.

## **Environmentally Superior Alternative**

CEQA requires that the EIR identify the Environmentally Superior Alternative among all of the alternatives considered, including the proposed project. If the No Project Alternative is selected as environmentally superior, then the EIR shall also identify an Environmentally Superior Alternative among the other alternatives.

Through a comparison of potential impacts from each of the proposed alternatives and the proposed project, the **No Project/Development Under Existing Approvals** alternative could be considered environmentally superior because it would result in the least amount of environmental impacts.

When compared to the proposed project, the No Project/Development Under Existing Approvals Alternative would result in less overall traffic volumes, but could potentially create increased trips to neighborhood services located outside the community. The No Project alternative would result in a less vehicular emissions, resulting in less impacts associated with air quality and GHG emissions. Additionally, this alternative would generate less noise associated with the reduction in traffic. Because the No Project alternative would result in less environmental impacts than the proposed project, it would be considered the environmentally superior alternative.

Of the other alternatives evaluated in this EIR, either the Light Industrial Park or Prime Industrial Lands Alternative could be considered environmentally superior to the proposed project. The Light Industrial Park and Prime Industrial Lands Alternatives result in similar impacts. However, neither of these alternatives would accomplish the project's main goals and objectives. When compared to the proposed project, the Light Industrial Park and Prime Industrial Lands Alternatives would result in greater overall impacts to the aesthetics of the Miramar Ranch North community and would not create a statement gateway for the community. These alternatives would also not be in compliance with the Community Plan's recommendation for a mix of land uses in this area and would not provide services and amenities to serve nearby residential neighborhoods. Due to the nature of manufacturing and research and development uses that would occur with either of these alternatives, these alternatives could result in the use of hazardous materials that could create health and safety risks and impact to urban runoff; and greater impacts to Air Quality, Global Climate Change, Noise, Hydrology/Water Quality, and Health and Safety would occur. With regards to Transportation/Traffic Circulation/Parking, both the Light Industrial Park and Prime Industrial Lands Alternatives result in less overall traffic volumes than the proposed project; however, these alternatives would not provide for mixed-use development that could potentally reduce trips for the surrounding residential community. The Light Industrial Park and Prime Industrial Lands Alternatives would result in the same level of impacts as the proposed project relative to Energy, Historic Resources, Geologic Conditions, Paleontological Resources, Public Services and Facilities, and Public Utilities.

The Reduced Intensity – 17 Percent Reduction in Trips Alternative would meet the project objectives but would not reduce traffic impacts to the degree that either the Light Industrial Park or Prime Industrial Lands Alternative would. The Reduced Intensity – 17 Percent Reduction in Trips Alternative would be similar in design and uses as the proposed project but would not provide for the same amount of office space as the proposed project and does not include a hotel, which is included in the proposed project.

# **1.0** INTRODUCTION

# 1.1 PURPOSE AND LEGAL AUTHORITY

This Environmental Impact Report (EIR) is an informational document intended for use by the City of San Diego decision-makers and members of the general public in evaluating the potential environmental effects of the proposed Watermark project. This document has been prepared in accordance with, and complies with, all criteria, standards, and procedures of the California Environmental Quality Act (CEQA) of 1970 as amended [Public Resources Code (PRC) 21000 et seq.], State CEQA Guidelines [California Administrative Code (CAC) 15000 et seq.], and the City of San Diego's EIR Preparation Guidelines. Per Section 21067 of CEQA and Sections 15367 and 15050 through 15053 of the State CEQA Guidelines, the City of San Diego is the Lead Agency under whose authority this document has been prepared.

In accordance with CEQA Guidelines Section 15161 and as determined by the City of San Diego, this document constitutes a "Project EIR" and has been focused "primarily on the changes in the environment that would result from the development project." The Watermark project proposes a mixed-use development comprised of commercial office space, commercial retail space, and a hotel on 34.96 gross acres [with 22.42 acres being rezoned and developed as a mix of office and retail (Area A), and the remaining 11.97 acres retaining the existing zone and being developed as the MedImpact office complex (Area B)]. (For a full description of Area A and Area B, please see Section 3.0, Project Description.) A Conditional Use Permit (CUP) to allow a movie theater use is also included in the project proposal, and Scripps Gateway Court would be vacated, requiring processing of a Street Vacation. The project requires discretionary approvals including: a General Plan Amendment to remove the Prime Industrial Land identification from Area A of the project site, an Amendment to the Miramar Ranch North Community Plan to change the current land use designation from Industrial/Business Park to Regional Commercial/Residential Prohibited, Rezone a portion of the site (Area A) from IP-2-1 to CR-2-1 (residential prohibitedCommercial - Regional), Vesting Tentative Map (VTM), Planned Development Permit (PDP), CUP, and a Street Vacation to vacate Scripps Gateway Court.

This EIR provides decision-makers, public agencies, and the general public with detailed information about the potential significant adverse environmental impacts of the proposed Watermark project. By recognizing the environmental impacts of the proposed project, decisionmakers will have a better understanding of the physical and environmental changes that would accompany approval of the project. The EIR includes recommended mitigation measures which, when implemented, will lessen or avoid project impacts. The development of mitigation measures to lessen or avoid project impacts provides the Lead Agency with ways to substantially lessen or avoid significant effects of the project on the environment, whenever feasible. Alternatives to the proposed project are presented to evaluate feasible alternative development scenarios that can further reduce or avoid any significant impacts associated with the project.

## 1.1.1 Authority and Intended Uses of the EIR

Acting as the Lead Agency, the City of San Diego has determined that the Watermark project has the potential to create significant adverse environmental impacts. The City of San Diego Development Services Department (DSD), Environmental Analysis Section (EAS), reviewed the proposed development and has required that an EIR be prepared as part of the project's environmental review process, in accordance with CEQA.

The analysis and findings in this document reflect the independent conclusions of the City of San Diego. Based on an environmental initial study conducted for the project, comments received at the public scoping session held on January 13, 2010 (see Appendix B), and the comments received in response to the Notice of Preparation (NOP) (see Appendix A), this EIR discusses the potential significant adverse effects of the project on a number of environmental issues. Where environmental impacts have been determined to be potentially significant, this EIR presents mitigation measures directed at reducing those adverse environmental effects and makes a determination relative to the ability of the mitigation measures to reduce impacts to below a level of significance. In the event that potentially significant impacts cannot be mitigated to below a level of significance, the EIR states that project approval would require that the decision-maker adopt Findings and a Statement of Overriding Considerations in accordance with Sections 15091 and 15093 of the CEQA Guidelines.

In addition, feasible alternatives to the proposed project have been developed - including the No Project/Development Under Existing Approvals alternative, the Light Industrial Park alternative, Prime Industrial Lands alternative, Reduced Density – 17 Percent Reduction in Trips alternative. An analysis of the impacts of those project alternatives compared to that of the proposed project provide a basis for consideration by decision-makers.

## 1.1.2 Availability and Review of the Draft EIR

After completion of the Draft EIR, a Notice of Completion (NOC) is published to inform the public and interested and affected agencies of the availability of the Draft EIR for review and comment. In addition, the Draft EIR is distributed directly to affected public agencies and to interested organizations for review and comment.

The EIR and all related technical studies are available for review or can be purchased for the cost of reproduction at the offices of the City of San Diego, Development Services Department, Land Development Review Division, located at 1222 First Avenue, Fifth Floor, San Diego, California 92101. Copies of the Draft EIR are also available at the following public libraries:

San Diego Public Library	Scripps Miramar Ranch Library
Central Library	10301 Scripps Lake Drive
820 E Street	San Diego, California 92131-1026
San Diego, California 92101	_

The EIR can also be reviewed on the City's web site through the following link:

http://clerkdoc.sannet.gov/Website/publicnotice/pubnotceqa.html

Agencies, organizations, and individuals have been invited to comment on the information presented in the Draft EIR during a 45-day public review period. Specifically, comments addressing the scope and adequacy of the environmental analysis have been solicited. Respondents have also been asked to provide or identify additional environmental information and/or other feasible

alternatives that are germane to the project, but which they feel may not have been addressed in the analysis. Following the public review period, responses to the public review comments relevant to the adequacy and completeness of the EIR are prepared and compiled into the Final EIR. The City of San Diego City Council, prior to any final decision on the project, will consider the Final EIR for certification.

# 1.2 SCOPE AND CONTENT OF EIR

## 1.2.1 Scope of EIR

An NOP, dated September 24, 2010, was prepared for the project and distributed to all Responsible and Trustee Agencies, as well as other agencies and members of the public who may have an interest in the project. The purpose of the NOP was to solicit comments on the scope and analysis to be included in the EIR for the proposed Watermark project. A copy of the NOP and letters received during its review are included in Appendix A to this EIR. In addition, comments were also gathered at a public scoping session held for the project on January 13, 2010. A transcript of the public scoping meeting is included in Appendix B.

Based on an initial review of the project and comments received, the City of San Diego determined that the EIR for the proposed project should address the following environmental issues:

- Land Use
- Transportation/Traffic Circulation/Parking
- Visual Effects and Neighborhood Character
- Air Quality
- Global Climate Change
- Energy
- Noise
- Biological Resources

- Historical Resources (Archaeological Resources and Historic Resources)
- Geologic Conditions
- Paleontological Resources
- Hydrology/Water Quality
- Health and Safety
- Public Services and Facilities
- Public Utilities
- Cumulative Impacts

Based on the analysis contained in Section 5.0 of this EIR, the proposed project would result in significant impacts to: Transportation/Traffic Circulation/Parking (direct and cumulative) and Air Quality (direct during construction). Mitigation measures have been identified which would reduce direct, and cumulative impacts to below a level of significance for all significant impacts except: Transportation/Traffic Circulation/Parking (direct and cumulative).

## 1.2.2 Format of EIR

Under each issue area presented above, Section 5.0, *Environmental Analysis*, of this EIR includes a description of the existing conditions relevant to each environmental topic; presents the threshold(s) of significance, based on the City of San Diego's CEQA Significance Determination Thresholds, for the particular issue area under evaluation; identifies an issue statement or issue statements; assesses any impacts associated with implementation of the project; provides a summary of the significance of any project impacts; and presents recommended mitigation measures and mitigation monitoring and reporting, as appropriate, for each significant issue area. *Cumulative Impacts* are presented under a

separate discussion section (Section 6.0) based on issues that were found to be potentially cumulatively significant. Section 7.0, *Effects Not Found to be Significant*, presents a brief discussion of the environmental effects of the project that were evaluated as part of the Initial Study process and were found not to be potentially significant. The EIR also includes mandatory CEQA discussion areas (Sections 8.0 and 9.0), which present a discussion of *Growth Inducement* and *Significant Irreversible Environmental Changes*, respectively, as well as a discussion of project *Alternatives* (Section 10.0) which could avoid or reduce potentially significant environmental impacts associated with implementation of the project. Based on this general format, the following presents an outline of the various sections of the EIR for the Watermark project:

- **Executive Summary.** An overview of the EIR, a description of the proposed project, and a summary of impacts and mitigation measures are provided in this section. Areas of controversy, as well as any issues to be resolved, are also presented.
- Section 1.0: Introduction. The purpose of the EIR and a discussion of the public review process are provided in this section. This section also includes the scope and format of the EIR.
- Section 2.0: Environmental Setting. This section provides a description of the project location and the environment of the project site, as well as the vicinity of the project site, as it exists before implementation of the proposed project. The existing environmental setting and conditions as presented in Section 2.0 form the baseline upon which the analysis of potential environmental impacts associated with the project is based. A summary of the project's relationship to the City's General Plan and the Miramar Ranch North Community Plan and existing zoning is also included as part of the Environmental Setting. This section also provides a general discussion of public services and facilities serving the project area.
- Section 3.0: Project Description. This section details the physical and operational characteristics of the project.
- Section 4.0: History of Project Changes. This section chronicles any physical changes that have been made to the project in response to environmental concerns raised during the City's review of the project.
- Section 5.0: Environmental Analysis. The existing environmental setting, potential environmental impacts, and recommended mitigation measures are discussed in this section. Unavoidable significant adverse impacts that remain after mitigation, if any, are also identified in this section.
- Section 6.0: Cumulative Effects. This section describes a list of past, present, and reasonably anticipated future projects in the surrounding area, which, in concert with build-out of the proposed project, may potentially contribute to significant cumulative impacts in the area. The impacts of these related projects in conjunction with the proposed project are analyzed in this section.
- Section 7.0: Effects Not Found to be Significant. This section identifies the issues where potential impacts were considered to be less than significant during the Initial Study process and

describes the reasons why these possible significant environmental effects were deemed not to be significant. For the Watermark project, four environmental issue areas – *Agricultural and Forestry Resources, Mineral Resources, Recreation,* and *Population and Housing* – were determined during the Initial Study not to be potentially significant and, therefore, are not analyzed in Section 5.0 of this EIR. A brief discussion of those environmental issues and why each was determined not to be potentially significant is presented in this section.

- Section 8.0: Growth Inducement. This section discusses the project's potential to foster economic or population growth in the adjacent areas or in the City, either directly or indirectly.
- Section 9.0: Significant Irreversible Environmental Changes. This section describes potentially significant irreversible environmental changes that may be expected with the development of the proposed project.
- Section 10.0: Alternatives. Projects or development scenarios, which may occur on the site and meet most of the project's objectives, were developed as alternatives to the proposed project and are described in this section. Alternative sites where the proposed project may be feasibly constructed are also discussed. Specifically, the *Alternatives* section of this EIR addresses the following project alternatives:

#### Alternatives Considered but Rejected:

Alternative Location for the Project

#### Alternatives Considered:

- No Project/Development Under Existing Approvals Alternative
- Industrial Park Alternative
- Prime Industrial Lands Alternative
- Reduced Intensity 17 Percent Reduction in Trips Alternative
- Section 11.0: Mitigation Monitoring and Reporting Program. This section documents the various mitigation measures required as part of the project.
- Section 12.0: References. A list of the reference materials consulted in the course of the EIR's preparation is included in this section.
- Section 13.0: Individuals and Agencies Consulted. Agencies and individuals contacted during preparation of the EIR are identified in this section.
- Section 14.0: Certification Page. Persons and agencies responsible for the preparation of the EIR are identified in this section.

The Technical Appendices are printed under separate cover as an accompaniment to this EIR. The appendices contain the various supporting documents used in preparing the EIR, including:

- Appendix A Notice of Preparation and Comment Letters The Notice of Preparation and Comment Letters pertain to the entire Watermark site (*Areas A and B*).
- Appendix B Public Scoping Meeting Comments and Transcript The Public Scoping Meeting Comments and Transcripts pertain to the entire Watermark site (*Areas A and B*).
- Appendix C Traffic Impact Analysis for Watermark

The Traffic Impact Analysis for Watermark evaluates traffic from Area A. This study includes the MedImpact traffic (Area B) in the analysis of other projects. As a result, the Traffic Impact Analysis evaluates traffic impacts of the entire project site, *Areas A and B*.

Appendix D - Air Quality Technical Report

The Air Quality Technical Report analyzes the construction impacts of Area A and the operational impacts of Area A and Area B (with Area B being analyzed as an existing condition). As a result, this report evaluates the entire project site, *Areas A and B*.

Appendix E – Greenhouse Gas Evaluation

By nature, greenhouse gas and global climate change evaluations are a cumulative study, taking into account the entirety of the immediately surrounding area. As such, the Greenhouse Gas Evaluation analyzes the impacts of the entire project site, *Areas* A and B.

Appendix F – Noise Study

The Noise Study determines the noise impacts associated with daily operations of the mixed-use project proposed for Area A. Noise impacts associated with the MedImpact facilities (Area B) have already been fully analyzed in the Scripps Gateway EIR. As a result, this study evaluates noise impacts for *Area A* only.

Appendix G – Watermark Biological Resources Report

The Biological Resources Report address the entire project site, Areas A and B.

- Appendix H Drainage Study The Drainage Study is an update to the drainage study previously conducted and approved for the Scripps Gateway project. As such, the report analyzes drainage for the entire project site, *Areas A and B*.
- Appendix I Preliminary Geotechnical Investigation Report

Because Lot 1 of Area B is constructed with approved MedImpact facilities and Lot 2 can develop in accordance with existing approvals for MedImpact, the Preliminary Geotechnical Investigation Report provides geotechnical recommendations for *Area A* only.

Appendix J - Sewer Study Amendment

Because Lot 1 is constructed with the MedImpact facilities and sewer facilities are in place for Lot 2, the Sewer Study Amendment presents the preliminary design of onsite sewer facilities for *Area A* only. Appendix K – Preliminary Water Quality Technical Report

Because Lot 1 of Area B is constructed with approved MedImpact facilities and Lot 2 can develop in accordance with existing approvals for MedImpact, the Preliminary Water Quality Technical Report addresses water quality impacts relative to *Area A* only.

- Appendix L Water Supply Assessment Report The Water Supply Assessment Report pertains to the entire Watermark site (*Areas A and B*).
- Appendix M Letters/Responses to Service Providers The Letters/Responses to Service Providers pertain to the entire Watermark site (Areas A and B).
- Appendix N Federal Aviation Regulations Part 77 Letters on Non-Obstruction and ALUCP Consistency Letter

The Federal Aviation Regulations Part 77 Letters on Non-Obstruction and ALUCP Consistency Letter pertain to the entire Watermark site (*Areas A and B*).

- Appendix O Prime Industrial Lands Criteria Analysis The Prime Industrial Lands Criteria Analysis pertains to the entire Watermark site (Areas A and B).
- Appendix P Collocation/Conversion Suitability Analysis The Collocation/Conversion Suitability Analysis pertains to the entire Watermark site (*Areas A and B*).
- Appendix Q Preliminary Waste Management Plan

The Preliminary Waste Management Plan pertains to the entire Watermark site (Areas A and B).

Appendix R – Preliminary Hydromodification Management Study.

The Preliminary Hydromodification Management Study assumes the approvals in place for Lots 1 and 2 (Area B) and the proposed development of the Watermark project in Area A. Therefore, this study pertains to the entire Watermark site (*Areas A and B*).

Appendix S – Final Paleontological Mitigation Report – Scripps Gateway

The Scripps Gateway Final Paleontological Mitigation Report pertains to the entire Watermark project site (*Areas A and B*)..

## 1.2.3 Incorporation by Reference

As permitted by Section 15150 of the CEQA Guidelines, this EIR has referenced several technical studies, analyses, and reports. Information from the documents, which has been incorporated by reference into this EIR, has been briefly summarized; the relationship between the incorporated part of the referenced document and the EIR is described. The documents and other sources, which have been used in the preparation of this EIR, are identified in Section 12.0, *References*. In accordance with Section 15150(b) of the CEQA Guidelines, the location where the public may obtain and review these referenced documents and other sources used in the preparation of the EIR is also identified (see Section 1.1.2).

# **1.3 EVALUATION OF ENVIRONMENTAL EFFECTS**

The environmental analysis contained in this EIR has been developed to adequately address the environmental issues identified as needing further analysis. Additionally, this EIR addresses concerns raised by comments on the NOP and those received at the January 13, 2010, public scoping session, as presented under *Potential Areas of Controversy* in the *Executive Summary*.

The environmental impact analysis seeks to determine the significance of potential impacts and to develop appropriate mitigation for impacts that have been determined to be significant. In order to facilitate the analysis of each issue, a standard format was developed to analyze each issue thoroughly. This format is presented below, with a brief discussion of the information included within each topic.

## 1.3.1 Existing Conditions

This introductory discussion of each issue section describes the existing environmental conditions related to the specific issue being analyzed. In accordance with Section 15125 of the CEQA Guidelines, both the existing local and regional settings are discussed as appropriate and as they exist prior to implementation of the proposed project and during the preparation of this EIR. This section provides the baseline conditions with which environmental changes created by the project are compared and analyzed. The existing environmental conditions section is the baseline setting for documenting the nature and extent of environmental changes or impacts anticipated to result from project implementation.

#### 1.3.2 Impact Analysis

This section presents an evaluation of the impacts that would result from implementation of the proposed project. The analysis is comprised of four subsections described below, specifically: *Threshold(s) of Significance, Impact Analysis, Significance of Impacts, Mitigation Measures, and Significance of Impacts following Implementation of Mitigation Measures* (as necessary).

#### Thresholds of Significance

Pursuant to Section 15064.7 of the CEQA Guidelines, a threshold of significance is an identifiable quantitative, qualitative, or performance level criterion or criteria. Non-compliance with the threshold(s) would normally mean the effect would be determined to be significant, and compliance with the threshold(s) would normally mean the effect would be determined to be less than significant.

The City of San Diego Development Services Department has developed significance thresholds, referred to as *California Environmental Quality Act Significance Determination Thresholds – Development Services Department* (January 2011), which provide the basis for distinguishing between impacts which are determined to be significant (i.e., impact exceeds the threshold of significance) and those which are typically less than significant. This EIR uses the Development Services Department's Thresholds of Significance to determine significance of potential impacts for each issue area evaluated in this document, with the exception of Global Climate Change.

The City is of San Diego is utilizing the California Air Pollution Control Officers Association (CAPCOA) report "CEQA & Climate Change" (CAPCOA 2009) to determine whether a GHG analysis would be required for submitted projects. The CAPCOA report references a 900 metric ton

guideline as a conservative threshold for requiring further analysis and possible mitigation. This emission level is based on the amount of vehicle trips, the typical energy and water use associated with projects, and other factors.

CAPCOA identifies project types that are estimated to emit approximately 900 metric tons of GHG's annually. This 900 metric ton threshold is roughly equivalent to 35,000 square feet of office space, 11,000 square feet of retail, 50 single-family residential units, 70 multi-family residential units, and 6,300 square feet of supermarkets. Because the Watermark project proposes greater than 35,000 square feet of office space and greater than 11,000 square feet of retail space, the proposed project exceeds the CAPCOA threshold, requiring CEQA review of GHG emissions.

#### Impact Analysis

The impact analysis presented in this EIR begins with a specific "issue question" intended to clearly focus the discussion of the specific environmental issue. The analysis then identifies specific project-related direct and indirect, short-term and long-term, and unavoidable impacts associated with implementation of the Watermark project. A discussion of cumulative impacts is presented in a separate section titled *Cumulative Impacts* (Section 6.0).

Section 15126.2 of the CEQA Guidelines requires that an EIR "*identify and focus on the significant* environmental effects of the proposed project." "Effects" and "impacts" have the same meaning under CEQA and are used interchangeably within this EIR. A "significant effect" or "significant impact" on the environment means "a substantial, or potentially substantial, adverse change in any of the physical conditions within the area affected by the project" (Section 15382 of the CEQA Guidelines). With respect to each potential effect, an analysis has been conducted in this EIR to determine if and to what extent:

- The project causes the identified "impact;"
- The impact produces a substantial, or potentially substantial, change in the physical conditions within the area affected by the project (i.e., "significant"); and
- The changed conditions are "adverse."

In accordance with Section 15145 of the CEQA Guidelines, if, after thorough investigation, a Lead Agency finds that a particular impact is too speculative, the agency should so note its conclusion and terminate discussion of the impact. Therefore, impacts found to be speculative in nature are not evaluated in this EIR.

#### Significance of Impacts

The *Significance of Impacts* subsection provides a concise and brief statement as to whether or not a project impact would constitute a significant environmental effect.

#### Mitigation Measures

This section identifies those mitigation measures that are required to reduce potentially significant environmental impacts and indicates whether those measures would reduce impacts to below a level of significance. As applicable, mitigation measures are discussed in the following terms:

• The specific technical requirements and details for all mitigation measures are described.

- The effectiveness of each measure; i.e., the extent to which the magnitude of impact will be reduced is addressed.
- If the proposed mitigation could result in a significant impact, the potential impact is disclosed and mitigation is provided.

#### Significance of Impacts following Implementation of Mitigation Measures

This conclusion statement addresses the level of significance following implementation of any recommended mitigation measures, as applicable.

## 1.4 **RESPONSIBLE AND TRUSTEE AGENCIES**

State law requires that all EIRs be reviewed by trustee and responsible agencies. A Trustee Agency is defined in Section 15386 of the State CEQA Guidelines as "a state agency having jurisdiction by law over natural resources affected by a project that is held in trust for the people of the State of California." Per Section 15381 of the CEQA Guidelines, "the term 'Responsible Agency' includes all public agencies other than the Lead Agency which have discretionary approval power over the project." For the Watermark project, one State agency would be regarded as a Responsible Agency: the California Department of Transportation – District 11 (Caltrans).

## 1.4.1 California Department of Transportation

The proposed project would result in impacts to State freeways under the control of Caltrans. Project mitigation measures may require contributions to freeway improvements and/or access rights for improvements within Caltrans' rights-of-way. The project applicant would be required to coordinate with Caltrans for these improvements.

## 1.4.2 Regional Water Quality Control Board

Pursuant to Section 401 of the Clean Water Act, the local Regional Water Quality Control Board (RWQCB) (Region 9) would be responsible for issuing a waiver or certification for any project actions resulting in the discharge of runoff from the site. Conformance with the Clean Water Act is established through compliance with the requirements of the National Pollution Discharge Elimination System (NPDES) for discharge of storm water runoff associated with construction activity. Compliance also requires conformance with applicable Best Management Practices (BMPs) and development of a Storm Water Pollution Prevention Plan (SWPPP) and monitoring program plan. A *Water Quality Technical Report* (WQTR) has been completed for the project, which addesses BMPs and the SWPPP (See Appendix K of this EIR.) (Water Quality is addressed in Section 5.12, *Hydrology/Water Quality*, of this EIR.)

## 1.4.3 Federal Aviation Administration

The project's proximity to MCAS – Miramar requires notification to the Federal Aviation Administration (FAA) in order to conduct an Obstruction Evaluation/Airport Airspace analysis under Title 14 code of Federal Regulations, Part 77. The project has completed an initial request for the aeronautical study and has received Determination of No Hazard to Air Navigation for the project (see Appendix N). Individual structures would be required to file subsequent notification to the FAA at least 30 days before the earlier of a) the date proposed construction or alteration is to begin, or b) the date the application for a construction permit would be filed. (The project's relationship to MCAS Miramar is addressed in Section 5.1, *Land Use*, of this EIR.)

Additionally, the Watermark project was reviewed for consistency with the MCAS Airport Land Use Compatibility Plan (ALUCP) by the San Diego County Regional Airport Authority Land Use Commission (ALUC). The project site is located within Review Area 2 of the Airport Influence Area (AIA) for the MCAS Miramar ALUCP. Based on its letter dated August 9, 2010, the ALUC staff determined that a determination of consistency with the ALUC is not required pursuant to Policies 2.6.1(a)(2) and 2.6.1(b)(2) of the MCAS Miramar ALUCP. A subsequent e-mail correspondence from MCAS Miramar to City staff dated May 3, 2013, further documents that no review is needed, as MCAS Miramar found no issues with noise, accident potential, or height guidelines in the ALUCP.

# 2.0 ENVIRONMENTAL SETTING

## 2.1 REGIONAL SETTING

This EIR addresses potential environmental impacts associated with the proposed Watermark project, which is located in the Miramar Ranch North community of the City of San Diego, within San Diego County (see Figure 2-1, *Regional Map*). The City of San Diego covers approximately 206,989 acres in the southwestern section of San Diego County, in southern California. The City is located approximately 17 miles north of the United States-Mexico border and is bordered on the north by the City of Del Mar, the City of Poway, and unincorporated San Diego County land. On the east, the City of San Diego is bordered by the cities of Santee, El Cajon, La Mesa, and Lemon Grove, as well as unincorporated County of San Diego land. To the south, San Diego is bordered by the cities of Coronado, Chula Vista, and National City, as well as the United States-Mexico border. The Pacific Ocean is the City of San Diego's western border.

The Miramar Ranch North community is located in the north-central portion of the San Diego Metropolitan area, predominantly within the northeast limits of the City of San Diego. The community is located approximately 16 miles north of downtown San Diego and 13 miles south of the City of Escondido. The City of Poway is located immediately to the north of Miramar Ranch North, and Interstate 15 (I-15) forms the community's western border. Beyond I-15 to the west lie the Rancho Penasquitos and Mira Mesa communities within the City of San Diego. The Scripps Miramar Ranch Community Plan area is adjacent to Miramar Ranch North on the south. As shown in Figure 2-2, *Vicinity Map*, the Watermark project site is located in the northernmost portion of the Miramar Ranch North Community.

# 2.2 PROJECT LOCATION

As shown in Figure 2-3, *Project Location Map*, the Watermark project site is located in the southeast quadrant of I-15 and Scripps Poway Parkway. Situated south of Scripps Poway Parkway, east of I-15, a distance north of Mira Mesa Boulevard, and west of Scripps Highlands Drive, the Watermark project site encompasses approximately 34.39 acres, with 22.42 acres being rezoned and developed as a mix of office and retail (Area A), 11.97 acres remaining as the MedImpact office complex (Area B), and 0.57 acre of off-site improvement area associated with road improvements along Scripps Poway Parkway. (For a full description of Area A and B, please see Section 3.0, *Project Description.*) Single-family residential development within the Scripps Highlands neighborhood occurs east and south of the project site at elevations above the project site. Steep slopes vegetated in native habitat and preserved through an open space easement separate the Watermark site from the Scripps Highlands residential neighborhood on the south and east. North of the project is a small neighborhood commercial center (with hotels and restaurants) and office buildings are located to the northeast of the project site.



Figure 2-1. Regional Map



Figure 2-2. Vicinity Map



Figure 2-3. Project Location Map

Access to the project site is provided off Scripps Ranch Parkway. I-15 freeway ramps occur at Scripps Poway Parkway providing north- and south-bound access to the interstate.

Marine Corps Air Station Miramar (MCAS Miramar) is located approximately 4.5 miles southwest of the project site. The project site is within the MCAS Miramar Airport Influence Area (AIA). (See Section 5.1, *Land Use*, for a discussion of the proposed project's relationship to MCAS Miramar's Air Compatibility Land Use Plan.)

# 2.3 PROJECT HISTORY

The Watermark project site was a part of the larger Scripps Gateway project site (LDR No. 92-0466). Approved in July 1998, the Scripps Gateway project resulted in the subdivision of the original 242.1-acre property and zoning the property for residential, commercial retail, and industrial park uses through the approval of a General Plan/Community Plan Amendment, Tentative Map, Planned Industrial Development Permits, and associated actions. Consistent with the original approvals, residential development has occurred east and south of the Watermark site, and retail commercial and office uses have occurred to the north.

The Watermark portion of the Scripps Gateway project is identified as the location of the industrial park uses, originally approved as a Planned Industrial Development (PID) permit, and was zoned M-IP (now the IP-2-1 zone) as part of the original approvals. A Final EIR for the Scripps Gateway project (dated July 16, 1998) was certified for the existing approvals/previous project (LDR No. 92-0466; SCH No. 92101036).

The PID portion of the Scripps Gateway project (i.e., the entire Watermark project site) was subsequently amended in 2001 (LDR 99-1027), granting approval for MedImpact to construct its corporate campus to be comprised of seven buildings for use as office, employee training, a cafeteria, exercise facility, and childcare facility (for employee use only). Total development approved for MedImpact under the existing approvals is 658,456 square feet. Public improvements and mass grading were completed at the MedImpact site in 2002. In early 2008, construction began on previously approved Lot 1 of the MedImpact site. Current project approvals include an approved CUP, PID Permit, and Extension of Time (CUP/PID No. 99-1027; CUP No. 174323/PDP No. 174234 Extension of Time). The existing approvals in part allow for the construction of two Class A office buildings, totaling 350,743 square feet (Area B), as the new corporate headquarters for MedImpact Healthcare Systems, Inc. The first of the two buildings (approximately 155,000 square feet) and parking structure have been constructed on Lot 1 of the project site and are currently occupied.

The Watermark project proposes development of a portion of the previously approved PID portion of Scripps Gateway with a mix of retail commercial, movie theater, hotel, office buildings, and parking (both surface and structured). A detailed description of the proposed project is provided in Section 3.0, *Project Description*, of this EIR.

# 2.4 EXISTING SITE CONDITIONS

The Watermark project site encompasses approximately 34.96 acres (Areas A and B plus off-site improvement) and has been graded in accordance with existing approvals. Approximately 11.97 acres of the project site (Area B) are being developed as the MedImpact office complex, with the remaining 22.42 acres (Area A) proposed to be rezoned and developed as a mix of office and retail uses. The remaining 0.57-acre encompasses an off-site area associated with proposed improvements along Scripps Poway Parkway. (A detailed discussion of the proposed project and MedImpact office complex is included in Section 3.0, *Project Description.*) Figure 2-4, *Existing Site Conditions*, depicts the current lotting and the graded conditions for the project site.



Figure 2-4. Existing Site Conditions

One MedImpact building and parking structure have been constructed on approximately 4.16 acres in the eastern portion of the site (previously approved Lot 1). The office building is six stories in height, and the parking structure provides four levels of parking. Previously approved Lot 2 (5.68 acres), located immediately southwest of the existing MedImpact office building, has been graded for development of additional facilities to serve MedImpact. Other areas of the site planned for development in accordance with the original project approvals are graded and void of vegetation. Steep slopes vegetated in native habitat rim the project site on the south and west. Landscaped manufactured slopes occur along the west, north, and east perimeters of the site and around the graded pad in the northeast portion of the site. Scripps Poway Parkway forms the project site's northern border, and Scripps Highlands Drive forms the site's eastern border. Scripps Gateway Court extends west off Scripps Highlands Drive to the east, providing access to the existing MedImpact facilities. An open space area exists immediately south of the project site (Lot A).

## 2.4.1 Topography

The Watermark project site is comprised of graded lots with internal perimeter slopes on previously approved Lots 1 and 6. Additionally, previously approved open space Lot A contains perimeter slopes. Elevations of the graded pads range from approximately 500 feet above mean sea level (AMSL) in the northwest portion of the site to approximately 585 AMSL in the southeast portion of the site. Due to site topography, drainage occurs to the northwest, away from open space areas.

## 2.4.2 Biological Resources

As previously stated, the project site has been graded in accordance with existing project approvals, leaving the site as large graded pads essentially void of vegetation. Within Area B, previously approved Lot 1 has been developed as the first phase of the MedImpact campus, and previously approved Lot 2 is awaiting construction of other buildings approved as part of the MedImpact facilities. As such, the proposed project site consists of an existing paved cul-de-sac road (Scripps Gateway Court), graded pads, manufactured slopes and construction trailers, as well as an office building and parking structure existing on previously approved Lot 1. Since the site has been previously graded in accordance with existing approvals, a majority of the on-site conditions consist of non-native habitat, landscaped slopes, disturbed/ruderal areas, and developed lands. Within the southern portion of the property (Lot A), seeded coastal sage scrub slopes and undisturbed coastal sage scrub/chaparral habitat exists. Wildlife use of the property is limited as would be expected from a previously graded parcel. Little diversity, shelter, or food is available for use by wildlife. Species observed are typical of urbanized or ruderal areas and lack the typical diversity observed in native habitats or non-native grasslands. *Biological Resources* are addressed in Section 5.8 of this EIR.

## 2.4.3 Cultural Resources

The Watermark project site has been graded in accordance with the approved Scripps Gateway project. The Final EIR prepared for the Scripps Gateway project identified the potential for two archaeological sites (CA-SDI-10,780 and CA-SDI-13,186) near the Watermark project site boundaries. CA-SDI-10,780 lies under Scripps Poway Parkway and has been fully mitigated. CA-SDI-13,186 was tested as part of the Scripps Gateway project and determined not to be significant under CEQA. Cultural resources are addressed in Section 5.9, *Historical Resources (Archaeological Resources)*, of this EIR.
### 2.4.4 Geologic Conditions

Compacted fill and Santiago Peak Volcanics (metavolcanic-type rock) underlie the project site. According to the *City of San Diego Seismic Safety Study, Geologic Hazards and Faults*, the project site is categorized as *Zone 53: level or sloping terrain, unfavorable geologic structure, and low to moderate risk.* There are no active faults crossing the site. Previous grading of the project site in accordance with existing approvals has alleviated any unfavorable geologic conditions. Based on the geotechnical investigation performed for the proposed project, the site has suitable conditions for the construction and support of the proposed Watermark development. *Geological Conditions* are addressed in Section 5.10 of this EIR.

### 2.4.5 Paleontological Resources

The project site is underlain by compacted fill and Santiago Peak Volcanics composed of metavolcanic-type rock. Neither of these geologic units have the potential for paleontological resources per the Final Paleontological Mitigation Report prepared for the Scripps Gateway EIR (Appendix S of this EIR). No resources were located during the previous mass grading over this site associated with the Scripps Gateway project and none are expected for the proposed Watermark project. Paleontological Resources are addressed in Section 5.11, *Paleontological Resources*, of this EIR.

### 2.4.6 Visual Resources

The Watermark project site is situated on approximately 34.96 acres, with 22.42 acres being rezoned and developed as a mix of office and retail, 11.97 acres entitled as the MedImpact office complex, and 0.57 acre associated with off-site improvements along Scripps Poway Parkway, in the Miramar Ranch North community. The project site has been graded in accordance with previous approvals resulting in the creation of flat development pads. Steep slopes vegetated in native habitat that occur along the southern and southeastern portions of the site are preserved in open space easements (Lot A). Landscaped manufactured slopes occur along the west, north, and east perimeters of the graded pad in the northeast portion of the site.

Development has occurred on Area B of the project site in accordance with existing approvals. Specifically, the MedImpact building and parking structure have been constructed on approximately 4.16 acres in the eastern portion of the site (previously approved Lot 1). This office building is six stories in height, and the parking structure provides four levels of parking. Scripps Gateway Court has been constructed and extends off Scripps Highlands Drive, providing access to the existing MedImpact facilities.

Visual resources are addressed in Section 5.3, Visual Effects/Neighborhood Character, of this EIR.

### 2.5 SURROUNDING LAND USES

The Watermark project site is situated just east of the I-15 freeway and south of Scripps Poway Parkway. To the east and southeast are the Scripps Highlands residential neighborhoods. North of the Watermark project site, across Scripps Poway Parkway, is a retail commercial site (Scripps Gateway), providing hotel, retail, financial, and restaurant uses. Beyond that is an office building site, identified in the Miramar Ranch North Community Plan as the *North Ridge Site*, at an elevation higher than the adjacent commercial center. Figure 2-5, *Surrounding Land Uses*, shows the land uses surrounding the project site.



## Figure 2-5. Surrounding Land Uses



Page 2-9 September 2013

### 2.6 PUBLIC INFRASTRUCTURE AND SERVICES

Public services are those amenities that serve residents on a community-wide basis. These services include fire protection, police protection, emergency medical, libraries, schools, and parks, as well as their maintenance. Future employees of and visitors to the Watermark project may require use of these services.

The following is a general discussion of the public services and facilities which would be required for the Watermark project based on correspondence and telephone conversations with service providers (see Appendix M, Letters/Responses to Service Providers), in addition to information obtained from the City of San Diego General Plan. (See Section 5.14, *Public Services and Facilities*, for an evaluation of the proposed project's possible impacts on public services and facilities. Figure 5.14-1, *Location of Public Services*, shows the location of public services and facilities that serve the project site and surrounding area.) This discussion does not include a description of parks, public schools, or libraries. Such services are residentially-driven. While employees of and visitors to uses within the Watermark could use these services, they would likely use them in the communities in which they reside.

### 2.6.1 Police

Police protection for the Watermark would be provided by the San Diego Police Department. The goals of police service within San Diego are to provide for safe, peaceful, and orderly communities; and to respond to community needs, respect individuals, develop partnerships, manage emergencies, and apprehend criminals with the highest quality of service. Until the 1980s, the City provided its police services citywide, primarily from a single centralized facility. Several in-house and consultant studies were conducted during the 1970s to evaluate the benefits of decentralizing police functions. As a result of these studies, it was determined that several area stations were to be established throughout the City to better serve individual communities. To accomplish this, a twenty-year plan was developed to establish four new area police stations (Southeastern, Western, Eastern, and Northeastern), replace the existing Southern Division station, construct a new Administrative and Technical Center to replace the existing police headquarters, and relocate the Central Division. Developing needs also led to the construction of a Mid-City Division facility and a centralized Traffic Division facility.

To better serve local communities, the San Diego Police Department has established Community Relations Storefront locations throughout the City. The Miramar Ranch North community is served by the Northeastern Division police facility, on beat 233, located at 13396 Salmon River Road. The Northeastern Division provides police services to the communities of San Pasqual, Rancho Bernardo, Carmel Mountain, Rancho Peñasquitos, Sabre Springs, Mira Mesa, Miramar Ranch North, Rancho Encantada, Scripps Ranch, and Miramar.

According to May 17, 2012, correspondence with Police Lieutenant Ken Hubbs of the SDPD (see Appendix M), the Northeastern Division is currently staffed with 96 sworn personnel and one civilian employee. The current patrol strength is 73 uniformed patrol officers. Officers work tenhour shifts. Staffing is comprised of three shifts which operate from 6:00 a.m. to 4:00 p.m. (First Watch), 2:00 p.m. to midnight (Second Watch), and from 9:00 p.m. to 7:00 a.m. (Third Watch). Using the Department's recommended staffing guidelines, Northeastern Division currently deploys a minimum of nine patrol officers on First Watch, 11 patrol officers on Second Watch, and seven

patrol officers on Third Watch. The goal citywide is to maintain 1.45 officers per 1,000 population ratio.

The San Diego Police Department has mutual aid agreements with all other Law Enforcement Agencies in San Diego County. Additionally, in order to best manage emergencies as development and population growth occurs, the City of San Diego has established the average response time guidelines. The calls are prioritized by the phone dispatcher and routed to the radio operator for dispatch to the field units. The priority system is designed as a guide, allowing the phone dispatcher and the radio dispatcher discretion to raise or lower the call priority as necessary based on the information received. Priority E and Priority 1 calls involve serious crimes in progress or those with a potential for injury. Priority 2 calls include vandalism, disturbances, and property crimes. Priority 3 includes calls after a crime has been committed, such as cold burglaries and loud music. Priority 4 include calls regarding parking complaints or lost and found reports.

The project site is located in the City of San Diego within the boundaries of police beat 246. The 2011 average response times for beat 246 are 7.7 minutes for Priority E calls, 15.2 minutes for Priority 1 calls, 21.2 minutes for Priority 2 calls, 44.8 minutes for Priority 3 calls, and 51.7 minutes for Priority 4 calls. The department's response time goals are:

- Priority E Calls (imminent threat to life) within seven minutes.
- Priority 1 Calls (serious crimes in progress) within 14 minutes.
- Priority 2 Calls (less serious crimes with no threat to life) within 27 minutes.
- Priority 3 Calls (minor crimes/requests that are not urgent) within 70 minutes.
- Priority 4 Calls (minor requests for police service) within 70 minutes.

The citywide average response times, for the same period, were 6.3 minutes for Priority calls, 11.1 minutes for Priority 1 calls, 22.8 minutes for Priority 2 calls, 62 minutes for Priority 3 calls, and 67.8 minutes for Priority 4 calls – all within the Department's response time goals. The Department strives to maintain the response time goals as one of various other measures used to assess the level of service to the community.

The department is currently reaching its targeted staffing ratio of 1.45 per 1,000 residents based on 2011 estimated residential population of 1,311,882. The ratio is calculated using the Department's total to take into account the support and investigative positions within the department. This ratio does not include the significant population increase resulting from persons who commute to work from outside the city of San Diego or those visiting. There are no current plans for additional police sub-stations in the immediate project area. Police response times in this community will continue to increase with the build-out of community plans and the increase of traffic generated by new growth. A Crime Prevention through Environmental Design Review (CPTED) is recommended by the police department to address general safety concerns.

### 2.6.2 Fire Safety

The goal of Fire-Rescue service within San Diego is to protect life, property, and the environment by delivering the highest level of emergency and fire-rescue services, hazard prevention, and safety education. The San Diego Fire-Rescue Department is responsible for the preparation, maintenance, and execution of Fire Preparedness and Management Plans and participates in multi-jurisdictional disaster preparedness efforts. In the event of a large wildfire within or threatening City limits, the City's Fire-Rescue Department can be assisted by the California Department of Forestry, Federal Fire Department, or other local fire department jurisdictions.

A policy of San Diego Fire-Rescue is to locate, staff, and equip fire stations to meet established response times. There are two fire stations located near the Miramar Ranch North community in order to facilitate expeditious response times: Station Number 37 located at 10750 Scripps Lake Drive, and Station Number 44 located at 10011 Black Mountain Road.

In order to best serve the community, San Diego Fire-Rescue has established the following guideline. Response time objectives are based on national standards.

- Total response time for deployment and arrival of the first-in engine company for fire suppression incidents should be within four minutes 90 percent of the time.
- Total response time for deployment and arrival of the full first alarm assignment for fire suppression incidents should be within eight minutes 90 percent of the time.
- Total response time for the deployment and arrival of first responder or higher-level capability at emergency medical incidents should be within four minutes 90 percent of the time.
- Total response time for deployment and arrival of a unit with advanced life support (ALS) capability at emergency medical incidents, where this service is provided by the City, should be within eight minutes 90 percent of the time.

According to May 31, 2012, correspondence with Assistant Fire Marshal Lawrence Trame (see Appendix M), to treat medical patients and control small fires, the first-due unit should arrive within 7.5 minutes, 90 percent of the time from the receipt of the 911-call in fire dispatch. This equates to one-minute dispatch time, 1.5 minutes company turnout time, and five minutes drive time in the most populated areas. To confine fires near the room of origin, to stop wildland fires to under three acres when noticed promptly, and to treat up to five medical patients at once, a multiple-unit response of at least 17 personnel should arrive within 10.5 minutes from the time of 911-call receipt in fire dispatch, 90 percent of the time. This equates to one-minute dispatch time, 1.5 minutes company turnout time spacing for multiple units in the most populated areas.

To direct fire station location timing and crew size planning as the community grows, the adopted fire unit deployment performance measures based on population density (per square mile) zones are listed in the table below:

	Structure Fire Urban Area	Structure Fire Rural Area	Structure Fire Remote Area	Wildfires Populated Areas
	>1,000-	1,000 to 500	500 to 50	Permanent
	people/sq.	people/sq.	people/sq.	open space
	mi.	mi.	mi.	areas
1 <sup>st</sup> Due Travel Time	5	12	20	10
Total Reflex Time	7.5	14.5	22.5	12.5
1 <sup>st</sup> Alarm Travel Time	8	16	24	15
1 <sup>st</sup> Alarm Total Reflex	10.5	18.5	26.5	17.5

Where more than one square mile is not populated at similar densities, and/or a contiguous area with different zoning types aggregates into a population "cluster," the following measures guide the determination of response time measures and the need for fire stations:

Area	Aggregate Population	First-Due Unit Travel Time Goal
Metropolitan	> 200,000 people	4 minutes
Urban-Suburban	< 200,000 people	5 minutes
Rural	500 - 1,000 people	12 minutes
Remote	< 500	> 15 minutes

Brush management is considered an integral, key component of an overall Fire Preparedness and Management Plan. For the Watermark project, brush management is addressed in Section 5.13, *Health and Safety.* 

### 2.7 PLANNING CONTEXT

Development projects within the City of San Diego are guided by the City's General Plan. More specifically, however, development proposals are reviewed in accordance with the community plan for the community in which they are located. The project site encompasses approximately 34.96 acres within the Miramar Ranch North Community Plan Area, with 22.42 acres being rezoned and developed as a mix of office and retail, 11.97 acres remaining as the MedImpact office complex, and 0.57 acre associated with improvements along Scripps Poway Parkway. In addition to the General Plan, for the Watermark project, the Miramar Ranch North Community Plan applies. (See Section 5.1, *Land Use*, of this EIR for a detailed discussion of the planning documents and policies affecting development of the project site.)

### 2.7.1 City of San Diego General Plan

The City's General Plan sets forth a comprehensive, long-term plan for development within the City of San Diego. As such, the plan and development guidelines it identifies pertain to the project site. Elements of the General Plan address the following issue areas: Land Use and Community Planning; Mobility; Urban Design; Economic Prosperity; Public Facilities, Services, and Safety; Recreation; Conservation; Noise; and Historic Preservation. The General Plan identifies the project site as Industrial Employment (Figure 2-6, *City of San Diego General Plan Land Use Map*). Land use is addressed in Section 5.1, *Land Use*, of this EIR.

The project site is identified as Prime Industrial Land in the City of San Diego General Plan. The project proposes to remove the Prime Industrial Land identification from a portion of the project site (Area A) and change the land use designation for that area from Industrial to Regional Commercial. Impacts to local and regional Prime Industrial Lands due to proposed land use designation and zoning changes are discussed in Section 5.1.

### 2.7.2 Miramar Ranch North Community Plan

The project site is governed by the Miramar Ranch North Community Plan, which was first adopted by the San Diego City Council in 1980. Several amendments have occurred since its adoption, with the most recent amendment occurring in 1998.

According to the adopted Miramar Ranch North Community Plan, the project site is designated for Industrial/Business Park uses (see Figure 2-7, *Miramar Ranch North Community Plan Land Use Map*). The project proposes an amendment to the Community Plan to change the existing land use designation for a portion of the site to Regional Commercial/Residential Prohibited. Section 3.0, *Project Description*, describes the proposed Community Plan Amendment, and Section 5.1, *Land Use*, addresses the environmental effects that would result from the proposed change in land use.

### 2.8 ZONING

Zoning for the Watermark project site is governed by the City's Land Development Code (LDC). Within the Miramar Ranch North community, the project site is currently zoned IP-2-1 (Industrial-Park). (See Figure 2-8, *Existing Zoning*.) The purpose of the City's IP zones is to provide for high quality science and business park development. The property development standards of this zone are intended to create a campus-like environment characterized by comprehensive site design and substantial landscaping. Restrictions on permitted uses and signs in this zone are provided to minimize commercial influence. The IP-2-1 zone allows a mix of light industrial and office uses.

The project proposes to rezone a portion of the project from the existing IP-2-1 zone to CR-2-1 (Commercial-Regional). Area B (where the MedImpact facilities are located), consisting of 11.97 acres, would remain zoned IP-2-1. The remainder of the project site (Area A), 22.42 acres, would be rezoned to CR-2-1. The purpose of the CR-2-1 zone is to provide for a broad mix of business/professional office, commercial service, retail, wholesale, and limited manufacturing uses. Specifically, the CR-2-1 zone allows regional-serving commercial and limited industrial uses with an auto orientation but no residential uses. *Proposed Zoning* for the project is presented in Section 3.3.2.

### 2.9 MCAS MIRAMAR ALUCP

As shown in Figure 2-9, *MCAS Miramar – Airport Influence Area Map*, the Watermark project area is located within the AIA identified in the Airport Land Use Compatibility Plan (ALUCP) for MCAS Miramar. The basic function of the ALUCP is to promote compatibility between airports and the land uses that surround them to the extent that these areas are not already devoted to incompatible land uses. The ALUCP safeguards the general welfare of the inhabitants within the vicinity of MCAS Miramar and the public in general. (See Section 5.1, *Land Use*, for a discussion of the project site's relationship with the MCAS Miramar ALUCP.)



Figure 2-6. City of San Diego General Plan Land Use Map



The Watermark Final Environmental Impact Report

Page 2-16 September 2013



nue z-o existing zuinig

The Watermark Final Environmental Impact Report





# Figure 2-9. MCAS Miramar – Airport Influence Area Map

The Watermark Final Environmental Impact Report

Page 2-18 September 2013 There are two Review Areas for MCAS Miramar. Review Area 1 consists of locations where noise and/or safety concerns may necessitate limitations on the types of land uses. Specifically, Review Area 1 encompasses locations exposed to noise levels of *CNEL* 60 dB or greater together with all of the safety zones depicted on the associated maps in the ALUCP. Within Review Area 1, *all* types of land use actions are to be submitted to the *ALUC* for review to the extent review is required by law. Review Area 2 consists of locations beyond Review Area 1 but within the airspace protection and/or overflight areas depicted on the associated maps in the ALUCP. Limits on the heights of structures, particularly in areas of high terrain, are the only restrictions on land uses within Review Area 2. The additional function of this area is to define where various mechanisms to alert prospective property owners about the nearby airport are appropriate. Within Review Area 2, only land use actions for which the height of objects is an issue are subject to *ALUC* review.

The project site is within Review Area 2. The project's proximity to MCAS Miramar requires notification to the Federal Aviation Administration (FAA) in order to conduct an Obstruction Evaluation/Airport Airspace analysis under Title 14 code of Federal Regulations, Part 77. The project has completed an initial request for the aeronautical study and has received Determination of No Hazard to Air Navigation for the project (see Appendix N). Individual structures would be required to file subsequent notification to the FAA at least 30 days before the earlier of a) the date proposed construction or alteration is to begin, or b) the date the application for a construction permit would be filed. (The project's relationship to MCAS Miramar is addressed in Section 5.1, *Land Use*, of this EIR.)

The MCAS Miramar ALUCP addresses four types of airport land use compatibility concerns: noise, safety, airspace protection, and overflight. Noise contours have been established for the purpose of evaluating the noise compatibility of land use development in the AIA of MCAS Miramar. The Watermark project site is well outside any impacting noise contours for MCAS Miramar. (See Section 5.7, *Noise*, for a discussion on noise impacts, including those from aircraft activity at MCAS Miramar.) Safety zones for the MCAS Miramar ALUCP have been established for the purpose of evaluating the safety compatibility of land use development in the AIA. The Watermark project site is not located within a safety zone. Airport protection zones have been established to evaluate the airspace compatibility of land use development within the AIA. The Watermark project is not within the conical surface Airspace Protection Compatibility Area. The project site is within an overflight zone. Impacts relative to the overflight zone are discussed in Section 5.1, *Land Use*.

### 2.10 PROJECT BASELINE

CEQA Guidelines Section 15125(a) guides the discussion of the environmental setting for the proposed project and advises in the establishment of the project baseline. According to CEQA, "[a]n EIR must include a description of the physical environmental conditions in the vicinity of the project, as they exist at the time the notice of preparation is published[...]. This environmental setting will normally constitute the baseline physical conditions by which a lead agency determines whether an impact is significant." The following discussion explains the baseline for the Watermark project.

The proposed Watermark project site encompasses the entirety of the approved 2001 Scripps Gateway project amendment for the MedImpact corporate headquarters, including the existing development on Lot 1 and the approved entitlements for Lot 2 (see Section 2-3, *Project History*), and the currently undeveloped portion of the MedImpact project (Lots 3 - 6). Under that approved

amendment, the MedImpact project site was divided into six lots. Lots 1 and 2 (referred to as Area B in this EIR) of that previous approval include the existing approximately 155,000 square feet of office space (Lot 1) and the yet-to-be constructed 195,743 square feet (Lot 2) of approved corporate office development for MedImpact development. Lot 1 (4.16 acres) includes the recently completed office building and parking structure. Lot 2 (5.68 acres) is entitled as part of the previous approved MedImpact project to construct an office building, parking structure, and ancillary building. While no construction has occurred on Lot 2, building permits can be issued and construction can proceed in accordance with the existing, vested approvals. When completed, Lots 1 and 2 will result in 350,743 square feet of corporate office space. The remainder of the project site (referred to as Area A in this EIR) encompasses 21.13 acres of previously graded but undeveloped pads for MedImpact and 3.42 acres of open space (Lot A). For purposes of this EIR, the existing development on MedImpact Lot 1, the vested development for MedImpact Lot 2, and the remainder of the graded and undeveloped project site establish the existing conditions and baseline for the project. (See Table 2-1, *Baseline Conditions.*)

	Area	Existing Conditions
Area A	21.13 Acres	Undeveloped/Graded Pads
	3.42 Acres	Open Space Lot A
Area B		
MedImpact Lot 1	4.16 Acres	155,000 square feet corporate office space
MedImpact Lot 2	5.68 Acres	195,743 square feet corporate office space
Off-Site Area	0.57 acre	Off-Site Improvement Area
TOTAL	34.96 Acres	<ul> <li>21.13 acres - Undeveloped/Graded Pads</li> </ul>
		<ul> <li>3.42 acres - Open Space Lot A</li> </ul>
		9.84 acres - 350,743 square feet corporate office
		space
		0.57 acre – Off-Site Improvement Area

	Table 2-1.	Baseline	Conditions
--	------------	----------	------------

## **3.0 PROJECT DESCRIPTION**

### 3.1 PROJECT BACKGROUND

This EIR analyzes potential environmental effects associated with the proposed Watermark project, located on approximately 34.96 acres in the Miramar Ranch North community, San Diego, California, with 22.42 acres (Area A) being rezoned and developed as a mix of office and retail, 11.97 (Area B) acres remaining as the MedImpact office complex, and 0.57 acre associated with improvements along Scripps Poway Parkway. As described in Section 2.3, *Project History*, the Watermark project site was a part of the larger Scripps Gateway project, approved in 1998. Industrial park land uses were originally approved for the Watermark project site as a PID permit, and the project site was zoned M-IP (now the IP-2-1 zone) as part of the original approvals.

The Watermark project site is the location of previous approvals associated with the larger Scripps Gateway project approved in 1998 (LDR No. 92-0466). The Scripps Gateway project was subsequently amended in 2001, granting approval for MedImpact to construct its corporate campus on the project site (LDR No. 99-1027). Under the approved 2001 amendment, the MedImpact facilities were to be comprised of seven buildings for use as offices, employee training, a cafeteria, exercise facility, and childcare facility (for employee use only) for a total of 658,456 square feet. (See Figure 2-4, *Existing Site Conditions.*) Public improvements and mass grading were completed at the site in 2002. Under the proposed project, two of the seven buildings approved with the MedImpact project have been (previously approved Lot 1) or will be constructed (previously approved Lot 2).

Development has occurred on the project site under the existing entitlements for the construction of the first of two Class A office buildings, totaling 350,743 square feet, as the new corporate headquarters for MedImpact Healthcare Systems, Inc. In 2008, construction for MedImpact began on previously approved Lot 1 of the project site, and the first of the two buildings (approximately 155,000 square feet) and parking structure have been constructed on the project site. The second office building and accessory facilities are approved and will be developed on previously approved Lot 2. The Watermark project encompasses the previously approved Lots 1 and 2 (Area B) and the proposed development of the remainder of the project site (Area A) as a mixed-use commercial retail and office development. The proposed project includes changes to the approved MedImpact PID, which involve constructing a restaurant in the northeast corner of MedImpact Lot 1 and shared use of parking garages on MedImpact Lots 1 and 2.

Figure 2-3, *Project Location Map*, shows development that has occurred on the project site to-date, including the MedImpact building and parking structure that exist on previously approved Lot 1 of the Watermark project site, as well as graded areas for Lot 2 and all of Area A that has not yet developed. Development of previously approved Lot 2 has been approved and can move forward with or without approval of the Watermark project. The Watermark project proposes development on Area A with a mix of retail commercial, movie theater, hotel, office buildings, and parking (both surface and structured), as described in detail below. Approved development of Area B is included within the Watermark project site boundary, except where noted for certain environmental impact analysis. (See Section 1.2.2, *Format of the EIR*, for a discussion of what areas each technical study addresses.)

The MedImpact corporate headquarters will ultimately be comprised of an 88-foot-tall, five-story and 112-foot-tall six-story LEED<sup>®</sup> Gold-rated office buildings with high visibility to the I-15 corridor. An ancillary structure of 26,469 square feet to serve as an employee cafeteria and fitness center will be located between the two office buildings. These first two phases of the project have already received City approval.

To implement the goal to create a coherent and signature design statement at this community gateway to Miramar Ranch North, these Class A office buildings draw from the same palette of colors and materials, including the use of natural stone, to achieve compatibility in the implementation of the office campus. The modern design is achieved through an extensive use of glass with metal and accents. Building articulation employs the subtle use of offsets and curves to provide relief to rectangular building design. The buildings are oriented to provide an offset view from the freeway and the primary drive from Scripps Highland Drive, as well as shield the view to the parking structures that serve each building.

### 3.2 PURPOSE AND OBJECTIVES OF THE PROPOSED PROJECT

CEQA Guidelines require that the Project Description include a statement of the objectives sought by the proposed project. A clearly defined written statement of the objectives would help the Lead Agency develop a reasonable range of alternatives to evaluate in the EIR and would aid decisionmakers in preparing findings and overriding considerations, if necessary. The statement of objectives also needs to include the underlying purpose of the project [CEQA Guidelines Section 15124(b)].

Actions associated with the proposed project include a VTM, a PDP with Design Guidelines, a Street Vacation for Scripps Gateway Court, and a CUP for a movie theater. To implement the Watermark project, the project also rerquires an Amendment to the Miramar Ranch North Community Plan to change the land use designation from Industrial/Business Park to Regional Commercial and associated General Plan Amendment to change the General Plan land use designation from industrial to Regional Commercial, a General Plan Amendment to remove the Prime Industrial Lands identification from Area A of the project site, and a Rezone for a portion of the project site from IP-2-1 (Industrial-Park) to CR-2-1 (Commercial-Regional).

### Project Purpose

The purpose of the Watermark project is to create a viable mix of commercial retail, visitor commercial, office, and entertainment uses that would serve the adjacent residential neighborhoods, the Miramar Ranch North community, and adjacent communities. The project's location and proposed uses would serve to reduce trips to outlying areas for similar retail services while also expanding employment uses proximate to residential development.

### Project Objectives

The project objectives associated with the Watermark project are as follows:

- Create a coherent and signature design statement at a community gateway to Miramar Ranch North.
- Provide flexibility in the allocation of commercial office and retail development based on market demand and consistent with limitations established by the Traffic Impact Analysis prepared for the project.
- Allow for retail uses currently unavailable in the surrounding market area.
- Implement design guidelines that would ensure high quality design and aesthetics, creating a landmark for the community.
- Provide quasi-public space for community use in the form of a pedestrian plaza as a focal point for the project, which would function as a lively gathering place for visitors, employees, and neighbors.
- Implement transportation (vehicular, bicycle, and pedestrian) improvements that would improve operations of the current roadway and bicycle network and would encourage pedestrian use.

### 3.3 **PROJECT CHARACTERISTICS**

To implement the Watermark project, the project applicant is requesting approval of an Amendment to the Miramar Ranch North Community Plan to change the land use designation from Industrial/Business Park to Regional Commercial and associated General Plan Amendment to change the General Plan land use designation from industrial to Regional Commercial, a General Plan Amendment to remove the Prime Industrial Lands identification from Area A of the project site, a Rezone for a portion of the project site from IP-2-1 (Industrial-Park) to CR-2-1 (Commercial-Regional), a Vesting Tentative Map, a PDP with Design Guidelines, <u>Vacation of a Public Utility</u> <u>Easement</u>, a Street Vacation for Scripps Gateway Court, and a CUP for a movie theater. The elements of these various project actions are described below.

In addition to new development, the proposed project includes development that is occurring on the project site in accordance with existing project approvals. Existing project approvals include an approved CUP, PID Permit, PDP, and Extension of Time (CUP/PID No. 99-1027; and CUP No. 174323/PDP No. 174234 Extension of Time). The existing approvals allow for the construction of two Class A office buildings, totaling 350,743 square feet, and four additional buildings as the new corporate headquarters for MedImpact Healthcare Systems, Inc. The first of the two buildings (approximately 155,000 square feet) and parking structure have been constructed on the site.

Table 3-1, *Proposed Project Development Intensity*, shows the proposed development for the Watermark project, including the existing approvals in effect on the site. In order to allow flexibility in the mix of regional commercial office and/or retail uses in a manner that is reflective of market conditions for employment and retail serving uses, the Traffic Impact Analysis is based on a "target development intensity." It is the target development intensity that forms the basis of analysis in this EIR. Depending on the needs of the marketplace at the time development occurs, other mixes of office and retail development could occur and may result in more or less than the target development intensity, provided that the overall development remains consistent with the Traffic Impact Analysis for both total traffic generated and the amount of peak-hour trips and that the

development complies with the Watermark Architectural Design Guidelines. Therefore, the following table includes the target development intensity, as well as the minimum and maximum development intensity range, that could be developed subject to the limitations of the Traffic Impact Analysis.

		DEVELOPMENT INTENSITY	
USE		PROPO	SED 1
	(Lots 1 and 2)	Development Intensity Range	Target Development Intensity
Commercial Office	350,743 sq. ft. <sup>2</sup>	400,000 - 658,456 sq. ft.	502,112 sq. ft.
Commercial Retail		0 – 500,000 sq. ft.	316,000 sq. ft.
Entertainment (Theater)		0 – 45,000 sq. ft.	43,917 sq. ft.
Hotel (130 rooms)		0 – 100,000 sq. ft.	90,540 sq. ft.
		TOTAL	953.566 sq. ft.

Table 3-1. Proposed Project Development Intensity

<sup>1</sup> Includes approved project of 350,743 square feet.

<sup>2</sup> Constructed Lot 1 – 155,000 square feet.

The proposed PDP includes the development square footage for the Watermark project (both Area A and B) and would supplant the existing vested approvals. For the purposes of the EIR, the development of Area B (Lots 1 and 2) will be considered as part of the existing conditions, except where noted.

### 3.3.1 Miramar Ranch North Community Plan/General Plan Amendment

The project site is identified as the Merey Site in the Miramar Ranch North Community Plan and shown for industrial park uses. (See Figure 2-7, Miramar Ranch North Community Plan – Existing Land Use Designations.) The project is proposing an amendment to the Miramar Ranch North Community Plan to change the land use designation from Industrial to Regional Commercial (see Figure 3-1, Miramar Ranch North Community Plan – Proposed Land Use Designations). Specific elements of the Community Plan that are affected by this proposed change include the Industrial and Commercial elements. Additional minor changes are proposed to the Miramar Ranch North Community Plan text and graphics to ensure consistency with the proposed amendment for the Watermark throughout. The proposed revisions to the Miramar Ranch North Community Plan are detailed below.

- Figure 4, Miramar Ranch North Community Plan: Land Use Plan: Proposed change in Land Use Plan to identify the Watermark project site as Regional Commercial (see Figure 3-1).
- Various tables and exhibits: Change the land use on the project site from Industrial to Regional Commercial and adjust area acreages, as applicable, throughout the Community Plan.
- Transportation Element: Update text to reflect current and on-going improvements to I-15; expand discussion of the Mercy Road Interchange to reflect change in land uses on the Watermark site.

- Industrial Element: Adjust area acreages and revise text and exhibits to remove the Mercy Site (the Watermark project site) from discussion as a location for industrial development.
- **Commercial Element:** In addressing the *Community Commercial Needs* (Section 6.1 of the Community Plan's Commercial Element Chapter), the proposed amendment expands discussion to include the Watermark project as a regional commercial center to provide *"shopping opportunities currently served by North County Fair in Escondido that are not provided by local neighborhood and community retail*", with Class A office buildings (MedImpact) designed as a corporate campus. Additional text is proposed for the Community Plan's Commercial Element that defines the purpose of the change in land uses for the project site, stating:

The growth in the I-15 corridor has increased the demand for specialty retailers that is being served by the emergence of lifestyle retail centers. The unique nature of this type of center allows for a site design on a much smaller footprint than the traditional regional center that includes multiple anchor tenants. The Mercy Interchange site provides an optimum location between two major regional centers to serve an emerging clientele. This site is also well located to serve an increased demand in high quality corporate office to reflect shifts in the economic structure of the region. The General Plan designation of Regional Commercial provides flexibility to meet the needs of any combination of regional commercial office and/or retail.

Proposed revisions to Section 6.2 - Location of Commercial Development – of the Community Plan's Commercial Element addresses the types of uses that can occur at the project site, stating:

Regional commercial development is proposed for the southeast corner of the Mercy Interchange site, with access provided from Scripps Poway Parkway and Scripps Highlands Road. As should in Figure 17 [of the Community Plan], this location takes advantage of the proximity to I-15 to provide employment and attract users from the I-15 corridor communities and the region at large. This designation allows a flexible range of commercial service, civic, retail, office, and limited industrial uses; residential is prohibited at this location. The larger size of this site (almost 30 acres) provides greater opportunities for an integrated design of commercial office and/or retail uses that would not typically compete with the community serving retail identified in the Plan. The following uses are appropriate:

- Commercial recreation, such as sports club, hotel, restaurant and/or movie theater.
- Corporate and business offices and commercial support services.
- *Financial institutions.*
- Retailing establishments.
- Service station and/or automobile repair.
- Food store and/or drugstore.

**3.0 PROJECT DESCRIPTION** 

Image: marked back of the control
Image: Decision of the Critery of the Crite
Image: constraint of the constraint
Land Location Level and Le
Land Use Plan 4

The Watermark Final Environmental Impact Report

Page 3-6 September 2013

Additionally, the proposed amendment inserts the following text into Section 6.3 – *Commercial Element Guidelines* – of the Community Plan's Commercial Element:

The regional commercial center should be developed under appropriate commercial and industrial zoning that prohibits residential uses and provides flexibility by allowing for corporate and multi-tenant office uses and community and regional serving retail establishments.

In addition, it is recommended that the regional commercial be undertaken as a planned development permit (PDP). The PDP should be utilized to implement the proposals in regard to uses outlined in Section 6.2 [of the Community Plan], while providing the flexibility to adjust the combination of commercial office and retail subject to the limitations of the final traffic impact study. The use of design guidelines in conjunction with the PDP would ensure development would be aesthetically pleasing and visually compatible within the site.

### 3.3.2 Proposed Zoning

As stated in Section 2.8, *Existing Zoning*, and shown in Figure 2-8, the project site is currently zoned IP-2-1 (Industrial-Park). The purpose of the City's IP zones is to provide for high quality science and business park development. The property development standards of this zone are intended to create a campus-like environment characterized by comprehensive site design and substantial landscaping. Restrictions on permitted uses and signs in this zone are provided to minimize commercial influence. The IP-2-1 zone allows a mix of light industrial and office uses.

The project proposes to rezone a portion of the project from the existing IP-2-1 zone to CR-2-1 (Commercial-Regional). Specifically, Lots 1 and 2 of Area B (where the MedImpact facilities are located), 11.97 acres, would remain zoned IP-2-1. The remainder of the project site, 22.42 acres, would be rezoned to CR-2-1 (see Figure 3-2, *Proposed Zoning*).

The CR zones provide areas for a broad mix of business/professional office, commercial service, retail, wholesale, and limited manufacturing uses. The CR zones are intended to accommodate large-scale, high intensity developments. According to the City's Land Development Code, property within the CR zones would be primarily located along major streets, primary arterials, and major public transportation lines. The CR-2-1 zone allows for regional-serving commercial and limited industrial uses with an auto orientation but no residential use.

### 3.3.3 Vesting Tentative Map

In order to facilitate development of the Watermark project, a Vesting Tentative Map (VTM) is proposed. The Watermark VTM details grading required for the project and final elevations, as well as necessary infrastructure, and has been prepared in accordance with the State Subdivision Map Act and City requirements (see Figure 3-3, *Watermark Vesting Tentative Map*). The VTM also subdivides the property for financing and ownership purposes. The *Proposed Lotting* for the project is shown in Figure 3-4 (see Figure 2-4, *Existing Site Conditions*, for existing approved lotting).



Figure 3-2. Proposed Zoning

Of the approximately 34.96-acre project site, roughly 18.9 acres would be re-graded for the Watermark project. As the project site has been previously graded in accordance with original and existing project approvals, for the most part, the proposed Watermark project would require only finish grading to accommodate development. Primary changes to the existing graded site involve grading for the proposed driveway access off Scripps Poway Parkway. At this location, previously graded slopes along Scripps Poway Parkway would be re-graded to allow for construction of the driveway entrance. Grading would also occur for a basement level below the proposed theater, where retail shops would be located. Earthwork for the project would require approximately 140,000 cubic yards of cut and approximately 75,000 cubic yards of fill. Approximately 65,000 cubic yards of material would be exported. Cut slopes would range up to 25 feet in height, and fill slopes would reach 17 feet in height. All manufactured slopes would have a gradient of 2:1.

Retaining walls are proposed at the edge of the graded pads in various locations on the site. A retaining wall would occur along the southern portion of the site, ranging in heights of three feet to six feet for a length of approximately 360 feet. A smaller section of retaining wall, approximately 150 feet in length and reaching a height of 15 feet, would occur in the western portion of the site. Retaining walls, ranging from three feet to six feet, would also be used along Scripps Poway Parkway. Smaller walls would be used at the project entry and internally in conjunction with buildings.

### 3.3.4 Planned Development Permit

A PDP is proposed for the Watermark project. According to the City's Land Development Code, the purpose of a PDP is "... to provide flexibility in the application of development regulations for projects where strict application of the base zone development regulations would restrict design options and result in a less desirable project. The intent of the Planned Development Permit regulations is to accommodate, to the greatest extent possible, an equitable balance of development types, intensities, styles, site constraints, project amenities, public improvements, and community and City benefits. ..." A PDP is proposed for the Watermark project to allow for flexibility in buildings heights and wall heights, and to allow shade structure on the roof of the parking garage as described below under Proposed Deviations. The Watermark PDP would apply to 22.42 acres of the project site where a mixed-use development comporised of commercial office space, commercial retail space, a movie theater, and 130-room hotel are proposed. The remaining 11.97 acres of the project site would retain the approved MedImpact corporate headquarters CUP/PID 99-1027.

**3.0 PROJECT DESCRIPTION** 

SYMBOL		11111111	482				0 0 0 0	+ - - -		7.0%				٥	▣.			(519.2FF)	(cs)	SECTION LETTER A SHEET # OF SHEET # OF A SHEET # OF SECTION CUT A SHEET # OF SECTION CUT A SHEET # OF	(E)	( <b>T</b> S	(NO OBSTACLES	TES	OR EQUAL PER WOTR	NICH LANDSCAPE OR EQUAL PER WOTR	al. Per wotr T upon final engineering and are
ITEM	PROPERTY UNE/TM BOUNDARY	RIGHT-OF WAY	NU ACCESS EASEMENT	PROPOSED CONTOUR	CUT/FILL SLOPE (2:1 MAX)	DAYLIGHT LINE	EXISTING DRAINAGE DITCH	DAYLIGHT LINE	PROPOSED DRAINAGE DITCH	FLOW DIRECTION AND SLOPE RETAINING WALL	LIMIT OF WORK	PROPOSED STORM DRAIN	EXISTING STORM DRAIN	CLEANOUT	CURB INLET	RIDGE LINE	UNDERGROUND BUILDING	BUILDING FINISH FLOOR ELEVATION	STEP BETWEEN BUILDING FINISH FLOORS	CROSS SECTION NODE	PROPOSED SIGNALIZED INTERSECTION	EXISTING SIGNALIZED INTERSECTION	useruty trangle per municipal code 113.0273 () over 3FT high alloned within visibility triangle)	STORWWATER BMP NOT	BELOW GROUND MEDIA FILTER VAULT	<ol> <li>BIU-SWALE PER WUR</li> <li>UD AREA - PERMOUS PAVEMENT, FLOW THROW</li> </ol>	CATCH BASIN FILTER INSERT OR EQUA STORMWATER QUALITY BMPS WIL BE DEPENDANT SUBJECT TO CHANGE.



The Watermark Final Environmental Impact Report

Page 3-10 September 2013

Figure 3-3. Watermark Vesting Tentative Map



Figure 3-4. Proposed Lotting

### Proposed Site Plan and Development Intensity

The project proposes numerous buildings to accommodate a variety of retail stores, restaurants, hotel, movie theater, and offices. Anchor tenants would be located in the northeast and northwest corners of the site. A movie theater is proposed in the northwest portion of the site, wrapped with retail stores at the base. A hotel is planned in the southwest corner of the site. Free-standing/sit-down restaurant(s) and smaller scale shops and eateries are planned throughout the site. A retail use could also occur on top of adjacent to the proposed parking garage, located in the western portion of the site. Buildings would range in heights of one story to six stories. (See Figure 3-5, *Watermark Site Plan.*)

The Site Plan includes pedestrian areas and gathering places, with smaller gathering areas and seating nodes contrasted by open lawn areas designed to accommodate large events and community interaction. Proposed as the central focal point for the Watermark project is a grand lawn and oversized water feature. The change in elevation through the project is addressed with decorative stairs intended to reflect the grand stairways of Europe. (See Figure 3-6, *Watermark Pedestrian Plan.*)

The project would provide a total of 2,191 parking spaces to serve the range of uses that could occur on the site. A parking structure, located in the southwest portion of the site, would provide 1,727 parking spaces in five levels. Other surface parking lots located throughout the site would provide an additional 246 parking spaces, and 218 parking spaces would be provided in an adjacent existing off-site parking garage. The proposed on-site parking structure would be four feet above the height limit of 60 feet allowed in the CR-2-1 zone. The additional height of the canopy/shade structures on the top level of the parking garage results in a structure height of 74-<u>100</u> feet. To articulate the parking structure, a corner feature may be included, which would be accommodated within the 103 feet structure height.

The Watermark Design Guidelines address form, colors and materials for the future commercial office and retail components of the project in a manner which, while they may not be the same, is compatible with the architecture of the existing commercial office campus. Comprehensive landscape, internal streetscape, and lighting plans are also of significant importance in creating a unified project and would also be reviewed by the developer to ensure conformance with the established design aesthetic.

The architectural guidelines utilize a *Kit of Parts<sup>1</sup>* that has been developed to be used in whole or in combination to achieve the intended ambiance and character of the project. Based on the proposed Design Guidelines, each new building should incorporate a sufficient number of architectural elements so as to reflect the Watermark design character. Buildings are to employ staggered setbacks, varied building heights, widths, shapes, orientations, colors, and materials. A variety of architectural elements have been identified in the Kit of Parts to achieve the project's goal of a lasting and high quality design. To provide visual interest to the roofline, building heights would be varied with roof materials that include metal and clay tile as examples of a high quality finish.

<sup>&</sup>lt;sup>1</sup> "Kit of Parts" refers to Term used to describe elements that are common amongst the buildings of a multiple building project that create a unified architectural character within the project. Included in these elements can be various types of building massing and articulation, forms and materials of entries, windows, awnings and other architectural treatments. The use of similar, but not identical, elements on multiple buildings creates a unified character within the project.





The Watermark Final Environmental Impact Report

Page 3-13 September 2013

**3.0 PROJECT DESCRIPTION** 



The Watermark Draft Environmental Impact Report

Page 3-14 September 2013

Horizontal rooflines should be further detailed with cornices and molding that embellishes the top of the building. All roof mounted equipment, apparatus and vents would be architecturally screened from view and painted for compatibility with the roof color.

The proposed Design Guidelines call for metal and fabric canopies and wood trellises to provide additional building articulation, as well as shade and opportunities for enhanced landscaping. Stucco siding is to be enhanced by the use of a variety of accent materials, including metal, wood, brick and stone veneer. Precast concrete elements for base and vertical building elements are to provide relief from the building wall.

As shown in Table 3-2, *Watermark – Maximum Structure Height*, the project includes design elements and structure heights that exceed the height requirements of the CR-2-1 zone. The allowable maximum structure height in the CR-2-1 zone is 60 feet. In order to allow for the architectural design elements and building height of the proposed parking structure, a deviation to the height requirements of the CR-2-1 zone is proposed.

As shown in Figures 3-7a - 3-7c, *Project Elevations*, the Watermark project would feature architectural elements that are to be complimentary to the project's design, as well as the architecture of the MedImpact office buildings. The project's architectural elements are intended to provide interesting and identifiable features, which would allow pedestrians and the motoring public to easily find their destinations. Architectural features would be located on the hotel, the market, the theater, in the plaza gathering space, and at the project entry. The architectural features are to follow the architectural theme set for the project and may include a clock tower or bell tower in the plaza open area to function as a way-finder for community events. Other architectural features would provide vertical relief to the façades and would create focal points around the project for both pedestrians and passing vehicles.

Project access is taken from a primary entry off Scripps Highlands Drive and a secondary rightin/right-out driveway from Scripps Poway Parkway. The entry from Scripps Highlands Drive continues into the project site along a private drive that runs between the IP-2-1 and proposed CR-2-1 portions of the project site. This private drive continues to the western perimeter of the project site, where access is available to the parking structure; the valet parking and surface lots may be accessed from the private drive as it enters the project site in the east. The driveway off Scripps Poway Parkway allows access to the surface lots in the northeast or the parking structure in the western portion of the project site.

### Design Guidelines

The project includes *Design Guidelines* which present the overall site development, architectural character, and controls for the project<u>Area A of the proposed project</u>. Area B would remain subject to the approved MedImpact corporate headquarters CUP/PID 99-1027. An objective of the Watermark project is to create a coherent and signature design statement as a community gateway to Miramar Ranch North. According to the proposed Design Guidelines, the purpose of the Architectural Design Guidelines is "to maintain the high level of architectural quality established with the existing MedImpact campus by maintaining a complimentary architectural vocabulary for the retail project in order to unify the multiple buildings on the site, to maintain a level of design quality in making the project aesthetically pleasing, and to enhance the area in which the project is located."
**3.0 PROJECT DESCRIPTION** 



**3.0 PROJECT DESCRIPTION** 







The Watermark Final Environmental Impact Report

**3.0 PROJECT DESCRIPTION** 



The Watermark Final Environmental Impact Report

The Design Guidelines address form, colors, and materials for the future commercial office, retail, and entertainment components of the project in a manner that is compatible with the architecture of the existing MedImpact commercial office campus.

The approach to implementing development of the Watermark project in a manner that is compatible with the architecture of the existing MedImpact commercial office campus is by utilizing a *kit of parts* that has been developed to be used in whole or in combination to achieve the intended ambiance and character of the project. Each new building within the Watermark project would incorporate a sufficient number of architectural elements to reflect the Watermark design character, employing staggered setbacks, and varied building heights, widths, shapes, orientations, colors, and materials.

A variety of architectural elements have been identified in the kit of parts. To provide visual interest to the roof line, the Design Guidelines require that building heights be varied with roof materials that include metal and clay tile as examples of a high quality finish. Horizontal rooflines should be further detailed with cornices and molding that embellishes the top of the building. All roof mounted equipment, apparatus, and vents shall be architecturally screened from view and painted for compatibility with the roof color. Metal and fabric canopies and wood trellises are recommended to provide additional building articulation as well as shade and opportunities for enhanced landscaping. Stucco siding is to be enhanced by the use of a variety of accent materials, including metal, wood, brick, and stone veneer. Precast concrete elements may be used for base and vertical building elements to represent durable and long-lasting construction materials while providing relief from the building wall.

The Design Guidelines require that light fixtures for major tenants, shops, and individual buildings be of matching and/or complimentary design. Additionally, landscaping and architectural features shall be illuminated and accented with appropriate lighting. Parking structure and lot lighting shall match the site lighting theme; building mounted flood lighting shall not be used to illuminate parking areas. All lighting shall be screened to respect the residential area located to the east of the Watermark. Signage for the Watermark project would conform to the City of San Diego Land Development Code Sign Regulations.

## Proposed Deviations

Approval of the PDP for the Watermark project would require adoption of deviations that are proposed in order to create "a more desirable project than would be achieved if designed in strict conformance with the development regulations of the applicable zone" [LDC Section 126.0604 (5)]. Specifically, the Watermark project proposes a deviation from the 60 foot maximum structure height of the CR-2-1 zone to allow for design and architectural elements, and to accommodate the parking garage.

A deviation request from the 60 foot maximum structure height of the CR-2-1 would allow Watermark to proceed with development of additional Class A office space as envisioned in the existing planned industrial development permit (CUP/PID No. 1027). This deviation is necessary to maintain the existing vested development rights under that approval. Should the feasibility of developing commercial retail uses be unachievable under future market conditions, the Watermark

## projecywould be completed as a corporate office park as envisioned under the existing entitlement. The structure height deviation for office buildings would apply to proposed Lots 3, 4 and 5.

A deviation for structure height on proposed Lot 9 for the plaza would enable the development of a landmark architectural feature, such as a clock tower, bell tower, or other architectural features to designate the gateway to the community. This would assist in achieving an objective of the Community Plan to develop the Mercy Interchange area as an attractive gateway to the community. Such a feature would provide a visible landmark to I-15 travelers, as well as persons entering Miramar Ranch North. The project proposes a deviation in height for landmark architectural features from 60 feet allowed in the CR-2-1 zone to 103 feet.

A deviation for height would also allow for the construction of a parking garage to accommodate parking needs for the project. This would reduce land required for surface parking, providing for more public amenity open space. The project proposes a deviation for the parking structure from 60 feet to a maximum of 103 feet.

A summary of the deviations for maximum allowable height are listed below in Table 3-2, *Watermark* – *Maximum Structure Height*.

LDC Zone	CR-2-1	
	Allowed	Proposed
Hotel (Lot 3)	60 feet	90 feet
Parking	60 feet	74 -103 feet
Structure (Lot 4)		
Theater/Plaza	60 feet	103 feet
(Lot 6V and 9)		
Market (Lot 9)	60 feet	75 - 85 feet
Justification	An increased height on proposed Lots 3, 4, and 5 enables the flexibility to implement the	
	previously approved Planned Industrial Development permit in a vertical and/or horizontal mixed use project of commercial office and retail uses. Allowing taller office buildings provides greater flexibility in the final site design for the location of both office and parking structures, increasing the available open space for employee amenities	
	and landscaping. An increase in structure height on proposed Lot 9 shall be for the	
	development of an architectural feature, such as a clock tower, bell tower, or	
architectural features to designate the gateway to the community.		way to the community.

Table 3-2. Watermark - Maximum Structure Height

## Landscape Concept Plan

The proposed Landscape Concept Plan (see Figure 3-8, *Watermark Landscape Plan*) includes the use of indigenous and/or drought tolerant plant material, whenever possible. No invasive or potentially invasive species shall-would be allowed. Planting is intended to be a connecting device linking the various pieces of the project and design style. The Landscape Concept Plan emphasizes a garden setting, where plant material would be used to help define spaces, screen objectionable views, encourage circulation paths, highlight entry points, and provide softness and scale to the architecture. Evergreen, deciduous, and flowering material are proposed throughout the project. Located adjacent to open space slopes, the perimeter planting is proposed as a blend of native material and native friendly (i.e. non-invasive) fire safe planting. (See Table 3-3, *Landscaping Materials*, below.)

Circulation throughout the project is accentuated with a hierarchy of landscape treatments. Enhanced paving at major intersections and nodes is proposed to signify pedestrian/vehicle interaction areas. Vehicle nodes with small medians are proposed to help slow the traffic flow, as well as break up long linear drives. Street trees are proposed to define vehicle/pedestrian spaces and to provide shade and scale to the street scene. Entry points would be highlighted with decorative trellis work and enhanced plantings.

Landscaping throughout the Watermark site is characterized by accent planting; foreground, midground, and background planting; and trees and palms. Around parking areas, evergreen trees and shrubs would screen parking from adjacent uses and, where applicable, canopy trees would be provided with permeable paving border. Plantings along primary roadways, such as Scripps Poway Parkway and Scripps Highlands Road, would seek to maintain the existing street tree program. Enhanced plantings would be present at project entries and other primary focal points. Areas of special interest would have customized planting schemes, described below. Locations of areas of special interest are shown on Figure 3-8, *Watermark Landscape Plan*.





The Watermark Final Environmental Impact Report

Page 3-22 September 2013

## A – Entry from Scripps Highlands Road

Landscaping at the intersection of Scripps Highlands Road and the private internal street (Figure 3-9, *Entry from Scripps Highlands Road Landscape Design*) would offer an opportunity for dual purpose landscaping that both accents this entry point as well as screens interior parking from the roadway. A shrub hedge, as well as a retaining wall and background planting, would be used to screen parking from the roadway. At the intersection, accent planting and a signage/monument opportunity area would announce the entrance to the Watermark. Within the parking area, where applicable, a permeable paving band would maximize planting area; and parking lot landscaping would include foreground, midground, and accent planting.



Figure 3-9. Entry from Scripps Highlands Road Landscape Design

## B – First Project Driveway from Private Drive

Accent landscaping, enhanced paving, and architectural embellishments would denote the primary site entry off the interior private drive (Figure 3-10, *First Project Driveway from Private Drive Landscape Design*). Palm and other trees in grate planters would line the drive aisle, as well as decorative pots with planting and enhanced pedestrian paving opportunities. Off the east side of the drive aisle, a pedestrian plaza would invite pedestrian access to the project site. Within this connector plaza, a water feature or other focal element would be present within view of the entry aisle, with access terracing into the project site via grand steps. Raised planters and decorative light poles or other focal elements would accent the enhanced paving and other hardscape elements. Roadway plantings would include low foreground planting, midground planting, and background/screening planting.



Conceptual design for illustrative purposes only.

Figure 3-10. First Project Driveway from Private Drive Landscape Design

## C – Primary Public Plaza

Central to the Watermark project is a large plaza space designed to accommodate community gatherings and events. For this purpose, this space encompasses a grand lawn/event area, edged with paving and background planting (Figure 3-11, *Primary Public Plaza Landscape Design*). Seating opportunities, as well as a children's play area, water feature, focal elements, and thematic elements, would be located surrounding the grand lawn/event area. Opportunities for enhanced vehicular paving are present at points where pedestrian crosswalks intersect interior drive aisle. Landscaping would include accent planting along the drive aisle, and foreground, midground, and background planting surrounding the grand lawn/event area.



Figure 3-11. Primary Public Plaza Landscape Design

## D – Secondary Driveway from Private Drive

Secondary access from the private street would have a less formal approach to its architectural elements and landscaping than the entry at Scripps Highlands Road (Figure 3-12, *Secondary Driveway from Private Drive Landscape Design*). Background planting and enhanced paving for both pedestrian and vehicular circulation would denote this entry. Removable themed bollards could restrict vehicular access to the focal element at the center of a roundabout with decorative paving, accent planting, and foreground and midground planting. Seating opportunities would exist along the private street with decorative planting in pots. Trees in grates surrounding a thematic element would be located to the west of the roundabout.





Figure 3-12. Secondary Driveway from Private Drive Landscape Design

## E – Off-ramp Frontage

An interior vehicular circulation roadway would run parallel to the I-15 northbound off-ramp in the northwest portion of the project site. This roadway would have access walkways and screening landscaping on the east side where the roadway meets with interior project elements (Figure 3-13, *Off-ramp Frontage Landscape Design*). On the west side, where the roadway is adjacent to the off-ramp, enhanced landscaping with accent planting, as well as foreground, midground, and background planting would create a pleasant aesthetic/landscape screening from the freeway.



Conceptual design for illustrative purposes only.

Figure 3-13. Off-ramp Frontage Landscape Design

## F – Entry off Scripps Poway Parkway

The project entry off Scripps Poway Parkway would be lined with foreground, midground, and background plantings which also screen retaining walls necessary for lot development (Figure 3-14, *Entry off Scripps Poway Parkway Landscape Design*). A T-intersection would be accentuated with enhanced vehicular paving and bollards directing traffic. Enhanced paving with seating opportunities for pedestrians would occur at the corners of this intersection, accented with decorative pottery with plantings, as well as taller planters. Thematic light/pole elements would tie in with the overall project aesthetic.



Conceptual design for illustrative purposes only.

# Figure 3-14. Entry off Scripps Poway Parkway Landscape Design

Provided in Table 3-3, *Landscaping Materials*, is the proposed landscaping materials that would be used at the Watermark project.

Botanical Name	Common Name	
Agave species	Agave	
Clystoma callistegiodes	Violet trumpet vine	
Cycas revoluta	Sago palm	
Distictis buccinatoria	Blood red trumpet vine	
Ficus pumila	Creeping fig	
Mandevillea species	Mandevillea	
Passiflora species	Passion vine	
Parthenocissus tricuspidata	Boston ivy	
Phormium species	New Zealand flax	
Strelitzia species	Bird of paradise	
Yucca species	Yucca	

## Table 3-3. Landscaping Materials Accent Shrubs

#### Foreground Shrubs and Groundcovers

Botanical Name	Common Name
Aeonium species	Aeonium
Agapanthus species	Lily of the Nile
Baccharis species	Coyote bush
Bougainvillea species	Bougainvillea
Carissa species	Natal plum
Cistus hybridus	Rockrose
Coprosma species	Mirror plant
Cuphea hyssopifolia	Mexican false heather
Geranium species	Geranium
Hemerocallis species	Daylily
Liriope species	Lily turf
Myoporum species	Myoporum
Rosmarinus species	Rosemary
Trachelospermum jasminoides	Star jasmine

#### Midground Shrubs

Botanical Name	Common Name
Anigozanthos species	Kangaroo paw
Buxus species	Boxwood
Hebe species	Hebe
Lantana species	Lantana
Lavandula species	Lavender
Salvia species	Sage

Botanical Name	Common Name	
Azalea species	Azalea	
Euonymus species	Euonymus	
Ceanothus species	Wild lilac	
Ligustrum species	Japanese privet	
Nandina species	Heavenly bamboo	
Photinia species	Photinia	
Plumbago species	Cape plumbago	
Rhaphiolepis species	Indian hawthorn	
Rosa species	Rose	
Viburnum species	Viburnum	
Xylosma species	Xylosma	

### Background Shrubs

#### **Trees and Palms**

Botanical Name	Common Name
Agonis flexuosa	Australian willow myrtle
Arbutus species	Strawberry tree
Archontophoenix cunninghamiana	King palm
Bambusa species	Bamboo
Bauhinia species	Orchid tree
Butia capitata	Pindo palm
Brahia armata	Mexican blue palm
Cercis occidentalis	Western redbud
Cinnamomium camphora	Camphor tree
Citrus species	Citrus
Cupaniopsis anacardiodes	Carrotwood
Eribotrya deflexa	Bronze loquat
Erythrina species	Coral tree
Eucalyptus species	Eucalyptus
Jacaranda species	Jacaranda
Juniper species	Juniper (topiary)
Koelreuteria bipinnata	Chinese lantern
Ligustrum species	Privet
Liquidambar styraciflua	Sweet gum
Liriodendron tulipifera	Tulip tree
Magnolia species	Magnolia
Olea species (fruitless varieties)	Fruitless olive
Phoenix dactylifera	Date palm
Pinus species	Pine
Platanus species	Sycamore
Podocarpus macrophyllus	Yew pine
Prunus caroliana	Carolina cherry
Prunus cerasifera	Purple leaf palm
Pyrus species	Pear
Quercus agrifolia	Coast live oak
Quercus ilex	Holly oak
Quercus virginiana	Southern live oak

Rhaphiolepis species	Majestic beauty/Indian hawthorn
Rhus lancea	African sumac
Robina species	Locust
Syagrus romanzoffiana	Queen palm
Tabebuia species	Pink trumpet tree
Tipuana tipu	Tipu tree
Tristania conferta	Brisbane box

## 3.3.5 Street Vacation

The project proposes the vacation of Scripps Gateway Court. Currently, Scripps Gateway Court is a public street that begins at Scripps Highlands Drive on the east and ends internal to the project site as a cul-de-sac. This street provides access to the recently constructed MedImpact office building. The proposed Street Vacation would allow for integration of this area into the overall project design. A private drive would be provided to maintain necessary traffic flow, provide for efficient access and movement of people and vehicles through the project, and would connect with proposed uses and the proposed parking structure and other surface parking areas.

## 3.3.6 Conditional Use Permit

According to the City's Land Development Code (Chapter 12, Article 6, Division 3: Conditional Use Permit Procedures), the purpose of a Conditional Use Permit is to allow "uses that may be desirable under appropriate circumstances, but are not permitted by right in the applicable zone." Uses are reviewed "on a case-by-case basis to whether and under what conditions the use may be approved at a given site", and "conditions may be applied to address potential adverse effects associated with the proposed use."

As part of the Watermark project and consistent with recommendations in the Miramar Ranch North Community Plan, a movie theater is proposed and would be located in the CR-2-1 zone. According Section 126.0303(b) of the City's Land Development Code, a Conditional Use Permit (CUP) is required for *theaters that are outdoor or over 5,000 square feet in size*. The size of the movie theater at the Watermark project could range up to 45,000 square feet. Therefore, the proposed movie theater would be permitted in the CR-2-1 zone, with application of a CUP.

# 3.4 DISCRETIONARY ACTIONS

A discretionary action is an action taken by an agency that calls for the exercise of judgment in deciding whether to approve or how to carry out a project. For the Watermark project, the following discretionary actions would be considered by the San Diego City Council:

General Plan Amendment and Community Plan Amendment – The approximately 34.96-acre project site is located within the Miramar Ranch North Community Plan Area and is designated for Industrial/Business Park uses. The project proposes to change the land use designation to Regional Commercial. Because the Community Plan would be amended, this would result in an amendment to the City's General Plan as the Community Plan functions as the land use plan for the Miramar Ranch North community of the City.

The project site is identified as a location for Prime Industrial Lands in the City. Prime Industrial Lands are defined in the Economic Prosperity Element of the City's General Plan as "areas that support export-oriented base sector activities such as warehouse distribution, heavy or light

manufacturing, research and development uses. These areas are part of even larger areas that provide a significant benefit to the regional economy and meet General Plan goals and objectives to encourage a strong economic base." In order to develop the site with the proposed mix of commercial and office uses, the project would remove the Prime Industrial Lands identification from Area A of the project site and would change the General Plan land use designation from Industrial to Regional Commercial for Area A. The General Plan would need to be amended to reflect these changes.

- Rezone A rezone is proposed for a portion of the site to change the existing IP-2-1 zone to CR-2-1.
- Vesting Tentative Map In order to facilitate development of the Watermark project, a VTM is processed. The Watermark VTM details proposed grading for the project, as well as necessary infrastructure, and has been prepared in accordance with the guidelines of the State Subdivision Map Act and City of San Diego requirements.
- Planned Development Permit The PDP approval would establish the Design Guidelines and development intensity ranges for the project and allows for minor variations deviations to the regulations of the proposed CR-2-1 zones through proposed deviations. The deviations would provide are proposed to provide for a superior project than could occur with a strict interpretation of the CR-2-1 zone regulations.
- **Conditional Use Permit** A CUP is required for the proposed movie theater.
- Street Vacation The Street Vacation is required to vacate a Scripps Gateway Court. In its place, a private drive would provide vehicular, bicycle, and pedestrian access for the project to efficiently serve existing (MedImpact facilities on Area B) and proposed developments.
- Public Utility Easement Vacation Vacation of a Public Utility Easement containing storm drain, water, and sewer is proposed.
- Environmental Impact Report Concurrent with the Watermark discretionary actions, an EIR has been prepared in accordance with the provisions of the CEQA. The EIR (SCH No. 2010091079) evaluates the land use, circulation, and infrastructure improvements resulting from implementation of the Watermark project and the potential environmental impacts that would result from their implementation. Review and certification of this EIR by the decision maker would complete the environmental review for the project in accordance with CEQA and City regulations.

As described in Section 1.4, Responsible and Trustee Agencies, of this EIR, review by Caltrans, a State agency, would be required for the proposed project.

**Caltrans** - Project features which necessitate encroachment into freeway easements and access rights for improvements within Caltrans' rights-of-way would require coordination with Caltrans for those improvements. <u>Caltrans approval would be required for any work within its rights-of-way.</u>

Additionally, the project requires review by the Regional Water Quality Control Board (RWQCB) and the Federal Aviation Administration (FAA).

- NPDES Permit The project would comply with National Pollutant Discharge Elimination System (NPDES) requirements for discharge of storm water runoff associated with construction activity. Compliance also requires conformance with applicable Best Management Practices (BMPs) and development of a Storm Water Pollution Prevention Plan (SWPPP) and monitoring program plan. (Water quality is addressed in Section 5.12, *Hydrology/Water Quality*, of this EIR.)
- Obstruction Evaluation/Airport Airspace Analysis, Part 77 Determination (Federal Aviation Administration) The project's proximity to MCAS Miramar requires notification to the Federal Aviation Administration (FAA) in order to conduct an Obstruction Evaluation/Airport Airspace analysis under Title 14 code of Federal Regulations, Part 77. The project has completed an initial request for the aeronautical study and has received Determination of No Hazard to Air Navigation for the project (see Appendix N). Individual structures would be required to file subsequent notification to the FAA at least 30 days before the earlier of a) the date proposed construction or alteration is to begin, or b) the date the application for a construction permit would be filed.

Additionally, the Watermark project was reviewed for consistency with the MCAS Miramar ALUCP. A letter from MCAS Miramar determined that the proposed project is contained within the MCAS Miramar AICUZ Study Area and is: within the adopted AIA; 2) outside the 60+ dB community noise equivalent level noise contours; 3) outside all Accident Potential Zones; 4) beneath the Outer Horizontal Surface of MCAS Miramar (Federal Aviation Regulation part 77); and beneath and/or near established fixed- and rotary-wing flight corridors for aircraft transiting to and from MCAS Miramar. It was determined that the propose project is consistent with the AICUZ noise and safety compatibility guidelines.

# 4.0 HISTORY OF PROJECT CHANGES

The section chronicles the physical changes that have been made to the project in response to environmental concerns raised during the City's review of the project.

## Changes to Bicycle Circulation

In the initial development of the proposed Watermark project, staff identified a potentially significant issue associated with bicycle circulation. Currently, a bicycle path parallels the east side of I-15, connecting the Scripps Miramar Ranch and the Miramar Ranch North communities. The bike path is constructed as a separated bike path outside the I-15 right-of-way that begins at Erma Road in Scripps Miramar Ranch and ends at Scripps Poway Parkway in Miramar Ranch North. The bike path joins with bike lanes on Erma Road at the south and Scripps Poway Parkway on the north. Because the project proposes the addition of a right-in/right-out driveway along Scripps Poway Parkway could cross into vehicles accessing the project site at the new driveway off Scripps Poway Parkway.

In order to eliminate the potential for bicycle/motorist conflicts, the applicant re-designed the project to include a separated bike lane. The project proposes that the existing bike lane on the south side of Scripps Poway Parkway be relocated and constructed as a multi-use path adjacent to the project site. This would require a limited dedication from the Watermark project site in order to be achieved. Figure 4-1, *Proposed Multi-Use Path*, shows a preliminary layout of the right in/out driveway with the proposed dedication for the bike path highlighted. Figure 4-2, *Multi-Use Path Cross-Section*, shows this facility in cross-section.



Figure 4-2. Multi-Use Path Cross-Section